Learning and Flexibility for Water Supply Infrastructure Planning under Diverse Uncertainties

by

Sarah Marie Fletcher

B.A., University of Pennsylvania (2010) S.M., Massachusetts Institute of Technology (2012)

Submitted to the Institute for Data, Systems, and Society in partial fulfillment of the requirements for the degree of

Doctor of Philosophy in Engineering Systems

at the

MASSACHUSETTS INSTITUTE OF TECHNOLOGY

June 2018

© Massachusetts Institute of Technology 2018. All rights reserved.

Author
Institute for Data, Systems, and Society
May $30, 2018$
Certified by
Kenneth Strzepek
Research Scientist, Thesis Supervisor
Certified by
Olivier de Weck
Professor of Aeronautics and Astronautics and Engineering Systems, Doctoral Committee Chair
Certified by
Dara Entekhabi
Bacardi and Stockholm Water Foundations Professor, Doctoral Committee Member
Certified by
James Wescoat
Aga Khan Professor of Architecture, Doctoral Committee Member
Certified by
Afreen Siddiqi
Research Scientist, Doctoral Committee Member
Accepted by
Stephen Graves
IDSS Graduate Officer

Learning and Flexibility for Water Supply Infrastructure Planning under Diverse Uncertainties

by

Sarah Marie Fletcher

Submitted to the Institute for Data, Systems, and Society on May 30, 2018, in partial fulfillment of the requirements for the degree of Doctor of Philosophy in Engineering Systems

Abstract

Water supply infrastructure planning faces many uncertainties. Uncertainty in short-term in rainfall and runoff, groundwater storage, and long-term climate change impacts water supply forecasts. Population and economic growth drive urban water demand growth at rapid but uncertain rates. Overbuilding infrastructure can lead to expensive stranded assets and unnecessary environmental impacts, while under building can cause reliability outages with impacts on the economy, ecosystems, and human health. This dissertation assesses the potential for Bayesian learning about uncertainty to enable flexible, adaptive approaches in which infrastructure can be changed over time to reduce cost risk while achieving reliability targets. It develops a novel planning framework that: 1) classifies uncertainties and applies appropriate, differentiated uncertainty analysis tools, 2) applies Bayesian inference to physical models of hydrology and climate to develop dynamic uncertainty estimates, and 3) uses stochastic dynamic programming and engineering options analysis to assess the value of flexibility in mitigating cost and reliability risk. This framework is applied to three applications. Chapter 3 evaluates the potential for modular desalination design to manage multiple, diverse uncertainties — streamflow, demand growth, and the cost of water shortages — in Melbourne, Australia. Chapter 4 addresses uncertainty in groundwater resources in desalination planning in Riyadh, Saudi Arabia, and Chapter 5 addresses model uncertainty in climate change projections in a dam design problem in Mombasa, Kenya. Across all three applications, we find value in flexible infrastructure planning with a 9-28% reduction in expected cost. However, the performance of flexible approaches compared to traditional robust approaches varies considerably and is influenced by technology choice, economies of scale, discounting, the presence of irreducible stochastic variability, and the value society places on water reliability.

Kenneth Strzepek Research Scientist Thesis Supervisor

Olivier de Weck Professor of Aeronautics and Astronautics and Engineering Systems Doctoral Committee Chair

Dara Entekhabi

Bacardi and Stockholm Water Foundations Professor Doctoral Committee Member

James Wescoat

Aga Khan Professor of Architecture Doctoral Committee Member

Afreen Siddiqi Research Scientist Doctoral Committee Member

Acknowledgments

I am deeply thankful for the many mentors, collaborators, friends, and family who have supported and inspired my work over the past four years.

First and foremost, thank you to my advisor Kenneth Strzepek for his enormous investment of time and energy in my growth as a researcher. Ken provided invaluable guidance and support while always giving me the space to explore my interests freely. A true *doktorvater* in spirit, Ken has treated me like a member of his own family. His dedication to the service of others will serve as a model throughout my career.

Thank you to my committee chair Olivier de Weck for his guidance and insights. Oli's contributions were essential to the development of the conceptual framework presented in this dissertation. I am grateful that he pushed me to do more than I thought was possible.

Many thanks to my committee members Dara Entekhabi, James Wescoat, and Afreen Siddiqi. Their diverse and creative insights have greatly improved this work. I appreciate their generosity and many efforts to support my professional development. I would also like to thank the many other mentors and teachers at MIT and UPenn who have guided and inspired my path. I am especially grateful to Gene Mele, Frank O'Sullivan, Ernie Moniz, Frank Field, Ken Oye, Joe Sussman, Lisa D'Ambrosio, Noelle Selin, and Richard de Neufville.

This work was made possible through the efforts of many dedicated collaborators including Megan Lickley, Marco Miotti, Jaichander Swaminathan, Magdalena Klemun, Adnan Alsaati, and Abdulaziz Alhassan. More broadly, the communities at IDSS, the Joint Program, CCES, and SERG have provided a productive and inspiring research environment. I am grateful to the always helpful administrative staff of these organizations, and especially to Beth Milnes.

I am grateful for financial support from the NSF Graduate Research Fellowship, the Center for Complex Engineering Systems at KACST and MIT, and the J-WAFS Rasikbhai L. Meswani Fellowship for Water Solutions.

Thank you to the many friends at MIT and beyond who have stood by my side through the successes, failures, and adventures of my graduate career. I am especially thankful for the amazing women of TPP and ESD, many of whom I count among my dearest friends. I am inspired everyday by their work, commitments to justice, and willingness to always lift each other up.

Finally, it is with deep gratitude that I thank my family for their unwavering love and support. My husband Bob is a true partner. I am grateful for his love, generosity, jokes, and feminism. Thank you to my twin brother Zach; I am excited our career paths are converging once again. I am ever thankful for my chosen family, Sarah and Jenaté. Many thanks to my dad Ed for inspiring my love of math and science and for his programming guidance.

This dissertation is dedicated to my mother Regina. Mom, you have given more of yourself to me than I can imagine giving to another person. This achievement belongs to you as well.

Contents

1	Intr	roduction	15
	1.1	Motivation and background	15
	1.2	Recent methods for water supply planning under uncertainty	17
2	Wat	ter supply infrastructure planning under multiple, diverse uncertain-	
	ties	: A differentiated approach	25
	2.1	Uncertainty classification and analysis method	27
	2.2	Integrated modeling and risk profiles	36
3	Des	alination planning under multiple, diverse uncertainties during Mel-	
	bou	rne's Millenium Drought	41
	3.1	Introduction	42
	3.2	Modeling approach	46
	3.3	Application to urban water supply planning in Melbourne, Australia	47
	3.4	Results	52
	3.5	Discussion	58
4	Ma	naging predictive groundwater uncertainty in Riyadh, Saudi Arabia	L
	usir	ng Bayesian updating and flexible planning	63
	4.1	Introduction	64
	4.2	Modeling approach	67
	4.3	Application to Minjur aquifer in Riyadh, Saudi Arabia	71
	4.4	Results	77
	4.5	Discussion	85

5	Bay	esian updating of climate change uncertainty to enable flexible water	
	infra	astructure planning 9)1
	5.1	Introduction	<i>)</i> 2
	5.2	Modeling approach	96
	5.3	Application to Mwache dam)2
	5.4	Results)8
	5.5	Discussion	Ι7
6	Disc	cussion 12	21
	6.1	Methodological contributions	21
	6.2	Cross-application insights	23
	6.3	Limitations and extensions	26
A	Mul	tiple uncertainties in Melbourne 13	15
в	Gro	undwater uncertainty in Riyadh 14	1
С	Clin	nate change uncertainty in Mombasa 14	15

List of Figures

1-1	Literature review and thesis roadmap
2-1	Planning framework schematic
2-2	Uncertainty matrix (Walker et al. 2003)
2-3	Bayesian inference and SDP schematic
2-4	Integrated modeling approach schematic
3-1	Melbourne study area
3-2	Bagplots of cost and reliability risk in Melbourne
3-3	Risk ranking of alternatives in Melbourne
3-4	CDFs of total cost by infrastructure alternative in Melbourne 57
3-5	Sensitivity analysis in Melbourne
4-1	Groundwater method schematic
4-2	Riyadh study area and MODFLOW grid
4-3	Hydrostratigraphic cross section
4-4	Bayesian updating of groundwater resource uncertainty in Riyadh
4-5	Expansion policy as function of hydraulic head
4-6	Histogram of desalination expansion decisions in Riyadh $\ . \ . \ . \ . \ . \ . \ . \ . \ . \ $
4-7	CDF of total cost by infrastructure alternative in Riyadh
4-8	Sensitivity analysis in Riyadh
5-1	Climate change method schematic
5-2	Mombasa study area
5-3	Climate change CIs comparing democratic and Bayesian methods 109
5-4	Bayesian updating of climate change uncertainty over time

5-5	Optimal infrastructure policy for Mombasa
5-6	Histogram for infrastructure expansion decisions in Mombasa $\ldots \ldots \ldots \ldots 115$
5-7	CDF of total cost by infrastructure alternative for Mombasa
A-1	Melbourne reservoir operations
A-2	Melbourne decision tree
B-1	ANN error histogram
B-2	Riyadh SDP planning assumptions
C-1	CLIRUN II model schematic
C-2	CLIRUN II calibration fit
C-3	Sample reservoir operations and yield
C-4	Alternate Bayesian updating of climate change uncertainty

List of Tables

2.1	Classification of key water planning uncertainties
3.1	Melbourne uncertainty classification
3.2	Infrastructure alternatives modeled in Melbourne
5.1	Infrastructure alternatives and planning scenarios in Mombasa
6.1	Methodological contributions summary
6.2	Key opportunities for extensions and future work
A.1	Parameter assumptions for water system moel
C.1	Climate model ensembles used

Glossary

ANN	artificial neural network.
ARMA	auto-regressive moving average.
BMA	Bayesian model averaging.
°C	degrees Celsius.
CDF	cumulative density function.
CI	confidence interval.
CMIP5	coupled model intercomparison project, phase
	5.
CRU	Climate Research Unit.
DAPP	dynamic adaptive policy pathways.
ENSO	El Niño–Southern Oscillation.
EOA	engineering options analysis.
GCM	general circulation model.
IPCC	Intergovernmental Panel on Climate Change.
K	hydraulic conductivity.
kL	kiloliters.
km	kilometer.

kNN k nearest neighbors.

m	meter.
m.a.s.l.	meters above sea level.
m.b.l.s.	meters below land surface.
M\$	million US dollars.
$\mathrm{m}^{3}/\mathrm{d}$	cubic meters per day.
MAR	mean annual runoff.
MCM	million cubic meters.
MCM/d	million cubic meters per day.
MCMC	Markov chain Monte Carlo.
mm	millimeters.
MORDM	many-objective robust decision making.
Р	precipitation.
RCP	representative concentration pathway.
RDM	robust decision making.
RO	reverse osmosis.
ROF	risk of failure.
S	storativity.
SDP	stochastic dynamic programming.
Т	temperature.
USGS	United States Geological Survey.

Chapter 1

Introduction

1.1 Motivation and background

Australia faced its driest decade in recorded history from 1997 to 2009. Known as the "Millennium Drought," this period had severe impacts on Australia's economy, environment, and citizens. The agriculture sector was especially damaged, with low crop yields forcing small farms out of production and mid-size farms into financial distress [4]. Salinity in the River Murray reached record levels, impacting water supplies for human and livestock consumption and damaging ecosystems [38]. The city of Melbourne was among the hardest hit. The utility Melbourne Water and the Victorian government implemented a host of measures to increase supply and reduce demand in the city, including increased water tariffs, wastewater recycling, water use restrictions, and rainwater harvesting installations [97]. Despite these measures, reservoir levels in 2007 reached below 30%, or less than one year's worth of supply [105]. This prompted Melbourne Water to contract the Wonthaggi desalination plant, a 150 million cubic meters (MCM) per year reverse osmosis (RO) plant with a capital cost of \$5 billion [58]. However, the drought ended in 2009, and intense rains filled the reservoir system before the Wonthaggi plant was completed in 2012 [97]. This left the \$5 billion asset completely idle for the next five years, sparking substantial public controversy. The Wonthaggi plant has been called "the most criticized desalination investment to have occurred in the last decade certainly in Australia, if not the world" [126].

However, the situation facing water planners in 2007 was dire. It is impossible to know how long a drought will last, and the El Niño–Southern Oscillation (ENSO) cycle commonly causes prolonged dry periods in the region. Similar cases of large desalination investments laying idle for years after completion have occurred in London [13], California [64], and Spain [22]. However, there are also plentiful examples of cities relying on desalination investments to weather droughts — or wishing they had made those investments earlier. In Perth, which was also severely impacted by the Millennium Drought, planners invested in a smaller (45 MCM/y) RO plant that was completed in 2006 at the peak of the drought and heralded as a success [126]. Perspectives are shifting in Melbourne, where a new dry period led Melbourne Water to use the Wonthaggi plant for the first time last year [1]. A longmothballed desalination plant from the 1990s in Santa Barbara, California was retrofitted during the recent drought in 2015 [64]. Finally, as ongoing drought in Cape Town, South Africa fuels debate on whether 2018 will bring "Day Zero" when water supplies in the city completely run out, water managers are scrambling to bring desalination capacity online quickly [27]. Researchers at the University of Melbourne predict that the controversial Wonthaggi plant could save Melbourne from a "Day Zero" of its own [155].

These examples highlight the challenge of planning water supply infrastructure for an uncertain future. Water planners must ensure reliable, high-quality access to freshwater for domestic, agricultural, and industrial end-users while meeting financial and institutional constraints and preserving the sustainability of natural resources. Over-investment in infrastructure can lead to stranded assets worth hundreds of millions or billions of dollars and unnecessary environmental impacts. Under-investment can lead to supply restrictions or outages with consequences for the economy, environment, and human health. This challenge is exacerbated in many regions of the world by the confluence of growing populations and economies with a changing climate.

Water planners rely on forecasts of supply and demand that are inherently uncertain. Variability in the hydrological cycle at varying scales — daily, monthly, annual, decadal — leads to uncertainty in surface water availability. Groundwater storage is also often affected by hydrological variability, often with greater lag times than surface water. Further, the substantial heterogeneity of groundwater aquifers in combination with the difficulty in measuring groundwater flow leads to additional uncertainty in groundwater resources. Over long time scales, climate change is expected to alter the hydrological cycle, rendering inappropriate the use of stationary stochastic processes to characterize uncertainty — and making water resources planning and management all the more difficult. [108]

Social and economic factors also present uncertainty. In addition to uncertainty in how

the climate system will respond to greenhouse gas emissions, political and institutional decision-making at regional, national, and international scales drives uncertainty in the magnitude and timing of greenhouse gas emissions. However, in many regions of the world, population growth, economic growth, and urbanization are expected to have a greater impact on water scarcity than climate change [164]. Finally, uncertainty is compounded by ambiguity [20], also known as evaluative complexity [43], or differing preference across stake-holders about the relative importance of water reliability, ecological preservation, cost, and sustainable resource use for future generations.

Water managers have a number of tools for managing uncertainty and variability in water resource systems. Demand policies, incentives, and restrictions can reduce end-user demand during times of low supply. Existing infrastructure, such as reservoir systems, pipelines, desalination plants, and wastewater treatment for reuse can be managed and operated to smooth variability in supply and hedge against future uncertainty. Coordination with neighboring water suppliers can be used to import additional supplies when reserves are low. New infrastructure can be added to address immediate drought conditions or meet long-term needs. However, new water infrastructure faces additional challenges from uncertainty. New infrastructure requires large capital investments often on the order of hundreds of millions or billions of dollars. These investments have long lifetimes: desalination plants and dams have lifetimes on the order of 30 years and 100 years respectively. Infrastructure projects are typically fixed, static investments that are difficult to change once built. However, they interact with complex and evolving human and natural systems over their lifetimes [35]. The confluence of these factors heightens the risk that water supply infrastructure investments will fail to meet performance goals in the face of uncertainty — and necessitates the development of new approaches to reduce and manage risk.

1.2 Recent methods for water supply planning under uncertainty

Historically, uncertainty in water infrastructure planning has been managed by adding a safety factor to a single best forecast [149]. Dams were sized using historical streamflow data, without accounting for the possibility that the future could bring more or different variability than the past. Since the 1980s, risk-based metrics such as reliability and vulnerability have

been promoted for managing water supply systems [67]. Scenario-based planning, in which planners develop a few alternate visions of the future and choose a set of strategies that address the potential for each scenario to occur, is now widely used in water infrastructure planning. Climate change has also driven the use of adaptive management, in which planners prepare to adapt as uncertainties unfold over time, over the past two decades [115].

In the past 15 years, more sophisticated approaches for water supply planning under uncertainty have developed. Three critical insights in this literature include: 1) explicit recognition of the limitations of forecasts and therefore the need to develop strategies that are relatively insensitive to forecasts; 2) the need for bottom-up vulnerability analysis, in which the conditions that lead to system failure are identified and evaluated, and 3) the potential for flexible or adaptive approaches to manage uncertainty.

The first insight is addressed by robust decision making (RDM), developed by Lempert et al. (2006) [93] with early applications in water planning [60, 92]. RDM uses a scenariobased approach to identify robust strategies that meet performance criteria across many possible futures. This approach is motivated by the concept of "deep" uncertainty in which there is so much uncertainty that probabilities cannot or should not be assigned to the various potential outcomes [166]. This is distinct from "recognized ignorance" or Knightian uncertainty [88], in which not even all the possible outcomes can be identified [166]. Groves and Lempert (2007) argue that climate change is a deep uncertainty and therefore that traditional probabilistic decision-analytic methods are inappropriate [60]. Many-objective robust decision making (MORDM) has extended RDM to include many performance criteria relevant for water planners and uses multi-objective evolutionary algorithms to efficiently identify Pareto optimal strategies in large scale systems [86].

Second, bottom-up vulnerability analysis has played an important role in water supply planning under uncertainty. In the info-gap approach developed by Ben-Haim (2001), the analysis develops increasingly large multidimensional uncertainty sets and identifies the solutions that meet threshold performance criteria for each uncertainty set [14]. This allows planners to identify uncertainty thresholds beyond which a certain planning strategy fails. More recently, decision scaling, developed to address climate change uncertainty specifically, links bottom-up vulnerability analysis with climate change projections [19]. Decision and performance thresholds are identified first and then combined with ensembles of general circulation model (GCM) projections, allowing planners to both make use of GCM projections and also understand their limitations.

Finally, a number of approaches have built on early calls for adaptive management as a strategy to manage climate change uncertainty. Adaptive management can enable reliability at reduced cost by developing plans to ensure short-term water needs are met without investing in permanent solutions before the long-term future is well understood. Recent methods such as adaptation tipping points [62] and dynamic adaptive policy pathways (DAPP) [61, 89] enable policymakers to identify tipping points or thresholds beyond which new polices will be needed. Identifying these thresholds in advance — and the policies to be implemented if they are reached — enables a planned adaptation approach in which adaptive policies are fully developed and can be nimbly executed when needed, reducing short-term transition risks [102]. Defining "dynamic robustness" as flexibility enabling a plan to change in response to changing conditions over time, Walker et al. (2013) raises the limitations of RDM in developing dynamic robust solutions and highlights the value of DAPP in achieving dynamic robustness [165].

A different approach to adaptive or flexible planning is used in engineering options analysis (EOA). EOA is an approach related to real options analysis, which applies financial valuation of options to capital investment decisions [33]. EOA focuses on assessing engineering options, or flexibility in engineering design and planning, using related methods. Flexible options can be identified using screening models, stochastic optimization, or simple decision rules [11]. In contrast to real options analysis, EOA focuses on exploring and weighing the various benefits and costs using uncertainty analysis rather than developing precise estimates of the value of flexible options [34]. Therefore, while it uses probabilistic approaches to assess uncertainty, the interpretation of the probabilistic results can be adapted to address deep uncertainties. This method has been applied to infrastructure domains ranging from satellite systems [35] to liquefied natural gas production [23]. In water resources, early applications have applied EOA to infrastructure planning problems in desalination [51], dams [80], and hydropower [153].

Research questions

This thesis addresses three main gaps in the above literature and develops a novel framework for planning under uncertainty to address them. First, while the recent methods addressing deep uncertainty, such as RDM, have played an important role in acknowledging the limitations of probabilistic forecasts and developing planning approaches that are relatively insensitive to these limitations, many of the common uncertainties that impact water supply infrastructure planning can be treated probabilistically. RDM typically addresses all the uncertainties using scenario analysis. However, uncertainties can be addressed in the same analysis using different methods. Previous work in dimensions of uncertainty — which differentiate deep uncertainty from statistical uncertainty, for example — can serve as a basis to classify uncertainties and map them to appropriate analysis tools.

Second, another limitation in existing approaches is a focus on static rather than dynamic approaches to characterizing climate and hydrological uncertainty in infrastructure planning. The scenario approach in RDM, for example, assesses the performance of a planning alternative within a single scenario, assuming the conditions of that scenario extend over the entire planning horizon; the uncertainty is addressed by comparing many different scenarios. This approach does not allow for updated assessment of uncertainty within a single scenario to drive adaptive strategies in which infrastructure plans or designs can be changed. Other approaches described above, such as DAPP and EOA, do take a dynamic approach to uncertainty that enables flexible or adaptive approaches. However, DAPP is focused on policy rather than infrastructure planning; it does not provide a framework to develop and assess the performance of flexible infrastructure planning and design, which face additional challenges due to the large capital investments required and long lifetimes. EOA does provide the necessary framework to evaluate flexible infrastructure planning and performance. However, it does not currently integrate models of the natural environment to characterize hydrological or climate uncertainty. The research gap, therefore, lays in developing a planning approach that integrates all three characteristics: a dynamic approach to uncertainty, infrastructure planning and performance, and appropriate physical models.

Finally, existing approaches do not provide a framework for evaluating the value of flexibility in managing hydrological uncertainty. While some recent approaches to water supply infrastructure sequencing include flexibility as an objective in a multi-criteria decision problem [12], these approaches do not provide a basis for assessing whether flexibility is a worthwhile objective. Flexibility is a life-cycle system property of an engineering system that can be useful in achieving performance goals, not a performance metric itself [35]. Flexibility has a cost: either physical infrastructure investments that must be made in order to enable flexible options in the future and/or institutional preparations to enable timely and efficient transitions once a decision is made to exercise an option. Comparing these costs to the potential benefits of flexibility is necessary in order to determine whether flexible approaches are worth implementing.

This leads to the three overarching research questions addressed in this dissertation:

- 1. How do multiple, diverse uncertainties impact water supply infrastructure planning?
- 2. How can Bayesian updating of hydrological uncertainty be used to develop dynamic uncertainty estimates for flexible planning?
- 3. How valuable is flexibility in managing diverse uncertainties?

This dissertation addresses these questions through the development of a new framework for water supply infrastructure planning under uncertainty and its application to three contrasting planning problems. The planning framework addresses question 1) by using existing dimensions of uncertainty as well as a new dimension in order to classify uncertainties and connect them to appropriate uncertainty analysis tools so that multiple, diverse uncertainties can be addressed in a differentiated approach in a single analysis. Question 2) is addressed using Bayesian inference applied to climate and hydrological models to develop dynamic assessments of uncertainty, which are embedded in a multi-stage stochastic planning approach to develop policies for exercising flexible infrastructure options. Statistical surrogate models are used as needed to enable computational tractability. Finally, the planning framework draws on engineering options analysis to assess the value of flexibility in mitigating hydrological uncertainty and address question 3).

The remainder of this dissertation is organized as follows. Chapter 2 introduces the framework for flexible water supply infrastructure planning under multiple uncertainties. Chapter 3 applies the framework to a retrospective desalination planning problem based on the Millennium Drought in Melbourne, Australia. The focus of this application is in addressing question 1) by integrating three different uncertainties with different classifications into a single analysis. Chapters 4 and 5, in contrast, focus on addressing question 2) by developing approaches for applying Bayesian inference to hydrological and climate models in order to characterize the transition probabilities in a stochastic dynamic program. Chapter 4 addresses parameter uncertainty in groundwater availability in a desalination planning problem in Riyadh, Saudi Arabia, and Chapter 5 addresses model uncertainty in climate





Figure 1-1: a) Key literature areas with prior work and gaps addressed in this dissertation and b) novel planning framework with key contributions (top) and applications to demonstrate those contributions (bottom).

change projections in a dam design problem in Mombasa, Kenya. All three applications address question 3) by assessing the value of flexible planning and design in comparison to static, robust approaches. Finally, Chapter 6 concludes with discussion of insights across the individual applications and potential for future development of the planning framework. Chapters 3, 4, and 5 are presented as stand-alone papers; while there is some overlap in literature review and methods across the chapters, this approach allows them to be read and used in isolation without reviewing the dissertation as a whole.

Chapter 2

Water supply infrastructure planning under multiple, diverse uncertainties: A differentiated approach

The previous chapter provided an introduction to the uncertainties that commonly impact water supply infrastructure systems and the tools available to manage those uncertainties in planning decisions. Research over the past 15 years has made important advances in 1) developing approaches that enable infrastructure planning strategies that are relatively insensitive to long-term uncertainty and 2) emphasizing the necessity of bottom-up vulnerability assessments as a complement to top-down planning approaches. Both of these advancements are important in addressing uncertainties that are poorly characterized or otherwise difficult to predict. However, they have also led to a tendency to address all uncertainties in a problem using the same approach. A variety of modeling tools and management strategies exist to address uncertainties; different approaches may be more effective for one specific uncertainty than another.

Here we present a framework for modeling and managing multiple, diverse uncertainties within a single analysis. The goal of this framework is to address three key limitations in current approaches for water supply planning under uncertainty. First, we take a differentiated approach to uncertainty analysis in which different types of uncertainty are addressed with unique, appropriate methods. There are many existing dimensions of uncertainty and uncertainty analysis methods. However, less attention has been paid to matching specific



Figure 2-1: Framework for classifying, modeling, and managing multiple diverse uncertainties in water supply infrastructure planning.

types of uncertainty to analysis or management tools. The methods that have developed to address deep uncertainty in water resource system planning, for example, tend to treat all uncertainties in the analysis as deep rather than separating out the deep uncertainties from the statistical ones. Second, when necessary we take a dynamic approach to uncertainty in which probabilistic estimates are updated in a Bayesian manner over time. Our planning model uses hypothetical future observations to update forecasts, allowing us to account for the potential to learn in the future in planning decisions today. Finally, we assess the value of flexibility in water supply infrastructure planning and design; current approaches either do not consider flexibility or include it as a planning objective without knowing whether it will be worthwhile in achieving performance goals.

Our framework for water supply infrastructure planning under multiple uncertainties comprises three main steps, illustrated in Figure 2-1. First, uncertainties are classified according to two dimensions: the level of uncertainty and learning potential. Analysis methods are then chosen based on this classification. Second, a modeling approach is applied in which infrastructure plans or designs are developed and integrated into a water resource system model that comprises a hydrological model of the region of interest and the impact of infrastructure on that system. This model is embedded in a multi-stage stochastic planning model, such as a stochastic dynamic programming (SDP). Finally, the result of this is to develop risk profiles for the different infrastructure alternatives that highlight areas of vulnerability and strength against key performance metrics. These three steps are discussed in more detail below.

2.1 Uncertainty classification and analysis method

The first step in the framework is to 1) classify uncertainties into three categories, and 2) apply a differentiated uncertainty analysis method appropriate for that category.

Uncertainty classification

Existing uncertainty dimensions. A diverse range of uncertainties impact water supply planning decisions. These include: short-term variability in rainfall and runoff, long-term climate change, uncertainty in groundwater availability, population growth and urbanization at uncertain rates, and the value of water during times of scarcity. Each of these uncertainties has unique characteristics and can be classified according to different dimensions. Several existing dimensions of uncertainty are presented in the "uncertainty matrix" developed by Walker et al. (2003) [166], updated by Kwakkel et al. (2010) [90], and illustrated in Figure 2-2. Here we summarize and extend the uncertainty matrix to use in our classification system.

The *level of uncertainty* as coined by Walker et al. (2003) [166] describes the magnitude of uncertainty ranging from determinism to recognized ignorance. These authors develop three levels: statistical uncertainty in which the full range of outcomes is known and probabilities can be reliably placed on these outcomes; scenario uncertainty in which the possible outcomes can be identified but they can not be characterized probabilistically; and recognized ignorance in which the full range of possible outcomes is unknown. This refines a much earlier distinction between risk, which can be defined probabilistically, and Knightian uncertainty, which is immeasurable [88]. Similar distinctions have been drawn more recently, such as Luce and Raiffa (1957) [99] who distinguish between decision making under risk and decision making under uncertainty and Morgan and Henrion (1990) [112] who distinguish between uncertainties that can be described probabilistically and those that cannot. Most recently, Kwakkel et al. (2010) [90] further refine this dimension of uncertainty to improve clear operationalization with four categories.

One of the four categories defined by Kwakkel et al. (2010) is "deep uncertainty", defined as "being able to enumerate multiple alternatives without being able to rank order the alternatives in terms of how likely or plausible they are judged to be" [90]. The concept of deep uncertainty has indeed motivated many of the approaches for decision-making under uncertainty described in the previous chapter such as RDM or decision scaling. However, there is disagreement about how to conceptually define and operationalize this concept. Groves and Lempert (2007) [60] alternatively define deep uncertainty as "the situation where decision makers do not know nor cannot agree upon the system model that relates action to consequences, the prior probabilities on the inputs to the system model(s), or the value function that ranks the desirability of the consequences". Despite definitional disagreements, in practice most researchers have argued that deep uncertainty requires 1) the use of scenario-based or otherwise non-probabilistic approaches to uncertainty and/or 2) bottom-up approaches that focus on identifying areas of vulnerability rather estimating the probability of reaching them [93, 19, 14]. However, a different school of thought used in engineering options analysis uses probabilistic approaches as a means to sample from a wide distribution of possible outcomes; the interpretation of the results is therefore not focused on precise calculations of expected value but rather on highlighting trade-offs between alternatives [34].

The *location of uncertainty* refers to where in the analysis or modeling process the uncertainty occurs including the input data, parameters, model, or interpretation of the results [166]. Uncertainty in input data, for example, recognizes the limitations and inherent error in the way measurements of hydrological processes like evaporation, streamflow, and hydraulic head are made. The distinction between model and parameter uncertainty is increasingly common in practice, with the use of multi-model comparison tools such as Bayesian model averaging (BMA) applied to estimate uncertainty across models [74]. This distinction can, however, be blurry when it is unclear what aspects of the analysis are "parameters" and which are components of the "model". For example, groundwater withdrawals and recharge are typically formulated as boundary conditions in numerical groundwater models, and boundary conditions are typically considered part of the model structure. However, withdrawals and recharge rates are also often treated as parameter uncertainty. In practice, this distinction is becoming less important as methods such as Bayesian hierarchical modeling are increasingly used to integrate parameter uncertainty into estimates of model uncertainty [140]. The context or interpretation of the results can also be viewed as an uncertainty. Some researchers describe this as "ambiguity" [20] or "evaluative complexity" [43]. While the uncertainty matrix addresses it as a location of uncertainty [166], others have suggested it as a

			Level		Na	ature
Lo	ocation	Statistical uncertainty	Scenario uncertainty	Recognised ignorance	Epistemic uncertainty	Variability uncertainty
Context	Natural, technological economic, social and political, representation					
Model	Model structure					
	Technical model					
Inputs	Driving forces					
	System data					
Paramete	ers					
Model O	utcomes					

Figure 2-2: Uncertainty matrix reproduced from Walker et al. 2003 [166]

different nature of uncertainty (see next paragraph) in comparison to epistemic or aleatory [20].

The *nature of uncertainty* refers to whether the uncertainty arises from natural stochastic variability or from a lack of information [166]. These two categories are often called aleatory and epistemic uncertainty, respectively. Natural variability arises through some kind of stochastic process: a random variables takes on different values in time or in space. Rainfall is a classical example of natural variability in hydrology. Many parametric processes, such as auto-regressive lag-1 processes or other Markov processes [138], as well as nonparametric processes, for example using re-sampling of historical observations [176] or bootstrapping methods [91], have been proposed as models of the stochastic process governing daily or monthly precipitation. Soil properties such as hydraulic conductivity or porosity exhibit natural variability in space rather than time; random fields are often used to represent this variability [175]. In contrast, epistemic uncertainty arises from a lack of information. For example, the population of the city of Melbourne at the start of 2050 will have a single value, but we do not have access to that information and therefore epistemic uncertainty arises.

While the difference between aleatory and epistemic uncertainty has a long history of use, the distinction between them is not always clearly defined and indeed many uncertainties have characteristics of both. For example, rainfall, as described above, clearly exhibits variability due to a stochastic process. However, precipitation in Cambridge, MA on May 1, 2020 will take on a single value, but we do not currently have the information to identify it precisely, only a statistical description of the underlying process. It therefore has properties of epistemic uncertainty as well. This dual interpretation of uncertainty has a long history in water resources. Dettinger and Wilson (1981) [40] present a statistical description of aquifer hydraulic conductivity using a mean and spatial covariance with two interpretations. In the first, the statistical description represents the underlying stochastic process that leads to spatial variability in the aquifer. In the second, it represents an estimation of the true field and the modeler's degree of belief in that estimate. These interpretations reflect the aleatory and epistemic aspects of uncertainty, respectively. Dettinger and Wilson (1981) [40] highlight an important distinction, however: aleatory uncertainty is irreducible while epistemic uncertainty can be reduced. An aquifer's hydraulic conductivity will always have a pattern of spatial variability, while our description of it can be more and more precise. In practice, however, probability theory is widely applied to both types of uncertainty even though the interpretation is quite different.

Our uncertainty classification approach. We classify uncertainties according to two dimensions; the "level" dimension above and a new dimension we call "learning potential". The dimensions of uncertainty can be understood as continua. For example, the level of uncertainty ranges from determinism at one end to recognized ignorance at the other. However, mapping uncertainties to uncertainty analysis methods requires discrete categories as the analysis methods (e.g. scenario analysis or Monte Carlo simulation) are themselves discrete. We therefore split the dimensions into discrete categories with operational definitions to reliably code the uncertainties.

The first dimension we use is the level of uncertainty. We address two discrete categories in this dimension: statistical uncertainties and deep uncertainties. We define statistical uncertainties as those that can be appropriately characterized probabilistically and validated using historical data. We define deep uncertainties as those for which the possible outcomes can be identified but probabilities cannot reliably be placed on them; or those probabilities cannot be validated. Recognized ignorance, in which the outcomes cannot be identified, is not addressed. This classification is important for choosing an uncertainty analysis method because it determines whether it is appropriate to use probabilistic uncertainty methods or not.

The second dimension of uncertainty used for classification is a new dimension herein termed learning potential. It is related to the idea of information or epistemic uncertainty, but it reflects pragmatically whether information uncertainty can be reduced. We define two categories along this dimension: high and low learning potential. Uncertainties with high learning potential are defined as those for which new observations can be collected, either presently or in the future as new observations become available, to reduce or update uncertainty in a way that meaningful impacts for easts for decision-making. Low learning potential uncertainties are those which cannot be meaningfully updated, either because new information cannot be feasible collected or because the new information does not meaningfully change previous forecasts. This dimension has two advantages over the traditional epistemic vs. aleatory definition. First, it does not suffer from the problem that, as described above, epistemic and aleatory are not mutual exclusive categories and in fact most uncertainties can likely be considered as having elements of both. For this new dimension, there are either opportunities to reduce or update the uncertainty, or not. Second, it is connected to an important choice in how we model uncertainty: whether we assess uncertainty statically or dynamically in which estimates of uncertainty can change over time. The distinction between epistemic and aleatory uncertainty often does not inform a choice of uncertainty analysis tool; probabilistic approaches are widely applied for both.

Together, these dimensions define three categories: deep uncertainties (which includes both high and low learning potential uncertainties), high learning-potential statistical uncertainties, and low learning-potential uncertainties. Common uncertainties impacting water supply infrastructure planning are listed in Table 2.1 along with their classification and analysis tool used.

Table 2.1: Key uncertai	nties in water sup	ply infrastructure planning with classific	ation, justification, and analysis tool
Uncertainty	Classification	Justification	Analysis Tool
Short-term variability in streamflow	Statistical; low learning potential	Well-characterized by historical data; stationarity assumption appropriate over short time scales	Monte Carlo simulation; applied using stochastic streamflow generation
Population growth rate	Statistical; high learning potential	Well-characterized by existing demographic forecasts; early changes in growth rate highly predictive of future growth	Multi-stage stochastic planning; dynamic uncertainty uses simple conditional growth rate assumptions
Water shortage penalty value	Deep uncertainty	This is an ambiguity, or uncertainty arising from differences in stakeholder perspective	Scenario analysis
Parameter uncertainty in groundwater	Statistical; high learning potential	Characterized probabilistically in previous hydrogeological studies; can be improved substantially with addition head observations	Multi-stage stochastic planning; dynamic uncertainty uses Bayesian inference with a groundwater model to update parameter estimates
Model uncertainty in climate change	Statistical; high learning potential	Validated statistical model exists for climate model uncertainty; Uncertainty changes as observations of long-term trends become available	Multi-stage stochastic planning; dynamic uncertainty uses Bayesian model averaging, updated as new observations are available
Scenario uncertainty in climate change	Deep uncertainty	Greenhouse gas emissions over long-time scales are influenced by global geopolitics	Scenario analysis

Uncertainty analysis methods

Next, we map the three uncertainty categories to uncertainty analysis methods. We apply probabilistic approaches for statistical uncertainties and scenario-based approaches for deep uncertainties. This allows us to take advantage of the full information available for statistical uncertainties without presenting deep uncertainties with over confidence. For deep uncertainties, a variety of scenario-based methods could be used. For example, the Latinhypercube simulation approach to scenario generation developed in RDM could be applied. In our framework, this approach would only be used for deep uncertainties, in contrast to RDM which applies this approach to all uncertainties. In our applications, we simply repeat the analysis under different input assumptions or scenarios to assess the impact of uncertain assumptions on the risk profiles of different alternatives.

For statistical uncertainties, we apply two different uncertainty analysis approaches for high and low learning potential uncertainties. Low learning potential statistical uncertainties are addressed using Monte Carlo simulation. This takes advantage of available statistical information in a computationally efficient manner.

High learning potential uncertainties, however, require a dynamic approach to uncertainty in which estimates of future uncertainty can be updated over time. Existing methods, for example the risk of failure (ROF) method developed by [123], allow uncertainty estimates to vary as a function of the state of the system but not as a function of time independent of the state of the system. For example, predictions of reliability outages may be higher in time period 5 than in time period 2 because reservoir storage levels are lower in time period 5; this is addressed under the ROF approach. However, the probability of reliability outages in time period 5 could be different than in time period 2 even if the reservoir storage level is the same because the planner's understanding of the hydrological system has changed; this is not addressed in the ROF approach. This is an important limitation. We therefore develop a dynamic approach to characterizing uncertainty so that it can be updated to reflect new or improved information.

We use Bayesian inference applied to a physical hydrological or climate model to characterize dynamic uncertainty estimates and embed these in a multi-stage stochastic dynamic program. SDP is a multi-stage stochastic optimization approach used to develop optimal policies as a function of the system state and time period [128]. It takes into account future uncertainty using transition probabilities $p(S_{t+1}|S_t)$ which describe the 1-step transition from a state in time period t to a new state in t+1. Non-stationary transition probabilities, in which $p(S_{t+1}|S_t = o)$ does not necessarily equal $p(S_{y+1}|S_y = o) \forall t \neq y$, are used to reflect learning about the system over time. The optimal policies are derived by solving the Bellman equation, shown in Equation 2.1, using recursion.

$$V(s,t) = \underset{a \in A}{\operatorname{argmin}} C(s(t), a(t), t) + \gamma \sum_{s \in S} p(s(t+1) | s(t), a(t)) * V(t+1, s(t+1))$$
(2.1)

where V is the optimal policy; s is the state of the system from the state space S; t is the time period or stage; a is an action from a set of possible actions A; C is the singleperiod cost as a function of the current state, action, and time; γ is the discount rate; and p(s(t+1) | s(t), a(t)) are the transition probabilities as a function of the current state, action, and time.

The challenge in addressing hydrological uncertainties using SDP lies in using highfidelity physical models to characterize the transition probabilities. An overview of our approach is shown in Figure 2-3. First, the state space S is defined and discretized to represent the uncertain hydrological variable(s) of interest. The groundwater application in Chapter 4 uses hydraulic head in a groundwater aquifer as the uncertain state variable; the climate change application in Chapter 5 uses decadal mean temperature and precipitation as the uncertain state variables. A planning horizon is also defined and discretized into time steps; the length is chosen to reflect the approximate lifetime of the relevant infrastructure.

Second, a physical model that forecasts the uncertain state variable is developed or selected. Chapter 4 uses the groundwater model MODFLOW, and Chapter 5 uses an ensemble of GCM projections. Optionally, a statistical surrogate model can be developed based on the physical model to enable or improve computational tractability. In Chapter 4, an artificial neural network is trained on MODFLOW output to serve as a surrogate. The physical model is used to develop a prior distribution $p(S_t)$. This is operationalized differently depending on the location of uncertainty (e.g. parameter vs. model uncertainty: see description in Section 2.1). For parameter uncertainty, Monte Carlo simulation on the uncertainty parameter can be applied; this is done in Chapter 4 for parameter uncertainty in hydraulic conductivity and storativity. Bayesian model averaging [74] can be applied to address model uncertainty;

Integrating Bayesian updating with physical model into SDP

1) Define and discretize state space



2) Develop or identify physical model; use to develop prior distributions



3) Apply Bayes to physical model to develop posteriors



4) Solve SDP using posteriors as transition probabilities



Figure 2-3: Schematic demonstrating the method for integrating probabilistic forecasts for uncertain hydrological variables into an SDP using Bayesian inference on a physical model this is used in Chapter 5 to address model uncertainty, or disagreement across models with different structures, in climate change projections.

Third, Bayes' Theorem, shown in equation 2.2, is applied to update the prior distributions developed in step 2 with a new observation. The new observation corresponds to a specific state in the hydrological state space defined in step 1. The reasoning is if the system reaches a state s at time t, then s will be an observation about the system state that can be used to update future predictions. For example, say mean decadal temperature is 26°C at the beginning of the planning period with an expected value of 1.5 °C increase after 40 years. If after 40 years, 2.5 °C of warming are observed, this suggests a higher than expected rate of warming. Bayesian updating allows us to use this observation to update the original forecast to reflect faster warming. As with the prior, the specific implementation varies depending on whether parameter or model uncertainty is addressed; see details in Chapters 4 and 5 respectively. The updating is performed for each feasible state in each time period, with the state corresponding to a hypothetical, future observation.

$$p(S_{t+1}|O) = \frac{p(O|S_{t+1})p(S_{t+1})}{p(O)} \ O \in S$$
(2.2)

Finally, the posteriors are embedded into the SDP as the transition probabilities, with each transition probability $p(S_{t+1}|S_t)$ equal to the corresponding posterior $p(S_{t+1}|O)$ where $O = S_t$. The Bellman equation is then solved using recursion. This process can be understood as a decision tree, as depicted in Figure 2-3. In each time period, a new observation is made, the transition probabilities updated to reflect an updated forecast, and then an action chosen for that time period. Although the transition probabilities reflect 1-step transitions from one time period to the next, the recursion enables optimal policies to be developed that take into account uncertainty over the entire planning horizon. Further, the use of the Bayesian updated posteriors as transition probabilities enables the optimal policies to account for the potential for learning as new observations are collected over time.

2.2 Integrated modeling and risk profiles

Once the uncertainties are classified and the corresponding analysis tool selected, they are incorporated into an integrated modeling approach depicted in Figure 2-4. This analysis is completed in four main steps. First, a hydrological or climate model is used to characterize


Figure 2-4: Schematic demonstrating the method for integrating probabilistic forecasts for uncertain hydrological variables into an SDP using Bayesian inference on a physical model.

non-stationary transition probabilities in an SDP; this step is described in detail in the previous section. Second, flexible infrastructure plans or designs are developed as alternatives to traditional static, robust approaches. Third, the impacts of different infrastructure alternatives on a water resource system are evaluated to characterize the SDP costs and actions. Finally, the optimal policies developed by the SDP are used to simulate and compare the performance of the different alternatives against key performance metrics. These steps are described below.

Flexible infrastructure alternatives

Flexible infrastructure plans or designs are developed as alternatives to traditional static plans; these are tailored to the planning problem at hand. Flexibility in infrastructure can take a variety of forms. Flexibility in planning can include flexibility in the timing, location, or type of infrastructure brought online. For example, a staged development approach in

which capacity is added modularly in stages rather than all at once can trade off cost savings due to economies of scale in large infrastructure projects to prevent over building capacity. Flexibility can also be designed into physical infrastructure itself so that it can be more easily retrofitted in the future. For example, designing a dam that can be increased in height in the future enables flexibility in reservoir storage volume; this is assessed in Chapter 5. A RO desalination plant can be designed modularly so that extra RO modules can be added to increase capacity in the future; this is assessed in Chapter 3. In the applications in this dissertation, we consider flexibility in volume of capacity and in timing of capacity. In the surface water applications in Chapters 3 and 5, the planning question is focused on volume of capacity is needed in order to meet demand given uncertainty in supply; therefore, the designed flexibility is in the volume of capacity. In the groundwater application in Chapter 4, it is known that new capacity will be needed in the future and the question is when the investment will be needed in order to prevent reliability outages; in this case the flexibility is in the timing of capacity. For all of the applications, the flexible option to add additional infrastructure capacity is identified and designed up front. Then, the policy for if and when to exercise the option is determined by the SDP.

Water resource system model

In addition to the transition probabilities and state space definition, the SDP formulation requires definitions of the actions and costs. The actions can include both 1) which infrastructure alternative is chosen in the first time period and 2) at what point the flexible option is exercised if available. The SDP also requires estimates for the costs associated with a particular action in a certain state of the system. The costs can be broadly defined to include a variety of performance metrics of interest to the water planner. In the applications in this dissertation, we have included the capital and operating costs of the infrastructure itself, as well as damages incurred if water supply reliability targets are not met. Water shortages are estimated using a water resource system model, which combines hydrological models with infrastructure operation models. For example, in the climate change example in Mombasa in Chapter 5, a rainfall-runoff model is used to estimate monthly streamflow time series. Then, a dam operation model with operating rules for releasing flows depending on reservoir storage is used to assess whether supply meets demand in each time period and if not, what volume of shortages are incurred. We do not attempt to optimize the operating rules. This analysis is repeated for each possible climate state. The shortage volumes are multiplied by a shortage penalty value, estimated using economic models of the damages caused to society by water shortages. These costs are summarized in equation 2.3

$$C(s, a, t) = Capex(a) + Opex(a, s) + D * U(a, s, t)$$

$$(2.3)$$

where C is the single-period cost, s is the state of the system including the uncertain hydrological state variables as well as available and planned infrastructure capacity, a is the action which includes building or expanding infrastructure capacity, t is the time period, Capex is the capital costs associated with infrastructure action a, Opex is the operating costs associated with existing and new infrastructure, D is the water shortage penalty representing economic damages for water shortages, and U is unmet demand as a function of the hydrological state and existing and new infrastructure as measured by the water resource system model.

Forward simulation and risk profiles

Finally, after solving the SDP to develop the optimal polices for exercising the flexible infrastructure alternatives, we assess their performance. Monte Carlo simulation is applied to 1) the transition probabilities in the SDP to create time series for the uncertain hydrological state variable(s) and 2) any statistical, low-learning potential uncertainties. The optimal policy is then applied to determine if and when the flexible expansion option is exercised in each simulated time series. The costs and water shortages of the expansion are estimated using the water resource system model. This generates distributions of cost and shortages for each infrastructure alternative that are used to assess the strengths, weaknesses, and trade-offs of different infrastructure planning approaches. For deep uncertainties that rely on scenarios, the simulation analysis is repeated to assess how the performance distributions change under different scenarios. Developing distributions for performance rather than a single estimate enables us to assess upside and downside risk, regret, and other decision criteria. These measures are used to assess the value of flexibility, enabling planners to decide if upfront investments in flexible are worth the cost.

Chapter 3

Desalination planning under multiple, diverse uncertainties during Melbourne's Millenium Drought

This chapter has been adapted from a previously published paper: Sarah M. Fletcher, Marco Miotti, Jaichander Swaminathan, Magdalena M. Klemun, Kenneth Strzepek, and Afreen Siddiqi. "Water Supply Infrastructure Planning: Decision-Making Framework to Classify Multiple Uncertainties and Evaluate Flexible Design." Journal of Water Resources Planning and Management 143 (10), 2017.

Abstract: Urban planners face challenges in water infrastructure development decisions due to short-term variation in water availability and demand; long-term uncertainty in climate and population growth; and differing perspectives on the value of water. We classify these multiple uncertainties and develop a decision framework that combines simulation for probabilistic uncertainty, scenario analysis for deep uncertainty, and multistage decision analysis for uncertainties reduced over time with additional information. We apply this framework to a case from Melbourne, Australia where a drought from 1997 to 2009 prompted investment in a \$5 billion desalination plant completed in 2012 after the drought ended. Our results show opportunities for significant reduction in capital investment using flexible design. Building no infrastructure is best in most simulations. However, in 10% of simulations building no infrastructure leads to regret of greater than \$10 billion compared to a small, flexible desalination plant. Scenario analysis for deep uncertainties underlines the significant impact of assumptions about the future and also on value judgments about the cost of water scarcity in evaluating infrastructure performance.

3.1 Introduction

Urban water planners face the decision of how much water infrastructure to build in order to reliably meet demand for high-quality water while minimizing cost to meet budget constraints. The challenge of balancing the tradeoff between shortage risk, cost, and environmental protection is compounded by several critical uncertainties. Urbanization is driving population growth in many cities at rapid but uncertain rates; natural variation in annual runoff is high and expected to increase due to climate change in many regions [78]; energy price volatility and variable maintenance requirements drive cost risk for desalination and other energy-intense infrastructure options. Additionally, numerous water supply systems in many industrialized countries are reaching the end of their planned lifetimes, prompting further need for infrastructure investment.

Most urban water supply infrastructure (such as distribution pipelines, treatment plants, and reservoirs) in Australia, the US, and other industrialized countries was built between the 1930s and 1980s, before sophisticated risk and decision analysis methods had been developed for practical use [149, 49, 52]. Traditionally, planners developed a long-term demand forecast and long-term supply forecast, and added a safety factor to account for uncertainty [149]. This approach can lead to overdesign when demand ends up being lower than forecasted. If demand exceeds the forecast, society can face economic impacts due to unserved demand, environmental degradation, and expensive measures for building additional infrastructure in short time frames.

Since the 1980s, reliability, or outage frequency, has been emphasized as a risk-based metric for assessing water supply system performance [67]. More recently, researchers and planners have developed strategies to improve resilience and robustness under uncertainty. Early work focused on the use of adaptive management to operate water supply systems more flexibly to ensure resilience under extreme operating conditions [115, 47]. Adaptive management requires planners to change from a predict-then-act approach to a learn-then-adjust approach [121]. A related strategy is flexible infrastructure design, which allows planners

to respond to future uncertainties. The use of engineering options, in which infrastructure is designed so that it can be modified or expanded in the future depending on how supply and demand evolve, provides flexibility that can reduce reliability and cost tradeoffs in water supply systems [168, 33]. The application of an EOA, including staged deployment, has been demonstrated in several recent studies on design of water distribution systems [101], urban water systems [37, 53, 162], water treatment investments [178], and dam investments [80]. Engineering options can be compared and evaluated using multi-objective decision analysis to incorporate multiple planning goals and evaluate flexible design options as uncertainties unfold over multiple planning stages [33].

Recent methods have also been developed for new infrastructure planning that is robust to "deep uncertainties" such as climate change. The likelihoods of deep uncertainties cannot be accurately quantified using historical data, rendering probability-based risk assessment challenging or inappropriate [60, 108]. The goal is to generate and evaluate planning strategies that are robust to a range of future outcomes. For example, RDM uses Latin hypercube sampling to generate many possible futures scenarios across multiple uncertainties, assumes equal likelihood of each scenario, and selects strategies that meet threshold performance criteria across a large percentage of scenarios [93, 92]. Info-Gap theory, in contrast, develops increasingly large multi-dimensional uncertainty sets and identifies the solutions that meet threshold performance criteria for each uncertainty set [14]. Decision scaling links decision analysis with bottom-up climate vulnerability analysis, identifying climate-driven action thresholds without relying on general circulation models to generate climate scenarios [19]. Such approaches have been widely applied to problems in water resources planning in various countries [122, 39, 87, 110]. Climate change and other deep uncertainties should be integrated into a water resources modeling framework that accounts for the full range of uncertainty planners face [135].

While scenario analysis has been demonstrated to be a powerful planning tool for urban water managers facing deep uncertainties and varying stakeholder concerns [94], this does not preclude the use of probabilistic approaches for different, more easily modeled uncertainties. Population growth, which is more easily modeled statistically, is expected to have a larger impact on water resource systems to mid-century than climate change [164, 144, 142, 48]. The stationary statistical variation in rainfall, which can be modeled stochastically, often drives uncertainty in the short- to medium-term more than climate change [95, 42]. In this paper, we build on traditional decision analysis and deep uncertainty methods for water supply planning by developing a framework to classify and model multiple uncertainties of different natures. It enables water supply planners to quantify statistical uncertainties that can be appropriately quantified and to assess the impacts of different, deep uncertainties on a risk profile using scenario analysis. It also identifies which uncertainties enable learning over time and incorporates them into a multistage decision analysis model. This model enables the evaluation of real options such as staged deployment where planners learn as uncertainties unfold and utilize flexible options if and when they are needed.

We apply the method to a case of urban water supply planning in Melbourne, Australia. The Millennium Drought in Southeast Australia from 1997 to 2009 motivated our work. Melbourne's water supply system comprises a network of 10 storage reservoirs with a total of 1,812 million cubic meters (MCM) of storage. Net average annual inflows (evaporation and losses subtracted) were 571 MCM between 1926 and 2014 and demand was 401 MCM in 2014/2015 [104]. This results in a storage-to-annual-runoff ratio of 3.2, which indicates a highly managed system. Catchment-level stream flow management plans require minimum environmental flows to be met [104]; environmental flows ranged from 100 to 410 MCM between 1995 and 2011 [97]. Minimal groundwater is used [97]. A single bulk wholesaler, Melbourne Water, is responsible for harvesting, storage, and treatment.

After more than a decade of below average rainfall, reservoir storage for the city reached a record low of 25.6% of capacity in 2009 [97]. This led to a range of demand management efforts including efficient appliance installations, reduced environmental flows, water restrictions for outdoor uses, a domestic rainwater tank installation program, and treated wastewater recycling. There were also two large infrastructure investments: the 150 MCM per year Wonthaggi RO desalination plant and the 100 MCM/year maximum capacity Sugarloaf pipeline at capital costs of \$5 billion and \$550 million, respectively, able to provide up to about 40% of the city's demand [58]. However, the drought ended before the desalination plant was completed, leaving the plant unused for years. Several studies discuss the political pressure on Melbourne Water and the Victorian government, a detailed timeline of actions taken, and the institutional decision-making process in responding to the drought [49, 58, 126, 97]. Melbourne's reservoir system and Wonthaggi plant are illustrated in Figure 3-1.

This work addresses the question: given similarly uncertain and dire situations like those



Figure 3-1: Melbourne study area map showing locations of reservoirs, streams, Wonthaggi plant, and city center.

Melbourne faced in 2007 and the inherent tradeoff between cost, supply risk, and environmental protection, what approach can water planners use in evaluating infrastructure investments when facing multiple uncertainties of different natures? A decision framework using classification and incorporation of multiple uncertainties is applied in order to 1) evaluate and communicate the cost and water supply risk of proposed infrastructure alternatives, and 2) identify the best infrastructure alternative based on a planner's risk preferences across simulated water supply futures. We do not present a single best solution, as that would require value judgments and preferences from stakeholders and planners. Rather, we present results for a range of sample preferences to demonstrate the usefulness of the framework. The remainder of the paper is organized as follows. Section 3.2 describes the methods used: the uncertainty framework is presented first, followed by a description of its application to Melbourne. Key results and conclusions are presented in Section 3.3 and Section 3.4, respectively. More detailed descriptions of the methods and additional results are shown in Appendix A.

3.2 Modeling approach

Decision Framework for Multiple Uncertainties

Many uncertainties simultaneously impact water supply infrastructure planning including annual reservoir inflows, population growth, energy prices, the cost of water supply shortages, and climate change. We develop a framework for decision making under multiple uncertainties, illustrated in Figure 2-1 in Chapter 2.

The first step in the framework is to classify uncertainties along two dimensions: whether they are deep or statistical, and whether there is high or low potential for learning over time. Deep uncertainties are those for which the likelihood of the outcomes cannot be determined; they are modeled using scenarios analysis in order to avoid estimating likelihood. In this context, scenario analysis refers to varying the value of uncertain parameters in order to understand the impact of their uncertainty on the results. Note that while scenarios are not explicitly probabilistic, the choice of values does assign some implicit probability to those outcomes; this is important to recognize in all deep uncertainty methods. Statistical uncertainties are those that can be estimated using data-driven probability distributions. Statistical uncertainties are modeled using Monte Carlo simulation in order to obtain the most precise risk profile. The high vs. low learning potential indicates whether, as the uncertainty is realized over time, additional observations meaningfully update the planner's expectations of the future. Based on updated expectations, planners decide to exercise or ignore flexible options. These uncertainties are therefore incorporated directly into a multistage decision analysis model [33], in which flexible options are optimally exercised or ignored. This allows the full value of flexible options to be modeled. Uncertainties that do not yield meaningful updates on expected payoffs as observations are made over time can be analyzed more efficiently by varying the inputs across model runs.

This uncertainty classification addresses the level of uncertainty [166] and the potential for learning over time. Note that the level dimension can also incorporate ambiguity, or uncertainties that arise through differences in stakeholder perspectives [20, 90], such as value judgments about the cost of water shortages. Uncertainties arising from ambiguity are similar to deep uncertainties in that they are not appropriate for probabilistic analysis, so scenario analysis is used instead. Different objective functions can be used to represent different preferences or risk profiles. For instance, a risk-averse planner may prefer to invest in water supply infrastructure that often sits idle as an insurance policy against drought that minimizes shortage risk. While there is a large literature on eliciting and incorporating stakeholder values into decision analysis [137, 59, 63], such analysis is not included in this paper. Rather, a sample objective function and a few scenarios comparing alternative objective weights are used to demonstrate the method. Additionally, other dimensions of uncertainty, such as the location of uncertainty within a decision-making process, structural uncertainty, and observational uncertainty, are not addressed here.

In the second step, a model of the water system is combined with the decision model to evaluate and compare infrastructure alternatives. These infrastructure alternatives can be generated through a screening model or planner input. The water system model is simulated to generate the distribution of cost and water supply risk for each infrastructure alternative over all statistical uncertainties. Statistical uncertainties with high learning potential are incorporated into the decision analysis model explicitly so that flexible options can be exercised in response to new information. The decision model ranks the infrastructure alternatives according to an objective function for each simulation run, resulting in a distribution of "best" infrastructure alternatives across simulated variables. Scenario analysis is then used to repeat the simulation process for different sets of input variables representing deep uncertainties to assess the impact of scenarios on the risk profile.

3.3 Application to urban water supply planning in Melbourne, Australia

This analysis framework is applied to an illustrative example from Melbourne, Australia, where planners in 2007 decide what, if any, additional supply investments should be made over a 30-year planning period (the approximate lifetime of an RO plant). Key uncertainties are identified and classified. Infrastructure alternatives are chosen based on those considered by Melbourne's water planners in 2007, with additional flexible alternatives designed for comparison. A model of Melbourne's water system is developed using a simple hydrological model and demand forecasts. A decision analysis model is then used to evaluate the infrastructure alternatives, including some with flexible expansion options, based on an objective function that considers lifetime costs and water shortages.

Table 3.1:	Methods	used for	r key	uncertainties	in	analysis	based	on	uncertainty	classification
framework										

Uncertainty	Classification	Justification			
Short-term variability in streamflow	Statistical; low learning potential	Well-characterized by historical data; stationarity assumption appropriate over short time scales			
Population growth rate	Statistical; high learning potential	Well-characterized by existing demographic forecasts; early changes in growth rate highly predictive of future growth			
Electricity price	Deep uncertainty	Influenced by deeply uncertain factors such as national and international policy and markets.			
Water shortage penalty value	Deep uncertainty	This is an ambiguity, or uncertainty arising from differences in stakeholder perspective			
Demand per capita	Deep uncertainty	Influenced by deeply uncertain factors such as policy, citizen behavior, and technology adoption			

Uncertainty classification

Five uncertainties, listed in Table 3.1, are included in this analysis. They were chosen because of their high degree of uncertainty and potential for impact on Melbourne's planning decisions; however, they are not comprehensive. Certain uncertainties were excluded, such as the potential for desalination technology costs to decline over time, because initial analysis of historical data suggested they were unlikely to impact planning decisions on a 30-year timeline. Other uncertainties, such as climate change, future agricultural production, policy changes, structural and observation uncertainty, could be included in future work. The included uncertainties are classified as indicated in Table 3.1 with brief justifications. Decisions to classify uncertainties using the framework developed in Chapter 2 ultimately rely on analyst judgment but are informed by analyses of available historical data and forecasts. We base the analysis on historical inflow data [104], historical and forecasted population growth [10], and historical and forecasted electricity price [2]. Further details are available in Appendix A.

Table 3.2:	Definitions	for the si	\mathbf{x} infrastr	ucture	alternatives	evaluated	and com	pared in	the
decision-m	odeling fran	nework.	S3 and S	5 were	developed to	o compare	a staged	deploym	nent
approach t	o the full, u	ipfront de	eployment	appro	ach used in	S4 and S6.			

Infrastructure Alternatives	Capital Expenditure (M\$)	Capacity (MCM/year)
S1: No Build	0	0
S2: Pipeline and irrigation upgrade	1,002	Variable: Max 100
S3: Small RO plant with expansion option	$2,045 \ [+1,095]$	Firm: 75 [+ 75]
S4: Large RO plant	2,900	Firm: 150
S5: Small RO plant with expansion option; Pipeline and irrigation upgrade	$3047 \\ [+1,095]$	Firm: $75 [+ 75]$ Variable: Max 80
S6: Large RO plant; Pipeline and irrigation upgrade	3902	Firm: 150 Variable: Max 80

Infrastructure alternatives

We design and evaluate six infrastructure alternatives comprised of combinations of three projects: a 150 MCM/year RO plant based on the Wonthaggi plant, a 100 MCM/year capacity pipeline and accompanying irrigation system upgrade based on the Sugarloaf pipeline and accompanying upgrades, and a 75 MCM/year RO plant designed with a flexible option to expand to 150 MCM/year if desired. The smaller (75 MCM/year) desalination plant was not considered by Melbourne Water in 2007; it was included here as an alternative to assess the value of staged deployment. The flexible design requires the plant site to be sized to fit twice as many membrane modules as needed before the expansion, incurring additional capital costs upfront in exchange for the option to expand cheaply and quickly later. The no-build alternative (S1 in Table 3.2) is included as a baseline option where no new infrastructure is developed.

The desalination plants incur both high capital costs and high operating costs. The operating costs are high due to the energy intensity of seawater desalination and membrane replacement requirements. In addition to substantially lower capital costs, the pipeline and irrigation upgrade also incurs much lower operating costs; pumping is the largest component. The desalination plants, however, provide firm capacity; they can reliably provide the maximum design capacity during a dry year. The pipeline system is market-based: farmers

whose irrigation systems have been upgraded to be more efficient sell excess water to the utility at their discretion. Farmers are less likely to provide water during dry years, so it is unlikely that the full 100 MCM/year is available each year. This dynamic is approximated by assuming a correlation between inflows to Melbourne's reservoir system and water available from the pipeline, with 80 MCM available during average wet years and no water available during dry years. Cost estimates (capital expenditures, fixed operating costs, and variable operating costs) are developed for each infrastructure alternative using cost data from the Sugarloaf pipeline project, cost information on the Wonthaggi plant, cost data from comparable RO plants, and input from desalination experts. Details are available in section B of the Supporting Information.

Water System Model

A water system model is developed using a water balance approach to model reservoir storage and water supplied to end users from both the existing reservoir system and new supply infrastructure. This model estimates the annual cost and water shortages (i.e. unserved demand) over a 30-year period starting in 2007 for each infrastructure alternative. Monte Carlo simulation is used to develop distributions for these estimates based on uncertainty in inflows and population growth.

The main components of the water balance are: net reservoir inflows (inflows minus evaporation), demand from end users, water imported from new infrastructure, and environmental outflows. To model future inflows, 100,000 synthetic annual 30-year inflows are generated using anauto-regressive moving average (ARMA) time-series model fit to historical inflow data dating back to 1926. This approach captures the year-to-year autocorrelation observed in runoff. It assumes inflows to be a stationary stochastic process over the 30-year planning period. Scenario analysis is used to vary the mean and variance of this process based on estimates of climate change impacts on runoff in Australia [150]. This is a simple approach; climate change is not a focus of this paper. Future work evaluating longer planning periods could use more sophisticated statistical approaches such as downscaling from general circulation models. Annual water demand is modeled as the product of population and demand per capita. Population projections from the Australian Bureau of Statistics for the city of Melbourne are used, ranging from 50,000 to 150,000 people per year [10]. The base case demand per capita of 100 kiloliters (kL)/person/year is based on historical demand data and is varied using scenario analysis to assess the impact of demand reduction measures; future work could assess and compare demand-side alternatives directly. Water is imported from any new infrastructure capacity if reservoir storage levels go below one of two thresholds. Pipeline water is imported below an intervention threshold of 980 MCM and desalination water is imported below an emergency threshold of 580 MCM, the maximum allowable drawdown. These thresholds are set by Melbourne Water [106] and used in other studies of Melbourne's water system [161]. In this paper, water shortage is defined as water demand that cannot be met without reducing demand or withdrawing below the emergency threshold. The water system model aggregates the individual reservoirs and uses an annual time step. This approach is appropriate to demonstrate the new method developed in this paper and given the capacity-expansion focus of the decision model [11] and the high intra annual storage in the system. More details on the water system model, including the assumed operational rules, are available in Appendix A.

Decision Analysis Model

The decision model uses the estimates for cost and water shortages from the water system model to evaluate the six infrastructure alternatives (Table 3.2) over a 30-year planning period. For each synthetic 30-year inflow series, the model ranks the six infrastructure alternatives using multi-stage decision analysis, which is frequently used to evaluate real options [33]. The model can be understood as a decision tree, in which the population growth rate has a probability of going up or down in each 10-year planning stage. In the two infrastructure alternatives with flexible desalination design, S3 and S5 (Table 3.2), the planner can react to change in population growth by deciding to exercise the real option of expanding a small desalination plant. The model chooses the infrastructure alternative and desalination plant expansion timing, where applicable, that minimizes the following objective function:

$$E[TotalCost_i] = ENPV[Cost_i + Penalty * WaterShortage_i]$$
(3.1)

where E denotes expected value; i is the choice of infrastructure alternative and expansion strategy; *Cost* is the total of capital costs and operational expenditures incurred from the infrastructure alternative over the planning period; *WaterShortage* is the total volume

of water shortages from years where water demand without conservation or environmental flow reductions exceeds available water supply, also interpreted as water supply vulnerability [67]; ENPV is expected net present value; and Penalty is a cost incurred by the decisionmaker for each MCM of water shortage during the year of the shortage. A monetary penalty for shortages does not exist in reality in Melbourne or most municipal water utilities. Instead, the cost of water shortages is borne by society as economic damages. The penalty value can therefore be interpreted as the planner's degree of risk aversion to water supply vulnerability. A base case value of million US dollars (M\$) 25 /MCM is chosen to be about 50 times higher than the bulk usage price of desalinated water (0.55 /m3) in Melbourne [106] and consistent with previous work [79, 133]. Using scenario analysis, different penalty values up to M\$ 250 /MCM are evaluated, consistent with prices in other previous work [96]. ENPV is calculated using a discount rate of 7%, which is in the middle of the range typically used by government agencies to evaluate projects [66]. Note that this representative objective function is intended to demonstrate the method and does not reflect real stakeholder preferences. Use of this method in future planning could incorporate stakeholder input on the value of the penalty through a collaborative stakeholder engagement process.

Finally, after the decision model ranks the infrastructure alternatives based on the expected value of total cost, Monte Carlo simulation is used to randomly choose a single population growth path for each model run. This allows us to see, across many synthetic inflow and population growth time series, how the infrastructure alternative chosen based on expected value performs against simulated actual conditions. For example, the decision model might choose to build no additional infrastructure because the expected shortages are low; however, there is a small probability that high population growth could lead the actual shortages to be high. Using this approach, results that show the performance of each infrastructure alternative across many realizations of the simulated uncertain variables, reservoir inflows and population growth, are presented. Further details on the decision analysis model are available in section C of the Supporting Information.

3.4 Results

First we present results showing the impact of the statistical uncertainties, reservoir inflows and population growth. This includes results from the water system model showing the cost and water shortage risk of each infrastructure option (Figures 3-2 and 3-4) as well as results of the decision model (Figure 3-3). These results are shown as distributions, since Monte Carlo simulation is used for statistical uncertainties. Then, results showing the impact of deep uncertainties using scenario analysis are presented. Figure 3-5 shows how the distribution-based results obtained from addressing the statistical uncertainties are shifted when scenario-based deep uncertainties are also addressed.

Statistical uncertainties

Figure 3-2 summarizes the model results for cumulative water shortages and infrastructure cost for each of the infrastructure alternatives across 100,000 water supply simulations using bagplots [141]. The bagplot is an analog to the boxplot for bivariate data [141]. For each infrastructure alternative except no-build, the center point marked with an asterisk is the Tukey median [45]; the inner, dark-colored shape is the "bag" which contains 50% of the data, similar to the inner section of a boxplot; and the outer, light-colored shape is the convex hull that encompasses the rest of the data excluding outliers. These plots show the distribution of cost and water supply risk faced as a result of the statistical uncertainties addressed, reservoir inflows and population growth. The no-build alternative costs nothing but incurs the greatest shortage risk, with the median total shortage spanning the 30-year period at 260 MCM. Cumulative shortages greater than 2,000 MCM (over the 30-year period) occur in 20% of simulation runs. The pipeline alternative (S2) consistently imposes a cost of close to \$1 billion, and decreases the shortage risk. The four infrastructure alternatives that include a desalination plant all increase the average cost and cost variability while decreasing the shortage risk further. Interestingly, the median water shortage for all four alternatives with desalination is 0, with 90th percentile shortages all on the order of 1,000 MCM over the 30-year analysis period. Within these four, the alternatives that have more built infrastructure capacity see modest reductions in water shortage risk for more significant cost increases. Most notably, the alternative of the small plant with expansion option (S3) has a very similar water shortage risk profile to the large plant (S4), as shown by the width of the inner bag, with median cost about \$1 billion lower, again demonstrating the value of staged deployment under uncertain demand. The same relationship is true of the infrastructure alternatives with desalination and pipeline. This value is also demonstrated by the variation in utilization of the expansion option: the option is exercised in about



Figure 3-2: 'Bagplots' visualize the distribution of infrastructure cost and total water shortage across a random sample of 1,000 total water supply simulations for each infrastructure alternative. Because the no-build alternative always incurs zero cost and is therefore univariate, the plot shows the median shortage and a line that spans the range of shortage between the 0th and 99th percentiles, which is analogous to the convex hull in the bagplots. Building desalination capacity increases cost while mitigating risk of high magnitude shortages. Flexible design can mitigate shortage risk at lower cost.

45% of simulations in the small plant without pipeline alternative (S3) and about 40% of simulations in the small plant with pipeline alternative (S5).

Across all of the infrastructure alternatives, more than 80% of the years simulated incurred no water shortage at all. Large annual water shortages of >100 MCM were more common than smaller shortages. These large annual shortages are concentrated in the second half of the 30-year period, as the population growth leads to higher demand in later years. This has important implications for the capacity factor of the built infrastructure as well as timing the decision to build new infrastructure. The median number of years that the built infrastructure is used is between 4 years and 6 years across all the infrastructure alternatives. This means that any infrastructure that is built is likely to be used for 20% or less of its lifetime. This is because the total water shortage incurred over the 30-year planning period is typically concentrated in a small number of years that incurred extreme lows in runoff coupled with high demand.

The decision model evaluates the six infrastructure alternatives based on the overall cost, which includes both infrastructure cost and penalties for water shortages over a 30-year period. In addition to choosing a best alternative based on expected values, it also calculates the simulated payoff for each alternative in each model run, and ranks the alternatives according to this simulated payoff. A simple way to communicate and compare the infrastructure alternatives is to compare the rankings, shown in Figure 3-3 for 100,000 simulations of the base case. The no-build alternative (S1) is the best performing infrastructure alternative in more than 50% of cases; however, it is the worst performing alternative in more than 30% of cases. This underlines the risk of planning based on a single forecast. At the other extreme, the large plant plus pipeline alternative (S6) performs the worst in more than 60% of cases and rarely performs within the top half of rankings. Notably, an interesting result is the flexible small desalination plant (S3): despite performing the best in only about 20% of simulations, this alternative performs in the top half of rankings in nearly 90% of simulations and never ends up as worst or second to last. This demonstrates the ability of paying a small premium for flexibility to mitigate downside risk significantly.

The total cost including shortage penalties incurred by each infrastructure alternative across the 100,000 inflow simulations were also analyzed and shown in Figure 3-4. The no-build alternative is the lowest cost in 50% of runs, with more than 40% of runs incurring zero total cost, indicating no water shortages over 30 years. However, the downside risk is significant; in 1% of runs the total cost of the no-build alternative is over \$10 billion greater than the cheapest alternative. Interestingly, the pipeline alternative (S2) performs similarly to the no-build alternative, with about \$1 billion higher cost in the low-cost half of simulations and lower cost in the high-cost half of simulations. This is due to the impact of the market-based system for water sales through the pipeline, demonstrating the limits of water markets to provide firm capacity when there is high correlation between drought in the two interconnected regions. The four alternatives that include desalination have high capital costs, but in more than 30% of simulations they incur cheaper overall costs than the no-build or pipeline alternatives. The small plant (S3) provides a mid-range capital cost



Ranks of Simulated Payoffs by Infrastructure Alternative

Figure 3-3: For each of 100,000 simulations, the simulated total payoff for each of the six infrastructure alternatives is calculated and used to rank the alternatives from best to worst. Doing nothing is best in more than half the simulations but also worst in more than 30% of simulations.



Figure 3-4: Cumulative density function (CDF) of the total cost including water shortage penalties by infrastructure alternative. In the median case, the no-build and irrigation alternatives perform best. However, they pose significant downside risk, with total costs about \$10 billion higher at the 90th percentile. The small plant with option has the lowest cost of the alternatives with desalination in the lower cost half of runs and has a similar downside risk.

alternative that significantly mitigates downside risk through the use of a flexible option.

Deep uncertainties

So far, results have incorporated uncertainties arising from inflow variability and population growth, the two statistical uncertainties addressed. Now, scenario analysis is used to show the impacts of the deep uncertainties addressed: demand per capita, the shortage penalty value, and electricity price growth (see Table 3.1. Each bar in Figure 3-5 shows the frequency that the decision model selected each infrastructure alternative as the best across a set of 100,000 simulations. In the base case, indicated using an asterisk, the model chooses both the no-build alternative and the small plant (S3) about 40% of the time each. The results are highly sensitive to the penalty factor, shown in Figure 3-5 (a). The no-build alternative performs best in more than 95% of simulations when the penalty is reduced to M\$ 5/MCM. Likewise, increasing the penalty makes alternatives with greater capacity more favorable. Increasing the penalty to M\$ 50/MCM makes the no-build alternative best in about 25% of simulations down from 40% and the small plant with option plus pipeline best in 20% of simulations up from about 5%. Increasing the penalty by a full order of magnitude to M\$ 250/ MCM, consistent with the values in [96], increases the magnitude of this shift. The no-build alternative is chosen as best in less than 5% of simulations and the small plant with option plus pipeline (S5) is chosen in about 60%. It is interesting that the small plant with option plus pipeline dominates the distribution of best options over the large plant (S4). Both alternatives have 150 MCM of capacity without the expansion in (S5); however, half that capacity in S5 is variable rather than firm. This suggests that the value of flexibility from the expansion option outweighs the variability in some of the capacity in S5.

The model results are also highly sensitive to the demand per capita, shown in Figure 3-5 (b). Lowering the demand to 80 kL per capita per year substantially reduces the need for additional infrastructure, with the no-build alternative performing best in more than 95% of simulations. The high sensitivity to demand per capita suggests that demand-side conservation measures may be able to significantly reduce the need for supply infrastructure investments, especially if demand reductions are firmly available during dry years. This demonstrates the profound potential of demand management, and points to the need for more rigorous analysis of infrastructure expansion decisions in conjunction with demand management options. Results are relatively insensitive to the electricity price growth rate, shown in Figure 3-5 (c).

3.5 Discussion

The primary contribution of this paper is the development of a method that can be used to inform decision-makers making water supply infrastructure plans under multiple uncertainties. We develop a classification framework for uncertainties that enables the use of probability-based risk quantification and information gathering where appropriate, while leaving deep uncertainties to be addressed using scenario analysis. This approach allows us to evaluate and compare infrastructure alternatives, including those with flexible design, as well as demand-reduction strategies in a way that takes into account the diverse and multifaceted uncertainties water planners face. The use of simple yet information-dense visualizations of risk profiles can provide decision-support for policy makers.



Figure 3-5: Scenario analysis on the distribution of the best infrastructure alternative chosen by the decision model across 100,000 model runs. In each model run, the best alternative is chosen based on the expected payoff. Base cases are indicated with an asterisk. In the base case, no build (S1) and small plant with option (S3) are each chosen in about 40% of runs. Results are highly sensitivity to demand per capita and the penalty value and relatively insensitive to electricity growth rate.

The results demonstrate that moderate investment increases, together with flexible infrastructure design, can mitigate water shortage risk significantly. Using a staged deployment strategy to expand if necessary when more information is available reduces shortage risk at lower cost compared to other, less flexible infrastructure options. The value of this flexibility is dependent on the nature and magnitude of uncertainties in the system. Furthermore, demand reduction strategies may reduce the need for additional capacity. These results also suggest that supply infrastructure can be regarded as an insurance policy against drought. Although the investments may be used infrequently over their lifetime, they mitigate the risk of severe economic consequences due to water shortages. This view can allow for improved risk assessment along with appropriate valuation and expectations of its utility. However, there are tradeoffs between infrastructure cost and water shortage risk.

The Victorian government's decision to build a 150 MCM/year desalination plant, one of the world's largest, has been the subject of heated debate and political backlash given the wealth of reservoir water available for years after plant came online in 2012 [126, 49]. However, as drought impacts the region again and the population continues to growth rapidly, the plant delivered its first water order in March 2017 [57], demonstrating that desalination capacity can play a role in mitigating supply and demand uncertainty even if it used infrequently.

These insights are relevant to other places facing infrastructure decisions driven by multiyear droughts. For example, several municipalities in California are considering desalination investments [54]. Although drought in California is often framed as "the new normal" because of climate change, it is important for both planners and the public to remember that high variability and extended periods of low rainfall are normal even in the absence of climate change. Using a learning-driven staged deployment approach or increasing demand reduction efforts may reduce the need for capital investment. Alternatively, framing any investments as drought insurance that may be used infrequently may increase public acceptance. This strategy was successfully employed in the UK, where Thames Water built a large desalination plant in east London in 2010 citing risks of severe water shortages [82].

Our analysis of the Melbourne case shows that much of the choice of the "best" answer is predicated not only on assumptions about the city's future but also value judgments about the value of water during times of scarcity and society's appetite for risk. The sensitivity to the penalty value suggests that working with stakeholders to choose a value that reflects society's tolerance to risk will be important in applying this method to prospective planning decisions. Future work could also incorporate economic methods to value the impacts of water shortages on society as a way to inform the choice of penalty. Modeling separate end uses — agriculture, municipal, industrial, and environmental — could be used to assess the impact of risk sharing across multiple end-use sectors. Alternatively, the water provider may invest in a water fund (at a much lower cost than committing capital for new infrastructure development) that allows the provider to subsidize costs incurred by users in years of drought. Additionally, the use of financial risk mitigation options, such as the private-public-partnership that Melbourne Water used when contracting the Wonthaggi plant, may further reduce the need for capital-intensive projects.

Future work can extend this study by incorporating additional uncertainties, and comparing the nature of key uncertainties in different regions. For example, while population growth provides significant learning opportunities for planners in Melbourne, there may be different uncertainties that drive staged deployment decisions in other regions. Future work could also explore the impact of different water conservation strategies on the necessity for water supply capacity expansion or sustainability goals and objectives [16]. Some demand reduction strategies such as fines or public education yield uncertain and variable reductions rather than the firm capacity provided by desalination. Future work could also include more detailed models of operational reservoir management to assess the impact of improved operations as an alternative to or in conjunction with infrastructure additions. Additionally, new strategies such as 'fit-for-purpose' water supply, in which water of different quality is used for applications for which the quality level is adequate [113], can be evaluated for impacts on infrastructure scale and design decisions.

Chapter 4

Managing predictive groundwater uncertainty in Riyadh, Saudi Arabia using Bayesian updating and flexible planning

This chapter has been adapted from a working paper: Sarah Fletcher, Kenneth Strzepek, Adnan Alsaati, and Olivier de Weck, "Managing Predictive Groundwater Uncertainty in Riyadh, Saudi Arabia using Bayesian updating and Flexible Design", 2018.

Abstract: Water supply infrastructure planning in groundwater-dependent regions requires predictions of the impact of pumping on the groundwater system in order to ensure that withdrawals can be used to reliably and sustainably meet demand. However, groundwater models often have high predictive uncertainty due to the heterogeneity of groundwater aquifers in combination with typically limited data availability. Previous work has assessed the impact of uncertainty in aquifer hydraulic conductivity (K) and storativity (S) on hydraulic head predictions and the potential for new head observations to reduce uncertainty. Here we extend this work to assess the impact of additional head observations not only on predictive uncertainty but also on infrastructure planning from a systems perspective. Further, we assess the potential for flexible infrastructure planning to manage predictive groundwater uncertainty. To do this, we develop an integrated modeling approach that uses Bayesian inference on a groundwater model to characterize a multi-stage stochastic program. An artificial neural network serves as a statistical surrogate of a numerical groundwater in order to make the problem computationally tractable. This allows us to derive and assess flexible planning strategies using engineering options analysis. We apply this approach to a desalination planning case in Riyadh, Saudi Arabia, where poor characterization of a fossil aquifer creates uncertainty in how long current groundwater resources can reliably supply demand. We find that a flexible planning approach has large value in mitigating the risk of over-building infrastructure with minimal reliability risk.

4.1 Introduction

Water planners face the challenge of developing policies, infrastructure, and management strategies to ensure that high-quality reliable water supply is available to meet societal demand in the future. This requires predictions about future water availability and demand. However, predictions face uncertainty due to limited information to characterize water systems today and how they will evolve in the future. Groundwater resources in particular often face substantial predictive uncertainty. The high degree of spatial heterogeneity in groundwater aquifers combined with data limitations often makes it difficult to accurately estimate key parameters such as hydraulic conductivity. However, filling in knowledge gaps with additional observations can reduce predictive uncertainty [31, 117]. This in turn enables more reliable and efficient planning. Therefore, if poor aquifer characterization makes it difficult to know when current withdrawals will become unsustainable, a flexible infrastructure development plan that can adapt as more information is collected may maintain reliable, sustainable water supply without over investment.

Assessing uncertainty in hydrological modeling, and in groundwater in particular, has a long history [40]. Recently, much attention has been paid to the value of the calibration process in reducing predictive uncertainty in groundwater modeling [111]. Bayesian calibration methods, in which observations are used to update a prior distribution of parameters, are increasingly applied to groundwater inverse problems. Approaches can use either formal likelihood methods [100] or a generalized likelihood uncertainty estimation approach, which takes a broader view of the likelihood function that incorporates analyst judgment [15]. Prior work has also assessed the value of new data in reducing predictive uncertainty. Targeted investment in specific types of data, strategic locations of data, or additional volume of data and incorporating it into the calibration process can reduce predictive uncertainty [31, 50, 117, 173]. Bayesian calibration is typically computationally expensive due to the need to use Monte Carlo approaches to numerically calculate the posterior; however, algorithmic advances and approximate approaches are reducing this barrier [21]. Kalman filters, a form of recursive Bayesian inference, have been applied in groundwater applications such as hydraulic head prediction using multiple conceptual models [172] and joint inference of head and recharge [46].

Recent work has also highlighted the importance of predictive uncertainty in groundwater management decisions [36]. Groundwater management models typically combine a simulation model, such as a numerical groundwater model, with an optimization model; a wide variety of techniques exist [125]. Stochastic and robust optimization methods for incorporating uncertainty into groundwater management models have existed since the 1980s [160]. However, embedding a groundwater simulation model into an optimization model is computationally expensive; the optimization may require many thousands or millions of calls of a simulation model that is already time consuming. Past approaches have used simplified groundwater simulation models, such as a response matrix/unit impulse approach [71, 160], analytical models [120], or statistical surrogate models [9]. Only a limited number of groundwater management models use a multistage stochastic decision approach, in which a decision in an early stage must be made in the face of uncertainty that will be realized in a later stage. Two-stage stochastic linear programming models typically use a simple groundwater planning formulation and/or an inexact solution approach [76, 81].

Multistage stochastic optimization models are powerful, despite their computational expense. These models, such as stochastic dynamic programming SDP, allow the user to identify an optimal action for each possible state of the system, accounting for uncertainty across all possible future states. Non-stationary formulations can be used to represent how uncertainty can change over time. These approaches are therefore widely used in other domains in the evaluation of flexible infrastructure planning, in which infrastructure is designed to be able to change in the future [167]. The performance of flexible planning strategies can be be evaluated using engineering options analysis (EOA), in which simulation models are used to evaluate the tradeoffs between flexible and static approaches [33]. Fletcher et al. (2017) [51] use decision analysis to evaluate flexible desalination plant design for drought resilience in the face of both uncertain runoff and demand growth. EOA has also been applied to water distribution network design [101], and adapting water supply systems to climate change [53]. Multi-stage flexible pumping strategies for groundwater remediation have been developed using dynamic programming and other multistage optimization approaches [124, 139, 29].

This study applies a multistage stochastic optimization model to water supply infrastructure planning under predictive groundwater uncertainty. We combine SDP with Bayesian inference to develop a planning model that assesses the potential for future observations to reduce uncertainty over time. Additional groundwater data can not only reduce uncertainty in the predictions from a groundwater model, but also mitigate the uncertainty impacting groundwater management decisions. We therefore link potential reductions in predictive uncertainty to a planning model that evaluates new water supply infrastructure alternatives. This demonstrates the value of additional information in both predictive power and in key planning metrics like reliability and cost. Finally, we use the planning model to develop flexible infrastructure strategies and use EOA to assess their potential to mitigate risk. As uncertainty is updated over time using the Bayesian inference approach, flexible options can be exercised in response. Our approach addresses computational constraints by training an artificial neural network (ANN) on a numerical groundwater model to serve as a statistical surrogate model that can be efficiently embedded in an SDP.

We apply this approach to a case from Riyadh, Saudi Arabia where non-renewable fossil aquifers comprise half of the city's water supply. Large withdrawals from these aquifers over the past 30 years have led to substantial decline in hydraulic head; eventually, maintaining current withdrawal rates with existing pumping infrastructure will no longer be possible. However, these aquifers are poorly characterized, leading to substantial uncertainty in how quickly hydraulic head will decline and therefore when a transition to alternative supply will be necessary. As Riyadh considers expensive investments in desalination and new groundwater development to replace current groundwater supply, planners can ensure reliability without over investment by monitoring head decline over time, updating predictions, and adapting as needed. We evaluate a flexible planning approach in which, rather than deciding upfront whether or not to develop new infrastructure over a 30-year planning period, the decision to develop infrastructure is deferred. However, a small upfront investment is made to enable new infrastructure to be developed quickly if and when it is needed.

The remainder of this paper is organized as follows. Section 4.2 presents the method in a generalized form. Section 4.3 applies this method to a case from Riyadh, Saudi Arabia, and Section 4.4 presents results. Section 4.5 concludes with summary and discussion.

4.2 Modeling approach

The primary contribution of this paper is to integrate existing tools to develop a method to assess the value of groundwater information and the value of adaptive, flexible planning in mitigating the risk of predictive uncertainty in groundwater management. We use Bayesian inference with a numerical groundwater model to update prior distributions of key aquifer parameters and therefore predictions about head in the future. These predictions are embedded in a SDP where each state of the model in future time periods serves as a potential hypothetical future observation. The SDP model is then used to evaluate flexible planning options, which are evaluated using EOA.

This method is implemented using the following steps, illustrated in Figure 4-1:

- Use a groundwater model to characterize predictive uncertainty in hydraulic head. This generates a prior distribution for hydraulic head over time that reflects the best available information at the beginning of the planning period.
- 2. Apply Bayes' theorem to develop posterior distributions for hydraulic head that update uncertainty to reflect possible future observations. The state space of the SDP comprises possible future values of hydraulic head. Each state value is applied as an observation in Bayes' theorem to assess how our understanding of uncertainty will change if that future state of hydraulic head is reached. A surrogate model, in which an ANN is trained on the groundwater model output, is used to enable efficient numerical integration to calculate the posteriors.
- 3. Develop a multistage SDP in which the transition probabilities (the probability of going from a certain head level in time t to a different head in time t+1) are characterized using the posterior distributions from step 2. Use the SDP results to develop flexible water infrastructure strategies that take into account the potential to learn in the future.
- 4. Evaluate flexible infrastructure strategies and compare to traditional static planning approaches using EOA.



Figure 4-1: Schematic of method used to develop and evaluate flexible infrastructure expansion policies that account for the potential to learn about uncertainty over time. The method integrates a numerical groundwater model and its ANN-based statistical surrogate into a multi-stage stochastic dynamic program. Bayesian inference is used to characterize the transition probabilities; each state of the SDP is treated as a hypothetical future observation and used to update the prior parameter distribution using Bayes' theorem. EOA is used to assess the performance of the flexible alternative and compare to traditional static planning alternatives.

Characterizing predictive uncertainty using a groundwater model

A groundwater simulation model is used to predict hydraulic head h in the future as a function of an uncertain input parameter vector θ . In our application, h is the hydraulic head in a single representative well, and θ comprises the K and S of a confined aquifer. We use the finite difference groundwater model MODFLOW [65]. This model can be interpreted as an operator g mapping parameter values θ to predicted state variable values h.

$$h(t) = g[\theta, t] + \epsilon \tag{4.1}$$

Predictive uncertainty arises from uncertainty in the model parameters θ and the measurement ϵ error. A prior parameter distribution $p(\theta)$ and error distribution $p(\epsilon)$ is developed. In our application where historical head data is limited, we use parameter estimates from a government hydrogeological study to characterize $p(\theta)$ and assume a normal error distribution with variance of 5m; future applications could use Bayesian calibration approaches. We assume that the pumping rate is fixed over time. Because our approach requires numerical calculation of many posterior distributions using the groundwater model, we develop a surrogate model g' that captures the key dynamics of g with greater computational efficiency. Our application uses an ANN, trained on 400 runs of the MODFLOW model with a variety of parameter inputs. This provides an instantaneous mapping from any realization of θ to the corresponding value of h(t). This surrogate approach, rather than an analytical alternative like a response matrix, enables generalization to unconfined nonlinear groundwater models in future applications.

Monte Carlo simulation is used to sample from $p(\theta)$ and $p(\epsilon)$, and run a forward simulation of head over time for each sample. We use this simulations to develop a 99% confidence interval for head over time; this is used as the feasible range in the SDP. This feasible range is then discretized to create $h_{dis}(t)$. The values in the set $h_{dis}(t)$ will serve as potential hypothetical future observations in the Bayesian inference. The discretization granularity should be informed by the range of the set and the SDP formulation; our application uses 1m resolution in hydraulic head.

Bayesian inference to characterize SDP transition probabilities

The objective of a SDP is to minimize the sum of the current costs plus the expected future costs over a set of possible actions a(t). The expected future costs are calculated using the transition probabilities p(h(t + 1) | h(t), a(t)) which describe the probability of being in a certain state in the next time period given the current state and action. In our application, the transition probabilities describe the probability distribution of drawdown in head in the next time period given the head today and pumping rate. By using non-stationary transition probabilities, in which the distribution for head in the next time period can change over time, we take into account the potential to learn about predictive uncertainty in the future.

The transition probabilities are characterized using Bayesian inference. $p(\theta)$ serves as the prior parameter distribution. The observation used to update the prior is taken from $h_{dis}(t)$; all values in $h_{dis}(t)$, which correspond to the state values for head in the SDP, are used in turn to calculate posterior distributions and therefore transition probabilities for each possible state in the SDP. The likelihood function $p(o | \theta)$ is characterized using equation 4.1.

$$p(\theta \mid o) = p(o \mid \theta) * p(\theta)/p(o)$$

$$o \in h_{dis}(t)$$
(4.2)

Because no analytical form of the posterior exists, numerical integration is used to estimate the posterior; Markov chain Monte Carlo (MCMC) methods could also be applied.

Deriving and evaluating flexible groundwater management strategies

With the transition probabilities now characterized, the SDP can be formulated. The objective is given by the Bellman equation:

$$V(s,t) = \underset{a \in A}{\operatorname{argmin}} C(s(t), a(t), t) + \gamma \sum_{s \in S} p(s(t+1) | s(t), a(t)) * V(t+1, s(t+1))$$
(4.3)

where V(s,t) is the optimal action for state s at time t, C is the cost of the current time period, γ is the discount rate, a and is an action taken from the set A of possible actions. The actions here are planning or management decisions undertaken by the water planner. They could include pumping volumes, demand management policies, and building new infrastructure. Costs are a function of the current state, time period, and action; they can include capital and operating costs of infrastructure as well as penalties for failing to meet demand or other performance metrics. The sum in the above formulation calculates the expected future cost; this can be extended to other decision criteria. For example, a risk averse planner may prefer to minimize the 5th percentile of future cost.

Flexibility in groundwater management can take many forms. Adaptive pumping policies can enable the resilience of groundwater resources by limiting withdrawals when aquifers are most vulnerable. Demand reduction policies can be triggered during times of scarcity. Infrastructure can be designed to change operations or expand capacity when needed. In our application, we focus on flexible infrastructure planning, in which the development of new infrastructure is conditional on the future state of the groundwater system. Planners can make design and siting decisions upfront, enabling timely development in the future if needed. In this method, flexibility is evaluated through the use of an action space that varies as a function of the state of the system and time. This enables modeling of the management or planning policies described above that are dependent on, for example, the level of hydraulic head and water demand. The key value of using a multi-stage stochastic programming approach in combination with Bayesian inference, as opposed to deterministic or robust optimization approaches, is that it explicitly allows for assessing how uncertainty can change over time as new information is collected and develop strategies that can anticipate this change in advance.

4.3 Application to Minjur aquifer in Riyadh, Saudi Arabia

The region of Riyadh had a population of 7.7 million in 2014 up from 5.5 million in 2004. [143]. Growing at an average of 3.5 % a year between 2004 and 2014 [143], Riyadh is both the most populous and fastest growing region in Saudi Arabia. Demand per capita including losses is currently about 300 liters per person per day (l/p/d), totaling an average of 1.86 MCM/d supplied in 2016 [6]. A highly arid region, receiving on average only 24 millimeters (mm) of rainfall per year [30], Riyadh's water needs are supplied by seawater desalination piped from the Arabian Gulf and groundwater extraction. In 2016, desalination supplied an average of 1.4 million cubic meters per day (MCM/d) of water to Riyadh and groundwater supplied 1.07 MCM/d, over 90% of which came from deep fossil aquifers [30]. The Minjur aquifer comprises the largest share of these aquifers, supplying about 300,000 cubic meters per day (m³/d).

The Minjur aquifer, which is the focus of our study, has faced substantial drawdown of hydraulic head due to the combination of large withdrawals and minimal recharge. The government has announced plans to promote efficient water use using tariff reform and reduce leakage in the distribution system, aiming to reduce net demand per capita to 170 l/p/d by 2030 [6]. Additional desalination capacity serving Riyadh is planned to be added in the next few years. About 370,000 m³/d of treated wastewater is currently reused in agriculture, landscape irrigation, and industrial applications, with goals to increase wastewater reuse in additional applications. [30]. However, Riyadh's water supply plans assume continued withdrawals from the Minjur at the current rate, or perhaps adding additional groundwater development. As the aquifer continues to drawdown, the current rate of withdrawals will eventually become uneconomical, even with substantial subsidies, or technically infeasible for existing pumping infrastructure.

A key challenge in planning for depletion of the aquifer is the substantial uncertainty in predicting how quickly hydraulic head in the aquifer will drawdown, even if pumping rates remain constant. Estimates for K and S range from 1.39×10^{-5} m/s to 5.31×10^{-4} m/s and 1.00×10^{-4} to 1.12×10^{-3} respectively [41]. Calibrating a numerical groundwater model to more precisely characterize the transmissivity and storativity of the Minjur in the Riyadh region, as well as the spatial heterogeneity of those parameters, requires large volumes of water monitoring data. In Riyadh as in many regions, this type of data is limited.

We apply our planning framework with uncertainty updating to assess 1) how predictive uncertainty in aquifer depletion rates can be updated over time and 2) how flexible, adaptive infrastructure planning can be used to mitigate risk from this uncertainty. In particular, we model the development of new desalination infrastructure to replace withdrawals from the Minjur. We contrast a traditional, static planning approach in which water planners decide at the beginning of a 30-year planning period whether or not to build new infrastructure with a flexible approach. In the flexible approach, planners observe hydraulic head over time and decide if and when to build additional capacity based on updated depletion prediction. To facilitate this approach, planners take advance preparations in the form of design, siting, permitting, initial contracting, etc. so that infrastructure can be developed quickly to avoid reliability outages if and when it becomes clear that it is needed.

Groundwater model

The Minjur is a Triassic aquifer composed primarily of sandstone and shale, extending over 800 km across the central Arabian peninsula. There is an outcrop area approximately 100 km west of Riyadh. The aquifer is estimated to be 315 m thick in the Riyadh area. Recharge in the outcrop area is small, estimated between 3 and 25 mm per year [5, 169].

We model the Minjur aquifer in the Riyadh area using the numerical groundwater modeling software MODFLOW [65]. The model is based on that of [169], which is a study by the United States Geological Survey (USGS) that was used in the Saudi government's 1984 Water Atlas [109]. While numerical groundwater modeling techniques have progressed substantially since the 1980s, the USGS report is the most recent publicly available study of the Minjur, and it provides a simple model appropriate for demonstrating our framework. Following the USGS study, we use a 423 kilometer (km) by 288 km two-dimensional rectangular grid with 315 meter (m) thickness. The lower left corner of the grid is positioned at latitude 22.5° and longitude 45.79°, putting Riyadh and the major pumping well fields in the center of the study area. The left side of the grid is bounded by an irregular no-flow
boundary representing the outcrop of the aquifer. The rest of the grid is also bounded by no-flow boundaries; these boundaries are far enough away that they have negligible impact on the main study area in and surrounding Riyadh. The grid cells are 1 km x 1 km in the main study area and gradually increases to as large as 15 km. While the large grid spacing may underestimate drawdown in individual wells, it allows us to keep the computational cost of the model small and is appropriate for assessing long-term regional impacts on head [8]. The study area, including groundwater pumping and desalination infrastructure, and a sample simulation of the groundwater model are shown in 4-2. A hydrostratigraphic cross section is shown in Figure 4-3.

To characterize the predictive uncertainty in head, we fit prior densities for K and S using the ranges described above; K is assumed lognormal and S uniform. Because of data limitations and for simplicity, we assume these parameters are uncorrelated and homogeneous throughout the aquifer and do not calibrate to monitoring targets. Future work could include MCMC [68, 98] or null-space Monte Carlo [159] methods to generate random fields of parameters consistent with calibration targets.

We model the impact of 120 pumping wells in the Riyadh area. The locations and historical pumping rates of these wells are provided in [169]. We have 2010 data on 60 of these wells which account for over 80% of the withdrawals. The starting head is between 200 and 250 meters below land surface (m.b.l.s.), the range reflects a substantial cone of depression around Riyadh. Recharge is assumed to be 5 mm/y in the outcrop area. We use a transient model with a 30-year simulation horizon and weekly time steps.

In order to develop a fast, statistical surrogate model that can be efficiently embedded in the SDP model, we train an ANN on the output of the MODFLOW model. Latin hypercube sampling is used on the prior parameter distributions to generate 400 K and S samples as input to the MODFLOW model, yielding 400 output time series of hydraulic head. Preand post-processing was completed using the Python library FloPy (Bakker et al., 2016). One representative well, chosen from the largest well field in the center of Riyadh, is used to represent head decline in the planning model. We therefore train the ANN to predict the 30-year time series in that well, using K and S as inputs. MATLAB is used to train a 2-layer feed forward network; the RMSE is about 1 m. Details on the ANN architecture and skill are provided in Appendix B.



Figure 4-2: Top: Schematic of MODFLOW grid with hydraulic head contours from one sample simulation. Pumping wells shown with red dots. Outcrop area colored black on left side of grid. Bottom: Study area with grid location, Riyadh city center, and location of proposed desalination plant in Jubail.



Figure 4-3: Hydrostratigraphic cross section of study area including Minjur aquifer. Reproduced and edited with permission from [41]

Planning Model

We formulate a planning model that represents a simplified version of the planning challenges described above but captures the key dynamics of the uncertainty in head decline. We use the head in a single well as a proxy for hydraulic head decline; we choose a well in the Salbulk well field, which comprises the largest share of withdrawals from the Minjur, close to the center of the cone of depression around Riyadh. Discussions with planners indicate that the most immediate threat to current withdrawals is the technical ability of current pumping infrastructure to continue to operate if hydraulic head goes below approximately 300 m.b.l.s.. Therefore, we impose a drawdown limit of 50 m; beyond this limit pumping is infeasible, and planners must either supply water from new sources or incur penalties for unmet demand. We consider one representative new infrastructure alternative: a desalination plant with 108 MCM/y of capacity, sized to replace all withdrawals from the Minjur aquifer. The Bellman equation in 4.3 is therefore parameterized as follows in Equation 4.4:

$$S = \{h_t, x_t\}$$

$$A = \{p_t, e_t(x_t)\}$$

$$C_t = P(h) * p_t + E * e_t + O * x_t * e_t + S * \max(0, D - x_t)$$
(4.4)

where

- t = time step in the model in years between 1 and 30
- h_t = hydraulic head (meters) in representative well at time t
- x_t = desalination capacity (MCM/y) at time t. x_t starts at 0 and changes to 108 two years after the expansion option is exercised if at all.
- $p_t \in \{0,1\}$ where 0 and 1 indicate respectively that pumping is off or on at time t
- $e_t(x_t) = \in \{0, 1\}$ where 0 and 1 indicate respectively that infrastructure option *i* is not expanded or expanded. If $x_t = 108$, indicating that the desalination plant has already been added, then e_t is constrained to equal 0.
- P(h) = Marginal cost of pumped groundwater at hydraulic head h (USD/MCM)
- E =Infrastructure expansion cost (USD)
- O = Desalination marginal cost (USD/MCM)
- S = Shortage cost (USD/MCM) reflecting damages that are incurred if the depth limit is reached before adding desalination infrastructure, and
- D = Demand (MCM/y), assumed to be 108 equivalent to estimates for current withdrawal rates from the Minjur.

Details of the cost assumptions using in the formulation can be found in Appendix B.

The transition probabilities for the individual state variables are assumed to be independent. The transition probabilities for hydraulic head are determined by the groundwater model and Bayes' theorem as described in equations 4.1 and 4.2. The transition for the capacity state is deterministic, determined by the expansion action e_t . We assume the desalination capacity is available two years after the decision to expand. This reflects the flexible planning process described in the previous section in which upfront planning enables timely capacity additions.

4.4 Results

We now present results from the application of the method to the case in Riyadh. First, we examine the potential for Bayesian learning about predictive groundwater uncertainty by simulating potential future observations. Then, we demonstrate the impact of integrating this learning into the stochastic planning model and derive the optimal capacity expansion policy under uncertainty. Finally, we use the optimal expansion policy to simulate the performance of the flexible approach in comparison to two static alternatives.

Groundwater Bayesian learning

We generate a confidence interval for hydraulic head predictions in our representative pumping well using Monte Carlo simulation: we sample many realizations of K and S from $p(\theta)$ and apply the ANN groundwater model g'. The initial predictive uncertainty is quite large due to the substantial uncertainty in both K and S — more than an order of magnitude for each parameter. A 90% confidence interval shows that hydraulic head in the well if current pumping rates continue could range between 305 meters above sea level (m.a.s.l.) and 245 m.a.s.l., based on a starting head of 337 m.a.s.l. The drawdown limit beyond which the current pumping infrastructure is no longer able to withdraw water is at 287 m.a.s.l. Therefore, at he beginning of the planning period, there is substantial uncertainty about whether this limit will be reached over 30 years.

However, the application of Bayes' theorem to new observations enables substantial reduction of this uncertainty. This is shown in Figure 4-4. On the right side, samples from the $p(\theta)$ are shown; the corresponding confidence intervals are at left. At each time step, a new hypothetical observation o is sampled from within the existing confidence interval for that time period. This hypothetical observation is used to update $p(\theta)$ to get $p(\theta|o)$. In the next period, a new set of samples is drawn from $p(\theta|o)$ and used to simulate a new confidence interval for hydraulic head. The updating process is done on a yearly time step; however, the results are shown bi-yearly in figure 4-4 for visual clarity. Different color shades are used to distinguish time steps; the darkest red samples correspond to the last time step. These dark red samples span a reduced subset of the original parameter space, showing that the model is able to infer which possible combinations of K and S could have led to the simulated observations.



Figure 4-4: A single simulated time series of hydraulic head observations are used to update the parameter distribution (right) and therefore the distribution for projected head (left). The darker red colors highlighting a subset of samples in the prior parameter space indicate the ability of the simulated observations to narrow in on a smaller range of realizations of Kand S that could have led to those observations. These updated parameter samples are used in turn to update the predictive uncertainty in hydraulic head. The large reduction in the confidence interval indicates high value of information; however, the value of information faces diminishing returns with additional observations.

This result shows the high value of information in reducing predictive uncertainty. By year 10, the 90% confidence interval has been reduced by half. These early observations have large value because the uncertainty at the outset was so large. Between years 10 and years 20, the confidence interval continues to narrow but at a much slower rate. The value of these observations faces diminishing marginal returns over time as the uncertainty is reduced. Note that this plot shows one possible simulated set of observations in which the final hydraulic head does not reach the drawdown limit. A different simulation may, for example, show a more rapid depletion rate of the aquifer in which the confidence intervals instead narrow at the bottom half of the original confidence interval. Across many tested time series of simulated observations, we observe a similar pattern of narrowing confidence intervals with large but diminishing value of information in reducing predictive uncertainty.



Figure 4-5: The SDP results provide a hydraulic head threshold beyond which expanding desalination capacity is optimal. The threshold is above the depth limit so that desalination capacity can come online after two years without incurring shortage damages in the interim. This threshold approaches the depth limit as time increases, reflecting that a particular head observation in later time periods reflects a slower drawdown rate than the same observation made in an earlier time period would have.

SDP

As described in the previous section, a single simulated set of hypothetical future observations reduces the predictive uncertainty in the groundwater model over time. The SDP models all the possible hypothetical future observations probabilistically through the use of the transition probabilities. The hypothetical observations correspond to possible hydraulic head states in the SDP state space. The optimal expansion policy in time period t, therefore, describes the best decision given the remaining uncertainty that exists after the information based on the current hydraulic head state is used to update the future uncertainty. It is a function of both the hydraulic head and time.

Figure 4-5 shows the drawdown threshold for the decision to add desalination capacity. Beyond this threshold, the optimal policy is to add desalination capacity if none has already been added. If the threshold has not been reached in a particular time step, the optimal policy is to not add new capacity. As seen in the figure, the drawdown threshold is monotonically decreasing in time; it approaches the depth limit at the end of the planning horizon. This pattern arises from the learning process described in the previous section. If an early observation in year three, for example, shows a hydraulic head of 300 m, we infer that we must have relatively low values for S and/or K and will likely continue on rapid drawdown path. Therefore, while the head is still 13 m from the depth limit, there is a high probability that the hydraulic head will reach the limit in the next two years. This necessitates a decision today to add capacity that will become available in two years in order to prevent shortages. Observing a hydraulic head of 300 m in year 15, however, indicates a much slower drawdown rate. Therefore, the optimal expansion threshold is closer to the drawdown limit because the risk of rapid head decline in the two years it would take the desalination plant to come online is much lower.

Engineering options analysis

Next, we use the transition probabilities and the optimal policy results from the SDP to simulate 1000 time series for hydraulic head and the capacity expansion decisions, respectively. These simulated time series are used to assess the cost and reliability performance of the flexible desalination strategy and compare it to static planning alternatives.

Figure 4-6 shows the distribution of capacity expansion decisions across the 1000 simulations. As indicated by the black bar at the right of the plot, desalination capacity is never added in just under half of the simulations. In these simulations, the drawdown of hydraulic head is slow, never crossing the drawdown expansion threshold. However, in over half of simulations, desalination capacity is added, and the time at which it is added varies considerably. Sometimes, as early year 8, a decision is made to bring new capacity online, indicating a very rapid drawdown rate. There is a gap in years 15-17 in which few expansion decisions are made. This corresponds to years where the amount of uncertainty reduction in hydraulic head prediction is limited because the range of annual drawdown across different groundwater models is more similar; if we have not learned enough to make the expansion decision before year 15, we are unlikely to learn enough in years 15-17 to make that decision either. Finally, the frequency of expansion tapers off around year 21. This is because of the planning formulation; when the desalination plant is brought online late in the planning period, it only provides value for a small number of years. Therefore, it is often optimal to incur shortage damages for a short time instead. This is a known limitation of finite-horizon



Figure 4-6: Histogram showing whether and when desalination capacity is added across 1000 forward simulations. The desalination plant is added in a little under half of simulations. When desalination capacity is added, the timing of that decision variables considerably. This variation is driven primarily by differences in the parameter realizations of K and S across simulations. However, different rates of learning and the finite-horizon formulation of the SDP also contribute to the pattern of time variability.

SDP models; they discourage investment late in the planning period when in reality those investments would bring value in the years after the planning horizon ends. The late-period results should therefore not guide planning decisions.

Finally, the performance of this flexible strategy across 1000 simulations is compared to the two static alternatives: one in which the desalination plant is built at the start of the planning period, and one in which the desalination plant is never built. Figure 4-7 presents cumulative distribution functions (CDF) of the total cost C over the 30-year planning period for each of the three planning alternatives. As described in the previous section, this cost includes capital and operating costs for the desalination plant, groundwater pumping costs, and damages for any water shortages incurred. The static no-build alternative, shown in orange, incurs only pumping costs in about 50% of simulations; this is shown by the vertical section of the CDF in the lower half of the plot. In the upper half of the plot, however, we see that the no-build alternative incurs substantial shortages in more than half of simulations. The static build alternative is a little over half a billion dollars more expensive than the no-build alternative in the 50% of simulations when shortages are not incurred; this is seen in the bottom half of the plot. This difference reflects the capital cost of the desalination plant and demonstrates that the static build alternative faces substantial over-build risk. In the upper half of the plot, however, we see that the presence of the desalination plant eliminates the risk of water shortages; the modest cost increases in the blue line here reflect increasing pumping costs and the presence of desalination operating costs.

The flexible alternative mitigates both the over-build risk faced by the static build alternative and the reliability risk faced by the static no-build alternative. In the top half of the plot, the yellow line representing the flexible alternative nearly overlaps the blue line. This shows that it eliminates nearly all the shortage damages. Small deviations to the right of the blue line show small shortages that are incurred either by bringing the desalination plant online slightly too late or by not building at the end of the planning period. In the bottom half of the plot, we see that the flexible alternative overlaps the static no-build option in about 40% of simulations, indicating its ability to prevent over-building. In the 10% of simulations between the 40th and 50th percentiles approximately, the flexible alternative does over build; it is aligned with the more expensive build option rather than the less expensive no-build option. Because of this and the small deviations at the top of the plot, the flexible alternative does not completely stochastically dominate the other two alternatives.



Figure 4-7: CDF of total cost by infrastructure alternative. Costs include capital and operating cost of desalination infrastructure, groundwater pumping costs, and damages for water shortages. The flexible planning alternative mitigates almost all the reliability risk in comparison to static no-build alternative. It also prevents overbuilding in comparison to the static build alternative in about 80% of simulations. The flexible alternative saves an expected \$200 million and mitigates both upside and downside risk substantially, demonstrating high value of flexibility. Horizontal axis truncated for visual clarity; no-build costs increase to a maximum value of \$45 billion

However, it clearly eliminates most of both the upside and downside risk. The advantages of the static alternatives are limited. Further, its expected cost of \$2.28 billion is modestly lower than the \$2.47 billion of the build alternative and substantially lower than the \$13.23 billion of the no build static alternatives respectively. The value of flexibility, therefore, is high.

Sensitivity Analysis

These results rest on a number of assumptions made in the groundwater model and planning formulation. Two assumptions that often strongly influence the results in infrastructure planning problems are the discount rate and the value of reliability, which is formulated here as a cost imposed on any water shortages that are incurred. In the results presented previously we assume no discounting. While investment decisions are typically made under discounted cash flow assumptions, this choice enables us to highlight the value of flexibility even in the absence of discounting, which incentivizes the delay of capital investments. Here we assess the impact of adding a 5% discount rate. The previous results also assumed a shortage cost of $20/m^3$, based on World Bank estimates of water productivity in Saudi Arabia. [158]. We now present results varying this shortage cost to be $5/m^3$ and $50/m^3$.

Figure 4-8 again presents CDFs of the total cost of the three planning alternatives across 1000 simulations as Figure 4-7 did; now, scenarios for a 5% discount rate, $5/m^3$ shortage cost and $50/m^3$ shortage cost are shown. These plots show that while the precise performance of each of the alternatives varies across scenarios, the high value of flexibility does not. The flexible alternative outperforms the other two on expected value, 90th percentile costs, and 10th percentile costs, highlighting again that there is little advantage to either of the static alternatives. When a 5% annual discount rate is added, the flexible alternative performs even better than before. It almost completely stochastically dominates the other two, with the exception of about 5% of simulations around the 50th percentile where it overbuilds. The only significant impact of the high shortage cost is to make the no-build alternative perform substantially worse. The horizontal axis is truncated for visual clarity; the 90th percentile costs reach 82 billion. The lower shortage cost improves the performance of the flexible alternative by decreasing the frequency of overbuilding. Thus we conclude that high value of flexibility with few trade-offs in comparison to the static alternatives is robust to changes in discount rate and shortage cost, both of which often impact the value

of flexibility substantially.

4.5 Discussion

Our results highlight both the value of information and the value of flexibility in reducing and managing predictive groundwater uncertainty. The application in Riyadh, Saudi Arabia assesses a planning problem in which alternative supply sources, in this case desalination, must be developed in response to aquifer depletion. A traditional planning approach would use the best hydrogeological studies available — or commission an expensive new study to make predictions about future water availability, and make plans to develop alternative sources in the future based on the best prediction today. Even our best predictions, however, are subject to uncertainty. Fully characterizing the transmissivity and storativity of an aquifer requires a large amount of data that is often not available, especially in the developing world. Characterizing this uncertainty and making plans that explicitly acknowledge the possibility of unexpected outcomes can enable plans that are robust to a wider range of futures.

However, while it is important to account for uncertainty in planning, it is also important to asses the extent to which that uncertainty can be reduced in the future with new information. Bayesian calibration methods have enabled probabilistic characterization of predictive groundwater uncertainty and demonstrated the value of additional data in reducing that uncertainty. In this study, we build on these approaches by connecting predictive uncertainty to planning decisions. We assess the value of information not only in reducing predictive uncertainty, but also in enabling effective planning under uncertainty. Further, we show that flexibility, when combined with uncertainty reductions over time, can enable reliable planning decisions at reduced cost compared to static, robust approaches. In this case, we found the high value of flexibility to be robust to variations in the cost of water shortages, or how much value society places on water reliability.

We achieve this by developing a new planning framework that combines Bayesian inference on a high-fidelity groundwater model with multi-stage stochastic programming; this is made computationally tractable through the use of a ANN-based statistical surrogate model. This optimization approach enables planners to develop strategies that explicitly account for the range of uncertainty in the future and to choose different, adaptive actions



CDF of Total Cost (Including Shortage Damages)

Figure 4-8: Sensitivity analysis

as the state of the system evolves over time. Uncertainty is characterized using transition probabilities, which describe the probability distribution for the state of the system in the next time period given the state today. Our approach characterizes these transition probabilities using Bayesian inference on a groundwater model, allowing us to model the potential for a certain hypothetical future observation to update the uncertainty. We can infer, for example, that observing only 10 m of drawdown in hydraulic head after 20 years means that we must have relatively high K and storativity and therefore that low drawdown will likely continue in the future. Including this potential future information — across all the possible observations we might encounter — enables adaptive strategies that account for the fact that we can learn in the future.

Each of the individual components in this analysis – MODFLOW, ANNS, SDP, Bayesian inference — is a well established existing method. The methodological contribution here lies in a novel framework that integrates them in a novel way to support water supply planning that accounts for uncertainty, learning, and flexibility using our best physical understanding of the system. This integration is enabled by the use of a statistical surrogate model, in this case an ANN trained on a MODFLOW model, which captures the dynamics of a non-linear physical model in a way that is computationally tractable. It allows us to use numerical methods for Bayesian uncertainty analysis for each possible head observation in each time period, corresponding to the full state space of the SDP.

This new method is applied using a simple groundwater model and planning formulation that allow us to demonstrate the approach with clarity. This formulation has limitations we must keep in mind in interpreting the results. The groundwater model is a lumped parameter, uncalibrated 2D model; the initial uncertainty in K and S is estimated using a recent hydrogeological study of the region. This approach does not capture the substantial spatial heterogeneity known to exist or detailed geologic information, and it should be interpreted as an approximation appropriate for assessing the first-order dynamic of long-term head decline. However, lumped parameter or zonal models can often have greater predictive power than more detailed alternatives which require more assumptions than can be justified with the available information [146, 44]. In our application, K and S can be interpreted as capturing information uncertainty in a parsimonious model rather than characterizing spatial heterogeneity. We use a single grid cell from the Buwayb well field at the center of Riyadh as a proxy for hydraulic head. This approach, while again an appropriate approximation for long-term head decline, does not capture the shape of the cone of depression in hydraulic head around Riyadh. Both the simple groundwater model and using a single proxy well impact the learning process shown in our results. A calibrated model may have had smaller initial uncertainty and therefore less learning at the outset. The presence of spatial heterogeneity may make learning more difficult. However, the presence of multiple wells may increase opportunities for learning.

The planning formulation assumes that groundwater pumping is either on or off and that the traditional, static planning alternatives are restricted to either building a desalination plant upfront or not at all over 30 years. In reality, restrictions on pumping could reduce rather than cutoff pumping. Similarly, if head decline made pumping infeasible in year 10, for example, planners would no doubt react before the end of the planning period in year 30. The specific quantitative comparison of expected value of the flexible approach to that of the static approaches should therefore be interpreted with caution. The goal is not to attempt to quantify the specific performance of the different approaches, but more broadly to assess whether there is value to invest in a flexible approach. The essential element of the flexible approach compared to a traditional approach is that a desalination plant can come online quickly in only two years, compared to a more typical 5+ year horizon. Therefore, while it is unlikely that many years of shortages would be tolerated, several years of shortages are still possible. The low sensitivity of the results to the shortage cost suggests that the value of flexibility would be similarly robust to changes in the duration of shortage.

Future work could extend this approach to more complex groundwater models and planning formulations and address a wide variety of groundwater systems and planning questions. Spatial heterogeneity in aquifer parameters could be addressed through a zonal approach to parameter estimation and the use of multiple monitoring wells. The ANN framework could accommodate this through a greater number of input and output parameters corresponding to different zones and multiple wells respectively; estimating the posterior over a greater number of dimensions could take advantage of MCMC methods if necessary. Initial analysis suggests that 5 wells, for example, could be accommodated into the SDP state space while still maintaining computational tractability; approximate SDP methods could be used for larger scale problems [127]. More active approaches to data collection could be tested; for example, as an alternative to passively collecting new head observations over time, the SDP formulation could include a decision variable to install new monitoring wells to collect information immediately.

In the planning formulation, a wider range of infrastructure or management alternatives such as adaptive pumping rates or demand reduction policies could be explored. Demand reduction measures have the potential to further reduce the need for expensive infrastructure investments; however, they have the potential to introduce further uncertainty given the difficulty in predicting human responses to incentives and restrictions. Wastewater reuse could also be explored as an infrastructure alternative, which would require extension of the framework to include constraints on water quality for different end uses. Evaluating these more varied alternatives would benefit from consideration of additional planning objectives. For example, desalination is a highly energy-intensive supply source; considering the energy and greenhouse gas intensity of different alternatives may further disincentive large desalination projects. Integrating the approach developed here for learning and flexible planning with the multi-objective simulation-based optimization approaches common in water resource systems analysis could address these limitations by enabling multi-objective evaluation of many alternatives in large-scale systems.

The value of information and flexibility are likely to vary substantially depending on the degree of uncertainty, planning decision, and water system. The high value of flexibility robust to deviations in the value of reliability may not hold in other applications with greater interactions with surface water and therefore stochastic variability or if infrastructure costs were less substantial or could achieve higher economies of scale. Exploring these elements in varied future applications can build theory around the drivers of and limits to flexible design and adaptive planning as an efficient and reliable approach for water supply planning under uncertainty. Our initial application of this method highlights the potential for Bayesian learning in combination with flexible planning to mitigate both cost and reliability risk in a poorly characterized groundwater system with unsustainable withdrawals. The prevalence of uncertainty and over-pumping in many groundwater supply systems highlights the potential for this approach to have widespread applicability.

Chapter 5

Bayesian updating of climate change uncertainty to enable flexible water infrastructure planning

This chapter has been adapted from a working paper: Sarah Fletcher, Megan Lickley, and Kenneth Strzepek, "Bayesian updating of climate change uncertainty to enable flexible water infrastructure planning", 2018.

Abstract: The design and planning of new water supply infrastructure must account for climate change in order to ensure reliability targets can be met over the infrastructure's multi-decade lifetime. However, climate change projections face substantial uncertainty in many regions. Recent work has focused on scenario-based approaches to develop robust planning strategies that perform well across many possible future climates. Robust approaches, however, face high risk of overbuilding expensive infrastructure if the worst outcomes are not realized. In this paper, we assess the potential for potential future climate observations to reduce uncertainty and enable flexible infrastructure approaches to mitigate overbuild risk without impacting reliability. To do this, we develop an integrated modeling approach that uses Bayesian model averaging to estimate and update dynamic model uncertainty estimates in climate change projections. These uncertainty estimates are used to characterize non-stationary climate change uncertainty in a stochastic dynamic program (SDP) to assess the potential to learn about climate change over time. We use engineering options analysis (EOA) to evaluate the value of flexible infrastructure design. We apply this framework to a dam design problem in Mombasa, Kenya. We find high potential to reduce uncertainty over time and high value in a flexible dam design in which the dam can be raised in the future to increase storage capacity. The value of flexibility, however, depends on discounting and technology choice.

5.1 Introduction

Long-term planning of water supply infrastructure faces uncertainty in how the climate system will change and how those changes will impact local water resource systems. Climate change uncertainty can increase the vulnerability of a water supply system if its sensitivity to climate is strong and its adaptive capacity to respond is limited [3]. Recent work has addressed both the limitations in projections of climate change by GCMs [78] and developed approaches to planning under climate change that account for those limitations by developing planning strategies that render the water system relatively insensitive to climate change projections [149]. Identified sources of uncertainty in climate projections include: emissions scenario uncertainty; internal variability in the climate system; and structural differences across models due to differences in representations of physical processes, initial conditions, and boundary conditions [69]. The magnitude of these uncertainties — in particular, the influence of global geopolitics on greenhouse gas emissions — has led many researchers in the planning community to address climate change as a "deep uncertainty" which is so severe that probabilities cannot or should not be placed on the possible outcomes [165]. Much of this work uses non-probabilistic, scenario-based approaches to reduce the vulnerability of water infrastructure investments to climate change uncertainty by developing strategies that meet reliability goals and other performance targets in many possible future climates [92, 19, 87, 61, 89]. Early approaches such as Robust Decision Making (RDM) [93, 92] focus on static robustness, or preventing system failure under the largest number of possible future scenarios [165, 73]. These approaches are effective in developing strategies that meet performance goals in a wide range of future climates over several decades but can lead to expensive overbuild of capacity if the worst outcomes are not realized. The impacts of overbuilding are especially severe in resource-scarce regions of the developing world.

Many studies have highlighted the importance of adaptive management approaches to climate change in which managers adapt and react as uncertainties unfold over time [108, 121]. Adaptive management can enable reliability at reduced cost by developing plans to ensure short-term water needs are met without investing in permanent solutions before the long-term future is well understood. Recent methods such as adaptation tipping points [62] and dynamic adaptive policy pathways (DAPP) [61, 89] enable policymakers to identify tipping points or thresholds beyond which new polices will be needed. Identifying these thresholds — and the policies to be implemented if they are reached — in advance enables a planned adaptation approach in which adaptive policies are fully developed and can be nimbly executed when needed, reducing short-term transition risks [102]. Defining "dynamic robustness" as flexibility enabling a plan to change in response to changing conditions over time, Walker et al. (2013) raises the limitations of RDM to develop dynamic robust solutions and highlights the value of DAPP in achieving dynamic robustness [165].

However, adaptive approaches are more difficult in infrastructure planning, as infrastructure requires large capital investments (often on the order of hundreds of millions or billions of dollars) that can last for a century and are difficult to change [35]. In infrastructure planning, therefore, adaptive approaches can pose trade offs or risks to static robustness if short-term supply gaps due to droughts, for example, cannot be addressed quickly enough with adaptive infrastructure planning [12]. Some recent studies have started to integrate adaptive management into infrastructure planning approaches, explicitly accounting for the potential for short-term adaptive strategies such as water transfers and demand reductions to reduce or eliminate the need to build infrastructure [177]. Similarly, Beh et al. (2015) [12] built on prior approaches in sequencing water infrastructure investments [134, 84, 75] and static robustness approaches by including both flexibility and robustness as objectives in an iterative planning approach.

These approaches, however, have some important limitations in how they address flexibility in infrastructure. First, the types of flexibility addressed in these approaches is limited, focusing primarily on flexibility in management or operations of water systems rather than on flexibility in infrastructure itself. Flexibility can be incorporated into infrastructure in a variety of ways. Flexible planning processes can enable flexible timing, location, or type of infrastructure. For example, a staged development approach to infrastructure development in which capacity is developed modularly in smaller steps can reduce the risk of overbuilding capacity. Infrastructure can also be physically designed to be more easily retrofitted in the future [168]. In water infrastructure, this has been assessed in hydropower projects [153] and desalination plant design [51]. Second, flexibility is a strategy or life-cycle system property for achieving other performance targets rather than a performance goal unto itself. Including flexibility as an optimization objective without assessing the value of flexibility in achieving performance goals fails to address this distinction. Proactive approaches to flexible infrastructure design and planning, in which preparations or investments are made upfront in order to enable future flexibility, have a cost. The benefit of the flexibility in achieving performance targets must be evaluated in order to asses whether this cost is justified.

An additional limitation of these approaches to flexibility is that while they account for changes in the *state of the system* to drive adaptive planning or operations, they do not account for changes in *understanding of uncertainty* to drive adaptation. The ROF metrics developed by [123] update the probability that a performance target will be violated based on the state of reservoir storage; historical data is used to assess how often a certain reservoir storage level has previously led to failures. This does not account for the ROF associated with a certain storage level to change over time as more information about the system is collected or the system evolves. This limitation is especially important in addressing climate change, where non-stationary runoff can substantially change the ROF associated with a certain system state over time.

We address these limitations through the development of a planning framework that integrates several existing analysis tools. Engineering options analysis (EOA), related to real options analysis in finance, identifies flexible infrastructure options that can be exercised in the future; Monte Carlo simulation methods are used to assess the performance of flexible approaches with options and compare them to static approaches without options [33]. Engineering options analysis can address two of the above limitations by 1) including "real-in-options" alternatives that include physical infrastructure design in addition to alternatives using levers in policy, planning, or management and 2) providing a framework to estimate the value of flexibility and its tradeoffs relative to traditional robust or static approaches. Additionally, the use of multi-stage stochastic dynamic programming (SDP) with non-stationary transition probabilities can enable the development of policies for exercising options that change over time as both the state of the system and the characterization of uncertainty are updated. While the computational expense of SDP has often limited its use with realistic simulation models, recent methods in water and other domains have enabled this integration through approximate methods [75, 127] and surrogate statistical models (e.g. Chapter 4 of this dissertation).

SDP requires the use of probabilistic approaches to characterizing uncertainty. This stands in contrast to the many approaches above which use non-probabilistic, scenario methods for addressing climate change uncertainty. However, climate change uncertainty has multiple sources. While emissions uncertainty is highly influenced by geopolitics and institutional decision-making, model uncertainty and internal variability are not subject to these forces. Different uncertainties of different types can be addressed uniquely in the same planning problem [51]. Previous work has shown that model uncertainty dominates long-term uncertainty in global mean precipitation change [70], and recent work has enabled probabilistic estimates of model uncertainty [156].

Probabilistic estimates for model uncertainty in climate change are typically developed by comparing and weighting individual model projections from ensembles of GCMs from the coupled model intercomparison project, phase 5 (CMIP5). CMIP5 has collected and validated a large number of GCM simulations from various modeling groups, where simulations are all forced by the same emissions scenarios, or representative concentration pathway (RCP), to allow for a consistent comparison across models. Several methods have been put forward to assess the uncertainty across models using ensembles of projections from CMIP [156]. An early approach by Räisänen and Palmer (2001) [130] derives probability distributions by assuming a "democratic" weighting in which each model projection is assumed equally likely. This democratic weighting of models is also adopted by the Intergovernmental Panel on Climate Change (IPCC), where the multi-model mean and standard deviation are used to characterize the trend and uncertainty in climate projections [78]. More recent research has used BMA, a statistical approach which weights models based on their ability to reproduce historical observations. For example, Giorgi and Mearns (2002) [55] present the "reliability ensemble averaging" methodology, which uses BMA-derived weights to estimate the mean and standard deviation of future change, assuming a normal distribution. Tebaldi et al. (2005)[157] and Smith et al. (2009) [148], extend this to develop a fully Bayesian approach that uses BMA-derived weights and MCMC methods to estimate a posterior without assuming normality.

This study develops an integrated modeling framework that addresses the impacts of model uncertainty in climate change projections on water supply planning. It uses and extends the Bayesian statistical model developed by Smith et al. (2009) [148] to develop probabilistic projections of change in 20-year mean temperature (T) and precipitation (P). These probabilistic estimates characterize uncertainty dynamically: they are updated in a Bayesian manner to account for potential future climate observations. The probabilistic projections are used to characterize the non-stationary transition probabilities in an SDP in which mean T and P comprise the states of the system. The change in T and P over a 20year time period (ΔT and $\Delta \% P$ respectively) are treated as hypothetical future observations used in the Bayesian updating: this allows us to assess how uncertainty in change in ΔT and $\Delta \% P$ would shift as more information about climate sensitivity is collected over time. For example, high rates of temperature change in the first 20 years are more likely to beget rapid temperature change in the next 20 years. By including all the possible future climate observations and their impact on uncertainty in the transition probabilities of a stochastic dynamic program, we explicitly account for dynamic, non-stationary climate change in our planning. Finally, we use the SDP to develop policies for exercising flexible options for infrastructure design and planning, and then assess these policies using engineering options analysis. The overall approach for assessing flexible planning by taking a dynamic approach to uncertainty that accounts learning in response to future observations is described in more detail in Chapter 2 and illustrated in Figure 2-3. We demonstrate this integrated modeling framework on a dam design application in Mombasa, Kenya.

The remainder of this chapter is organized as follows. In section 5.2, we present the generalized framework for integrating dynamic, probabilistic assessments of model uncertainty in climate projections into a SDP to evaluate flexible infrastructure. Then we describe the application to a dam design problem in Mombasa, Kenya in section 5.3. Section 5.4 present results, and we close with conclusion and discussion in section 5.5.

5.2 Modeling approach

Assessing the impact of climate change uncertainty on water supply infrastructure planning requires integration of several components. Long-term estimates of climate change must be translated into the monthly runoff time series needed for dam sizing. In order to quantify model uncertainty in climate change and propagate its impacts through an infrastructure planning model, we develop an integrated modeling approach that combines Bayesian modeling of uncertainty, stochastic weather generation, a rainfall-runoff model, and a reservoir



Figure 5-1: Schematic of integrating modeling framework to assess model uncertainty in climate change projections on infrastructure planning and assess the potential for Bayesian updating of uncertainty and flexibility to mitigate risk. Climate and hydrological modeling components are at left in blue; infrastructure planning and operations shown in center in yellow; integrated results at right in green.

operation model. These components are embedded in a SDP to develop and evaluate flexible infrastructure planning strategies against key performance metrics. A schematic of this process is shown in Figure 5-1, and each of the components are described below.

Stochastic dynamic programming (SDP)

SDP is an optimization approach that models decision-making under uncertainty over multiple stages or time periods. An SDP model is formulated by defining: a set of actions the decision maker can take, a set of states of the system, the cost of the actions as a function of the state of the system and the time period, and the transition probabilities. The transition probabilities characterize the uncertainty in the system by defining a probability distribution for state of the system in the next time period given the state and action in the current time period. These components are integrated into the Bellman equation, shown in equation 5.1, which is solved to develop an optimal policy, or the optimal action for each state in each time period.

$$V(s,t) = \underset{a \in A}{\operatorname{argmin}} C(s(t), a(t), t) + \gamma \sum_{s \in S} p(s(t+1) | s(t), a(t)) * V(t+1, s(t+1))$$
(5.1)

where V is the optimal policy, t is the stage or time period, a is an action from a set of possible actions A, s is a state from the state space S, γ is the discount rate, and p(s(t+1)|s(t), a(t)) are the transition probabilities. In our application, the state space S comprises two climate variables important for runoff modeling, mean T and mean P over a 20-year period, as well as the type and amount of infrastructure capacity added. The action space A describes the amount of infrastructure capacity for a dam or desalination plant at the outset of the planning period, and then whether infrastructure capacity is expanded in later time periods. Future applications could also include other water planning and management alternatives such as other infrastructure alternatives, reservoir operations, and demand-side policies. The costs C include the capital and operating costs of infrastructure capacity additions as well as damages if the infrastructure fails to meet reliability targets. Bayesian modeling of climate change uncertainty is integrated into the optimization framework through the transition probabilities. Transition probabilities $p(T_{t+1}|T_t)$ and $p(P_{t+1}|P_t)$ are developed using the Bayesian modeling approach described in the next section.

We develop a novel approach in which each of the T and P states in the SDP state space are treated as hypothetical, future observations and used to develop updated posterior distributions that account for that observation. For example, let us assume that 40 years from now, we observe a mean T of 28 degrees Celsius (°C) and mean P of 68 mm/month. This suggests a rapid warming and drying trend. We use those values as observations in our Bayesian analysis to develop an updated posterior for change in T and P over the next 20 years, and that posterior is used to characterize $p(T_{t=3}|T_{t=2} = 28)$ and $p(P_{t=3}|P_{t=2} = 68)$ where t is measured in time steps of 20 years. These particular posteriors would have higher probability on warm and dry states in the next time period than the prior. This analysis is repeated for every possible climate observation in each time period to develop transition probabilities that reflect Bayesian learning based on each potential future climate. This approach enables us to address two limitations in many current approaches to water supply planning under climate uncertainty: 1) the ability to characterize uncertainty dynamically so that it can be updated over time, and 2) the development of flexible strategies that depend on the state of the system and time period. Non-stationary transition probabilities in which the probability distribution of change in T and P can vary with time are used to characterize expected non-stationary change in the climate system. Similarly, deriving an optimal policy which identifies unique optimal actions for each state and time period can be used to develop flexible strategies for if, when, and how to add or change infrastructure as we observe climate change over time. The optimal policy, therefore, includes decision rules for whether to add infrastructure given a certain observed T and P in each time period; this defines thresholds for T and P beyond which additional capacity should be added.

Bayesian modeling of climate change uncertainty

To characterize the transition probabilities described above, we extend the Bayesian statistical model presented in Smith et al. (2009) [148]. The statistical model of Smith et al. (2009) treats historical temperature observations X_0 as realizations of a random variable Xwhose unknown mean μ reflects the underlying mean temperature of the current climate. Similarly, future temperature projections are treated as realizations of a different random variable X' whose unknown mean μ reflects the underlying mean temperature of the future climate. Weights or reliabilities of each individual GCM projection j from a set of M models are estimated using the difference between the historical projection X_j and the historical observations; larger differences indicate that X_j must have a higher variance and therefore model j a lower reliability [148].

We extend this approach in three ways. First, we apply the model to P as well as T. We assume that T and P are independent, reflecting that a model's performance in estimating T may be unrelated to its ability to estimate P. Second, we apply the model to observations and projections of change in T and P rather than absolute T and P due to greater model skill in GCM projected changes in temperature and precipitation rather than absolute values [107, 131]. This is especially important in our application in Mombasa where there is less disagreement in temperature change than there is disagreement in hind-casted absolute temperature. Finally, we apply the model to multiple time periods in series. Smith et al. (2009) assumed two periods: a historical climate (1961-1990) and a future

climate (2071-2100). We use pairs of 20-year time periods from 1980 to 2100, in which the "historical" climate corresponds to the time period in the SDP and the "future" climate corresponds to the next 20-year period; this provides the 1-stage transition probabilities needed in the SDP.

Following [148], the statistical model is formulated as follows for ΔT ; an identical model is used for $\Delta \% P$.

$$X_0 \sim N(\mu, \lambda_0^{-1})$$

$$X_{j,t} \sim N(\mu, \lambda_j^{-1})$$

$$X_{j,t+1} | X_{j,t} \sim N(\nu + \beta (X_j - \mu), (\theta \lambda_j)^{-1})$$
(5.2)

where μ , ν , β , θ , and λ_j have prior distributions:

$$\nu, \mu, \beta \sim V(-\infty, \infty)$$
$$\theta \sim G(a, b)$$
$$\lambda_1 \dots \lambda_j \sim G(a_\lambda, b_\lambda)$$
$$a_\lambda, b_\lambda \sim G(a^*, b^*)$$

The joint density of μ , ν , X_0 , $X_{j,t}$, $X_{j,t+1}$, β , θ , a_λ , b_λ , and $\lambda_j \forall j = 1...M$ is therefore proportional to:

$$\theta^{a+M/2-1}e^{-b\theta}e^{(1/2)\lambda_0(X_0-\mu)^2}a_{\lambda}^{a^*-1}e^{-b^*a_{\lambda}}b_{\lambda}^{a^*-1}e^{-b^*b_{\lambda}}$$
$$\times\prod_{j=1}^{M}\left[\frac{b_{\lambda}^{a_{\lambda}}\lambda_j^{a_{\lambda}}e^{-b_{\lambda}\lambda_j}}{\Gamma(a_{\lambda})}e^{-(1/2)\lambda_j(X_{j,t}-\mu)^2-(1/2)\theta\lambda_j\{X_{j,t+1}-\nu-\beta(X_{j,t}-\mu)\}}\right]$$

 X_0 is the observed ΔT in time period t; $X_{j,t}$ is model j's projection of ΔT in the current time period t, and $X_{j,t+1}$ is model j's projection of ΔT in the next time period t + 1. X_0 , $X_{j,t}$, and $X_{j,t+1}$ are treated as observations from unique normal distributions. μ and ν are the underlying means for the 20-year ΔT distributions in the current (t) and future (t + 1)time periods respectively. The goal of the analysis is to estimate a posterior distribution for ν , which will characterize the transition probabilities. We set $a = b = a^* = b^*$ which were chosen so that θ , a_{λ} , and b_{λ} have proper but diffuse priors. In the above distributions, N, U, and G represent normal, uniform, and gamma distributions, respectively. λ_j is the inverse variance of X_j , representing the reliability of model j. β is a regression parameter that introduces correlation between $X_{j,t}$ and $X_{j,t+1}$; it is estimated by the model rather than assumed. θ is also an estimated parameter that enables a model to have different reliability in the future compared to the present. The marginal densities for each of the parameters are estimated using MCMC methods; we use the Gibbs sampling approach and code developed in [148]. More details on the statistical model are available in Smith et al. (2009) [148].

In order to characterize the transition probabilities, the statistical analysis is replicated for each climate state in each time period. The temperature state in the SDP is treated as the observed temperature X_0 in equation 5.2. The resulting posterior marginal distribution for ν is discretized, translated from ΔT and $\Delta \% P$ to absolute values of T and P, and used as the transition probability for that temperature state. This process allows us to simulate learning over time as more climate observations are collected; the transition probabilities in the later time periods incorporate more information about how the climate has changed.

Stochastic weather generation

Climate impacts on river runoff depend on changes in month-to-month variability in precipitation and temperature in addition to changes in the mean. We model these two changes separately. To develop monthly time-series of T and P, we follow the k nearest neighbors (kNN) approach as described in Rajagopalan et al., (1999) applied to GCM projections. This non-parametric statistical approach allows us to impose the mean T and P from the SDP while also capturing the standard deviation in monthly values and month-to-month autocorrelation projected by the GCMs. This approach was chosen for its simplicity and ease of implementation; future studies could use other non-parametric approaches such as the local polynomial regression method developed in [17].

Rainfall-runoff model

Next, the synthetic T and P time series are input to a hydrological model to asses the impacts on runoff. We use CLIRUN II, the latest in a family of hydrological models developed to assess the impact of climate change on runoff [152, 151, 174, 83]. CLIRUN II is a two-layer, conceptual, lumped-watershed rainfall-runoff model. It averages soil parameters over the watershed and models runoff at one gauge station at the mouth of the basin. It can be run on a monthly or daily time step. Using the 100 T and P monthly time series as inputs, CLIRUN II generates 100 monthly time series for runoff.

Infrastructure alternatives and operations

A key outcome of this approach is to assess the value of flexible water supply infrastructure planning in mitigating climate uncertainty. To do this, our approach develops flexible infrastructure alternatives and compares them to static approaches. Two static alternatives are developed: one designed with enough capacity to meet reliability targets under today's climate, and another "robust" alternative designed to meet reliability targets under any of the possible future climates projected by the Bayesian climate uncertainty analysis. Then a flexible alternative is developed in which the smaller amount of capacity sufficient under the current climate is installed initially, but plans are made to be able to expand to the full robust capacity in the future if needed. The performance of these alternatives is evaluated using infrastructure operation models, which estimate the yields obtained from each of the alternatives for each of the generated time series for runoff in the watershed. Yields are compared to demand and reliability targets to estimate unmet demand or water shortages under each possible future climate state. These shortages, in combination with cost estimates for the capital and operating costs of the different infrastructure alternatives, are used to characterize the cost function in the SDP.

Simulated risk profiles

Finally, the SDP results develop a policy for whether to invest in the flexible or static alternative and, if the flexible alternative is chosen, under what climate states it should be expanded. We develop forward simulations for different climate change paths by sampling from the transition probabilities. We use these simulated climate change paths to assess the performance of the different alternatives when they operate according to the policies developed by the SDP. We develop distributions for the performance against key performance metrics including cost and reliability.

5.3 Application to Mwache dam

Background

We demonstrate this method with an application in Mombasa, a coastal city in Kenya. Mombasa is the second largest city in Kenya with an estimated population of 1.1 million [24]. Urban water demand is currently estimated at 150,000 m³/d and expected to grow to 300,000 m³/d by 2035 [119]. Mombasa has a warm, humid climate with average annual precipitation of about 900 mm/y and a mean annual temperature of about 26°C [118]. Precipitation is driven by monsoon winds with two rainy seasons: a long rainy season from April to June with average monthly precipitation of 113 mm/month and a shorter rainy season from October to November with average monthly precipitation of 90 mm/month. The World Bank is financing a dam project on the nearby Mwache River to address water scarcity in the region and expected demand growth [170, 114]. The dam is targeted to supply a firm yield of 186,000 m³/d (67.9 MCM/y) at 90% reliability for urban demand with excess for agricultural supply. The Mwache catchment covers approximately 2250 km² west of Mombasa[25]. Mombasa, the Mwache river, and the proposed dam location are shown in figure 5-2. Mean annual runoff (MAR) is 113 MCM per year [25].

Uncertain projections of climate change in the region make it difficult to assess how large to size the dam in order to meet the yield and reliability targets over its full lifetime. While there is robust agreement across GCMs projecting warming in the region, precipitation response is more uncertain. While Held and Soden (2006) established the well known 'wet-get-wetter, dry-get-drier' hydrological response to global warming, this mechanism was shown to break down over land in the tropics [72]. Uncertainties in the rainfall response in the tropics are driven by thermodynamic changes and large scale dynamical changes such as shifts in convergence zones in response to changes in sea surface temperature [26]. Therefore, there is also substantial uncertainty in whether runoff and yield, which are primarily driven by precipitation in this region, will increase or decrease as well. A previous study commissioned by the World Bank has assessed the climate vulnerability of the Mwache dam [171]. This study uses an ensemble of 121 climate projections from CMIP3 and CMIP5 and applies RDM to evaluate design storage options ranging from 40 MCM to 140 MCM; they recommend designs ranging between 80 and 120 MCM. [171].

Infrastructure alternatives and demand scenarios

We apply the framework developed in section 5.2 to develop a flexible dam design in which extra storage capacity can be added in the future and assess its performance in comparison to static alternatives. We hypothesize that by explicitly addressing the opportunity to learn about climate change over time and update uncertainty estimates, a flexible design will



Figure 5-2: Map of study area including Mwache river, proposed dam location, Mombasa city center.

perform favorably. We apply this framework to two planning scenarios. In the low-demand scenario, we assume a target yield of 150,000 m³/d (54.8 MCM/y) with 90% reliability from the Mwache dam; this yield target reflects current demand and can be fully met by the new dam. We evaluate the two dam sizes proposed by the previous World Bank study [171], 80 MCM and 120 MCM, as well as a flexible alternative in which the height of the small dam can be raised, increasing the reservoir capacity to 120 MCM. Our analysis using CLIRUN II and the reservoir sizing model confirms that the 80 MCM dam meets the reliability targets in the current and expected future climate but does not meet reliability targets if the climate gets substantially warmer and drier. The 120 MCM dam meets reliability targets across all projected future climates, providing a robust alternative. Cost estimates for the small and robust dams were developed using the cost tool from the previous World Banks study [171]. For the flexible dam, the cost per m³ of additional capacity added is assumed to be 50% greater than that of the original capacity.

In a second, high-demand scenario we assume a target yield of $300,000 \text{ m}^3/\text{d}$ (109.6 MCM/v) with 90% reliability, reflecting the potential for rapid demand growth in the next decade. This high value of demand is consistent with 2035 projections from [119]. In this scenario, the target yield is greater than the mean annual runoff in the Mwache river and therefore the dam cannot meet the target yield regardless of its size. We model the combination of a 120 MCM dam and a desalination plant that is used to supply demand when reservoir storage is low. Three desalination alternatives are chosen, analogous to the dam design alternatives. A low capacity alternative designed to meet reliability targets in the current and expected future climate has 60 MCM of capacity; the robust alternative that meets the reliability targets across all projected future climates has 80 MCM of capacity; a flexible alternative starts with 60 MCM and can be expanded to 80 MCM. Capex and opex estimates for the desalination plans were developed using the Cost Estimator tool from DesalData [56]. Evaluating this second scenario allows us to compare the value of flexibility across two technology options, earthen dams and desalination, which have unique water supply profiles and cost structures. These planning scenarios, and the cost and capacity of the infrastructure considered in each, is summarized in Table 5.1.

Demand Scenario		Technology	DR	Capacity [MCM]			Capex [M\$]			
				Small	Large	Small	Large	Exp	$\operatorname{Flex} + \operatorname{Exp}$	
a	Low	Earthen dam	3%	80	120	76.5	99.2	49.6	148.8	
b	Low	Earthen dam	0%	80	120	76.5	99.2	49.6	148.8	
с	High	RO desalination	0%	60	80	183.1	232.2	72.4	255.5	

Table 5.1: Key planning scenarios and corresponding infrastructure evaluated

Climate and hydrological analysis

To implement the Bayesian uncertainty analysis in Mombasa, we use a total of 21 CMIP-5 members whose modeling group and model run are included in the appendix. For each GCM, monthly temperature and precipitation values are averaged over 2°S to 6°S and 38°E to 42°E, overlaying the Mwache catchment; GCM projections are regridded from their original resolution following the approach in Boehlert (2015) [18]. The same is done for the observed climate, where monthly values are taken from the Climate Research Unit (CRU) dataset version TS.3.21 [118]. The analysis is repeated for the five 20-year time periods starting with 2001-2020 for t=1 and ending with 2081-2100 corresponding to t=5 in the SDP. The 20-year time interval was chosen so that year-to-year variability was not driving the trend in precipitation and temperature across time periods.

The hydrological model CLIRUN II is calibrated using 14 years of monthly streamflow data. Only one streamflow gauge, RGS 3MA03, is available in the Mwache basin [25]. However, it is directly upstream of the dam location, making it representative for this study. The same monthly temperature and precipitation data from CRU used in the Bayesian climate analysis is used to calibrate CLIRUN II for consistency. This temperature and precipitation data is different than the local data used in the previous World Bank study [171], leading to different calibration results but similar performance (historical MAR: 113 MCM/y; World Bank MAR: 133 MCM/y; our MAR: 103 MCM/y). The infrastructure operation model includes dam operations (and desalination operations when necessary) that seek to meet the specified yield target while accounting for dead storage, net evaporation, and environmental flows. Unmet demand is measured for each of the 100 streamflow time series, and the average 20-year unmet demand is used to characterize U in the SDP formulation in equation 5.3. More details on the implementation of CLIRUN II and the operation model are in the appendix.

SDP formulation

These two planning scenarios are modeled using the SDP framework depicted in figure 5-1. The components of the SDP shown in equation 5.1 are formulated as follows:

$$S = \{T(t), P(t), Z(t)\}$$

$$A = e(Z, t)$$

$$C = I(T, P, Z, e, t) + D * U(T, P, Z, e, t)$$
(5.3)

where

- $t \in \{1...5\}$ is a 20-year time period ranging from 2001-2020 for t = 1 to 2081-2100 for t = 5
- T(t) is the mean temperature in °C in time period t, ranging from 25 to 33 at 0.05°C increments.
- P(t) is the mean precipitation in mm/month in time period t, ranging from 66 to 97 at 1 mm/month increments.
- Z(t) ∈ {1...4} is the available infrastructure, in which the states correspond to a small infrastructure alternative, large infrastructure alternative, flexible unexpanded alternative, and flexible expanded alternative, respectively. The infrastructure alternatives are either a set of dams or a set of desalination plants.
- $e(Z,t) \in \{0...4\}$ is the choice of infrastructure in which 0 is no change, 1 is a small alternative, 2 is a large/robust alternative, 3 is a flexible alternative, and 4 is the expansion of the flexible alternative. The alternatives include a set of dams or a set of desalination plants. The choices are constrained by time period and available infrastructure such that $e(Z, t = 1) \in \{1, 2, 3\} \forall Z$; $e\{Z, t\} \in \{0, 4\} \forall t = 2...5, Z = 3$; and $e\{Z, t\} \in \{0\} \forall t = 2...5, Z = 1, 2, 4$
- *I* is the cost of the infrastructure including capital costs (capex) and operating costs (opex). Desalination opex is a function of the water produced in each time period.
- D is unit cost of damages incurred for unmet water demand, set at 15 \$ /m³ in our base case based on estimates of water productivity in Kenya from the World Bank [158].

• U is the volume of unmet demand as a function of the climate states, existing infrastructure, and any new infrastructure brought online in time t. U=0 in t=1, reflecting that t=1 is a planning and construction period and performance is not measured until the beginning of the second 20-year time period.

5.4 Results

Here we present results from the application of the modeling framework to Mombasa. First, we show results from the Bayesian uncertainty analysis demonstrating the reduction in uncertainty compared to a democratic weighting and the potential to reduce uncertainty by incorporating new information over time. Then we present results from the SDP showing the flexible policy for the SDP and its performance in comparison to the two static alternatives. We first present these results for our base case (scenario a: low demand with discounting) and then discuss how the results change without discounting and with desalination in the high demand scenario.

Bayesian climate uncertainty analysis

Figure 5-3 shows historical T and P from CRU and projected change in T and P from each of the GCMs. Then, confidence interval (CI) using first the IPCC democratic weightings in which all models are assumed to perform equally well are compared to CIs generated from the Bayesian uncertainty analysis for the first time period. We see that the CIs from the Bayesian analysis have a smaller range of uncertainty in comparison both the full range of the individual projections as well as the democratic weighting. This is because the Bayesian analysis puts most of the weight on a subset of models that match historical change in this area better than the others.

The process of updating uncertainty over time using the Bayesian analysis from later time periods is shown in Figure 5-4. We present one sample time series of how the climate states T and P could evolve over time. For each simulated observation in the time series, we use the transition probabilities to simulate 10,000 time series starting at the current observation and going to the end of the planning period. These time series are used to construct a 99% CI. This process is repeated for each time period, with darker colors in the plot corresponding to the CIs developed with more observations later in the planning period. This approach


Figure 5-3: Modeled and observed temperature (left) and precipitation (right) relative to 1990 values with uncertainty estimates. The gray lines are 20-year moving averages of GCM simulations in the grid cells over Mombasa. The purple (left) and green (right) shaded regions show CIs using the democratic IPCC method for characterizing uncertainty plotted at 20-year increments. The orange (left) and blue (right) shaded regions show the 90% CI developed using the Bayesian uncertainty method applied in the first time period. The Bayesian CIs are narrower than the democratic CIs, demonstrating the ability to reduce uncertainty by taking advantage of the additional information that some models perform better in this region than others.

allows to compare how the uncertainty at the end of the planning period in 2100 changes as more observations about temperature and precipitation are collected over time. The top half of Figure 5-4 shows this process for temperature and precipitation, for which the simulated observations are sampled independently. In both cases, the CIs narrow over time, demonstrating the value of more information in reducing predictive uncertainty. Sometimes, however, a new observation does not reduce the uncertainty at the end of the planning period. This is seen with 2010 temperature estimate; the updated CI completely overlaps the previous. This suggests that value of information can be limited by the noise inherent in the statistical model, especially when the simulated observation is close to the center of the previous CI. Figure 5-4 shows a single sample time series of potential observations; a different time series, for example, may show a decreasing rather than increasing precipitation trend, in which case the precipitation CI would narrow closer to the bottom of the initial uncertainty range.

The bottom half of Figure 5-4 shows how the same simulated temperature and precipitation observations update uncertainty in MAR and water shortages. MAR correlates closely with precipitation, especially in the first half of the planning period. In the second half of the planning period, more rapid temperature increases offset modest increases in P, showing a slight decline in MAR. The learning profile is similar to that of P. Mean annual water shortages, at bottom right, are measured against a 90% monthly reliability goal: the first 10% of shortages in any month do not incur damages. This plot assumes the low-demand scenario (55 MCM/y) with the small dam (80 MCM of reservoir capacity). Here we see strongly asymmetric uncertainty; the lower bound of the CI is 0. This reflects the lowprobability, high-severity risk of droughts; shortages occur only when runoff is substantially below MAR for several months or years in a row. In this sample time series, the modest increase in temperature and relative stability of MAR are enough to substantially reduce the 90th percentile risk over time. An alternate version of this figure is shown in Appendix C in which small decreases in P in combination with high warming drives MAR down and the risk of shortages substantially higher. Across many different simulated T and P observations we find a similar trend of narrowing of uncertainty, regardless of the direction of change, demonstrating a robust high value of information.



Learning over time: 99% Cls

Figure 5-4: One sample realization of Bayesian learning over time. Black dots represent a single simulated time series for potential future climate observations. These simulated observations are used to develop and update probabilistic estimates for change represented using 99% confidence intervals for projected change in T, P, MAR, and water shortages. In this example, modest increases in P lead to stable MAR, allowing us to eliminate the risk of high shortage values by the end of the century. An alternate simulated time series with decreasing P and increasing shortage risk is shown in the appendix. Across many simulated time series we find robust high value of information, demonstrated by reduction in CI range.

SDP

The previous analysis demonstrated the ability of the Bayesian uncertainty analysis to reduce uncertainty over time through the use of simulated potential future observations of T and P. These simulated observations correspond to the states of the SDP. In this way the SDP accounts for all the possible future observations and their probabilities and chooses an optimal strategy based on all the possible learning outcomes. This optimal strategy results in two main outcomes, shown in Figure 5-5. In the first time period, shown in panel a) at left, the SDP develops a threshold as a function of T and P in the first 20-decade time period beyond which the large robust infrastructure is optimal; lower than this threshold the flexible alternative performs better. This is because the relatively small cost difference between the flexible and large dam means that if the risk of shortages at the outset is high enough, it is better to invest in the robust option upfront. In time periods 2 through 5, the SDP policy results in a T and P threshold beyond which the flexible infrastructure capacity is expanded if available; this is shown on the right side of the figure. Similar to the first period policy, expanding infrastructure capacity is optimal in drier and warmer states. The threshold moves somewhat left in the plots in the later time periods. This is because a modest shift towards warmer and drier climate in the early time period signals that further warming and drying is expected. The same modest shift in later time periods indicates a slower rate of warming and drying, suggesting that the risk of rapid transition to a state where water shortages are likely is lower.

Forward simulation

We now present results showing how the flexible strategy developed by the SDP performs against the static alternatives. The transition probabilities are used to simulate 1000 time series for the climate states, and the optimal policies are used to assess how the alternatives would perform in each time series. Figure 5-5 shows histograms for the decisions made under the optimal policies across the 1000 simulations; the top row corresponds to the low-demand scenario a) discussed here. At top left, we see that the flexible alternative is chosen in 90% of simulations under the low-demand dam scenario with a base case discount rate of 3%. At top right, we see that when the flexible alternative is chosen, the option to expand is never expanded in about 70% of simulations, highlighting the relatively low



b) Expansion Policy for Flexible Dam

Figure 5-5: Optimal policies from SDP results. Panel a) at left shows the initial dam decision and the four panels b) at right show the policy for exercising the option to increase height of dam. Results are shown for the low demand dam scenario under 3% discount rate. The robust dam is chosen for hotter, drier climates in the first time period. Similarly, the flexible option is exercised if the climate transitions to a hot, dry climate in later time periods. The expansion threshold shifts in later decades. This is because a modestly hotter and drier climate in early time periods signals more rapid change in the coming decades through the learning process.

probability risk of transition to a climate that is substantial drier enough to lead to shortages beyond 10% of demand. The time period at which it is exercised varies; sometimes rapid warming and drying leads to an early expansion decision while sometimes the change is more gradual. There are relatively few expansions in the final time period. This reflects a known limitation of finite-horizon SDP formulations; they discourage investment toward the end of the planning period because there is limited time to reap the benefits of those investments. In reality, however, those investments would bring benefits after the end of the planning period.

Finally, we compare the performance of the flexible strategy following the optimal policy with the two static alternatives against the objective function C of the SDP in equation 5.3. Figure 5-7 shows CDFs of the total cost of each alternative across the 1000 forward simulations; the top row shows planning scenario a) discussed here. The large robust static alternative, shown in blue, has the same cost (equal to the capital cost of the dam) across all simulations; as designed, no shortage damages are incurred in any of the simulated future climates. The small dam, shown in orange, has the most variable costs. It performs better than the large dam in about 70% of simulations, but has substantially higher costs in 30% of simulations due to large damages from water shortages. The flexible alternative has the same cost as the small dam in close to 70% of simulations, but the reliability risk is substantially mitigated because of the potential to expand. The high-end costs are higher than the robust alternative because 1) the cost of building the 80 MCM dam and expanding to 120 MCM is higher than building the 120 MCM dam upfront and 2) sometimes the dam is not expanded even when modest water shortages are incurred. The ability of the flexible alternative to mitigate both the risk of overbuilding and the risk of severe shortages demonstrates the high value of flexibility in this case.

Alternative scenarios

We now assess how the value of flexibility changes 1) without discounting and 2) under the high demand scenario. Results for these two scenarios are shown in panels b) and c) of Figures 5-6 and 5-7. We see in the low demand, no discounting scenario a) that the value of flexibility in our base case was largely driven by discounting, which incentivizes delays in capital investments. With a 3% discount rate, when the flexible dam is expanded 40 years in the future, the cost of flexible dam plus is expansion is 4% more expensive than building



Simulated Infrastructure Decisions by Planning Scenario

Figure 5-6: Histogram for infrastructure decisions. Top row: scenario a) low demand with discounting; middle row: scenario b) low demand without discounting; bottom row: scenario c) high demand without discounting. In each row, the left plot shows the initial decision of which alternative to choose in the first time period. The right plot shows, when the flexible dam is available, how often it is expanded and when it is expanded. The flexible alternative is chosen most often in scenario a) because discounting incentivizes delayed capital investments but not in b) because large economies of scale incentivize a single, large investment. In scenario c), more modest economies of scale lead to high value of flexibility in the absence of discounting. Across all scenarios, the flexible dam is expanded in no more than a third of simulations, highlighting the low probability of risk of reaching a climate that is hot and dry enough to incur substantial shortages.



CDF of Total Cost by Planning Scenario

Figure 5-7: CDF of total cost for each infrastructure alternative by planning scenario. Top: scenario a) low demand with discounting; middle: scenario b) low demand without discounting; bottom: scenario c) high demand without discounting. Each plot compares the small and large static alternatives, shown in blue and red, compared to the flexible alternative shown in yellow. The ability of the flexible alternative to reduce risk of over building is demonstrated by its low cost in the low percentiles in all three scenarios. The flexible alternative, demonstrated in the high percentiles. However, the median cost of the flexible alternative is higher than the large robust alternative in scenario b) due to substantial economies of scale.

the large dam upfront, compared to 27% more expensive without discounting. Without discounting, the robust dam looks more favorable; it performs best in 60% of simulations, has no cost variability risk, and is chosen upfront in about 75% of simulations. The low value of flexibility without discounting is driven by the cost structure of the technology; large economies of scale in the dam mean that an 120 MCM is only 30% more expensive than an 80 MCM dam for 50% additional capacity. This suggests it is often better to build the large dam upfront even if there is a relatively low probability that it will be needed. The high-demand scenario in which a 120 MCM dam is combined with a desalination plant shows a high value of flexibility even without discounting. The flexible alternative is chosen upfront in over 80% of forward simulations, and the CDF demonstrates that it outperforms the static alternatives by substantially mitigating the over build risk in comparison to the robust alternative and also modestly reducing the shortage damage risk in comparison to the small alternative. While the flexible alternative only reduces cost at the 90th percentile and above, this substantially reduces the expected value as the maximum cost of the small plant reaches almost M\$400 (truncated on plot). In this case, the economies of scale for a desalination plant are much less substantial. These results highlight differences in the value of flexibility across technologies.

5.5 Discussion

The results in the Mombasa application demonstrate both the value of information and flexibility in managing climate change uncertainty as well as its limitations. In the lowdemand scenario we see a high value of flexibility when costs are discounted, but limited value of flexibility without discounting. In the high-demand scenario we have high value of flexibility regardless of discounting. We measure the value of flexibility by assessing not only the expected value of performance against the two objectives cost and reliability, but the value-at-risk and value-at-gain as measured by the 10th and 90th percentiles respectively. This highlights that although the uncertainty and learning is driven by the climate system, decisions about whether flexibility is a valuable tool in mitigating uncertainty are strongly influenced by technology choice and cost structure. Large economies of scale make flexibility less valuable; it is better to choose a robust alternative when it is not much more expensive to do so. However, delaying capital investments also provides financial value, which can be especially impactful in resource-scarce areas where unused capital could support other critical infrastructure services.

The value of flexibility is also driven by opportunities for learning about future risk of water shortages by incorporating new climate observations. The planning framework developed here accounts for learning in both the evolution of the state of the system over time and also for learning in how the risk of being in a certain system state changes over time. For example, being in a climate with mean T of 28° C and mean P of 70 mm/month poses a greater risk to future water shortages if that state has been reached rapidly over a short time period. The use of multi-stage stochastic planning can be used to explicitly model learning about uncertainty through the use of non-stationary transition probabilities characterized by the Bayesian climate uncertainty analysis. It also accounts for risk that depends on the state of the system by developing an infrastructure policy that varies with the current state of the system. By enabling shortage risk to vary with both the state of the system and with dynamic uncertainty estimates, our approach captures the full value of flexibility. This approach treats flexibility as a strategy to achieve performance metrics of interest, such as cost and reliability, rather than an end goal itself. This provides more directly useful information to planners who must decide whether upfront investments in flexibility are worthwhile. Future extensions to other applications which have differences in degree and nature of uncertainty, hydrological sensitivity to climate change, and infrastructure alternatives available are likely to have different value of flexibility. This approach can enable planners in these unique settings to assess what types of flexibility if any should be pursued proactively.

Using SDP in combination with EOA enables us to both develop policies for exercising flexible alternatives and evaluate those alternatives using a simulation model. We acknowledge that SDP has limitations for real world planning applications due to its computational expense, which scales exponentially with the size of the problem. In our application, however, the state space is comprised only of long-term climate variables and infrastructure capacity, limiting the size of the problem. To embed the performance using high-fidelity physical models, a hydrological model and infrastructure operation model is run separately for each state in the SDP and the expected value of its performance is saved and used to characterize the SDP cost function. Using a differentiated approach in which uncertainties with high learning potential use a multi-stage stochastic planning approach while others are addressed using more computationally efficient methods can further address this concern when additional uncertainties are incorporated. For example, short-term adaptive operations could be incorporated into this framework in the infrastructure operation model without expanding the state space of the SDP. The limitations of expected utility maximization could be addressed through the use of alternative decision criteria such as those developed by McInerney et al. (2012) [103]. Future work could also use a screening model to optimize the choice of the static and flexible alternatives rather than the heuristic approach used to select the alternatives here. Finally, the approach could be extended to address a more comprehensive consideration of multi-objective planning.

Our approach takes advantage of recent developments in assessing the uncertainty in climate change projections. CMIP5 and related model comparison projects have enabled more sophisticated, probabilistic approaches to quantifying climate change uncertainty in ways that recognize that all models do not perform equally in all parts of the world. We adapt the Bayesian uncertainty modeling approach of Smith et al. 2009 [148] to 1) address change in both T and P, 2) update uncertainty based on potential observations in the future, and 3) use change in mean T and P from one 20-year period to the next, rather than absolute T and P, to reflect greater model skill in predicting changes rather than absolute temperature at the regional level. This approach does have limitations. It assumes that GCMs are independent of one another, when in fact some models borrow entire components from other models [156]. Additionally, we are simulating the potential to learn in the future using only models available today; repeating the analysis in 40 years with a broader range of models reflecting the new state of the science may produce larger shifts in CIs. However, this approach is the best available to assess learning in the future, which impacts planning decisions today. It enables a more precise, validated measure of uncertainty in comparison to the democratic approach used by the IPCC.

Our approach addresses only model uncertainty and not emissions uncertainty. This is an important contribution because model uncertainty dominates overall precipitation uncertainty in the long-term [70], and the lack of political influence on model uncertainty and availability of valid statistical approaches makes a statistical rather than scenario-driven approach appropriate. Future work could extend the approach to address emissions uncertainty by repeating the analysis under different RCPs and assessing regret and robustness of each alternative across scenarios. This framework therefore enables a differentiated approach to climate change uncertainty, in which different types of climate uncertainty are addressed with a unique and appropriate modeling approach.

Chapter 6

Discussion

6.1 Methodological contributions

Developing and assessing water supply infrastructure plans requires an integrated approach that addresses many concurrent sources of uncertainty in the natural, built, and human environments and their impacts on infrastructure. This dissertation makes a contribution to the literature on water supply infrastructure planning under uncertainty through the development of a novel planning framework and its application to develop insights about the potential for flexible infrastructure approaches to mitigate uncertainty in three contrasting planning problems. The planning framework makes several methodological contributions, summarized in Table 6.1

First, the planning framework uses and extends existing dimensions of uncertainty that are important for how uncertainty is modeled to classify uncertainties and match them to appropriate uncertainty analysis tools that are integrated into a single analysis. This addresses a tendency in several current approaches to address all uncertainties in the same way. For example, RDM treats all uncertainties as deep uncertainties and relies on scenariobased approaches; this does not take advantage of the full information available that can be used to characterize some uncertainties probabilistically. Similarly, some uncertainties can be substantially updated as more information becomes available, requiring a dynamic approach; others cannot and are therefore more efficiently addressed using Monte Carlo simulation, which takes a static approach to uncertainty.

Second, it addresses uncertainties with high learning potential — or those for which additional information can feasibly be collected presently or in the future to meaningfully

Contribution	Description		
Uncertainty classification	Framework classifies uncertainties according to two dimensions of uncertainty important for modeling and maps them to appropriate uncertainty analysis methods.		
Dynamic uncertainty analysis and management	Bayesian inference combined with multi-stage stochastic programming enables planner to assess the potential to learn about uncertainty in the future and develop planning strateg that account for future learning in decision-making today.		
Framework to test value of flexibility	Combines engineering options analysis with hydrological modeling to measure value of flexibility as a planning objective		
Parameter uncertainty analysis using statistical surrogate	Artificial neural network maps any combination of uncertain parameters to model predictions, enabling computational efficiency necessary to embed in a stochastic dynamic program.		
Dynamic climate model uncertainty analysis	Bayesian modeling using CMIP-5 extended to update initial weightings as additional climate observation data becomes available.		

Table 6.1: Key methodological contributions of this dissertation

update or reduce uncertainty — using a dynamic approach in which the uncertainty associated with a certain state of the system can change over time as more information is collected. These dynamic uncertainty estimates are developed by applying Bayesian inference to climate models and hydrological models. The hydrological state of the system in the planning model is used as a hypothetical, future observation used to develop a posterior distribution for uncertainty that takes advantage of this new information. By using a multi-stage stochastic planning approach which accounts for all possible future states — as well as the probabilities of reaching those states — our approach develops optimal policies for exercising flexible options that takes into account the potential to learn in the future. Together, these elements achieve an integrated infrastructure planning approaches that combines three elements important for assessing infrastructure planning that had not previously been integrated: 1) dynamic assessments of uncertainty using 2) high-fidelity models of the climate and hydrological cycle to 3) assess infrastructure performance and planning.

Finally, the framework uses engineering options analysis in order to assess the value of flexibility in mitigating the impacts of uncertainty. Previous applications of EOA in water supply have addressed demand uncertainties or made simplified assumptions about supply uncertainty [37]. Here, we integrate uncertainty estimates using hydrological models with EOA to assess the value of flexible planning and design to address hydrological uncertainty. Some recent work has included flexibility as an objective in a planning model [12]; we provide an approach to assess the value of flexibility as a strategy for achieving other planning, rather than assuming it is a worthwhile goal.

The individual applications of this planning make additional methodological contributions, notably in the integration of physical climate and hydrological models into stochastic programs using statistical approaches to overcome computational limitations. Chapter 4 on parameter uncertainty in groundwater in Riyadh, Saudi Arabia develops a statistical surrogate model of a numerical finite-difference groundwater model using an artificial neural network that provides a mapping of any combination of parameters values to predictions of hydraulic head. This enables computationally tractable numerical calculation of a Bayesian posterior distribution for each of value of the state space. While demonstrated on a simple, 2D MODFLOW groundwater model, the flexibility of artificial neural networks, which have been widely applied in hydrological applications, enables extension to more complex groundwater models in the future. This approach could have applications to other engineering systems domains in which complex physical models are needed to appropriately characterize uncertainty in stochastic planning approaches.

In Chapter 5 on model uncertainty in climate change in Mombasa, Kenya, a similar approach is developed to integrate climate uncertainty into a multi-stage stochastic program. However, the key uncertainty addresses is a model uncertainty rather than a parameter uncertainty. Therefore the approach used in Chapter 4 to map uncertain parameter values to predictions using a single model is not appropriate. Instead, we applied Bayesian model averaging across an ensemble of climate model projections. This approach takes advantage of recent advancements in developing probabilistic climate projection using model intercomparison. To our knowledge, it is the first time these probabilistic climate projections have been used to characterize a stochastic program to assess the impacts of climate change uncertainty on water resources planning.

6.2 Cross-application insights

The individual applications addressed in this dissertation provide insights for planners in each region on the potential for flexibility in infrastructure planning and design to mitigate uncertainty in water supply planning; these insights are discussed in each chapter. Some additional insights about flexibility in water supply infrastructure planning can be drawn by looking across the three applications. In all three applications, we find value in flexible planning or design as means to mitigate uncertainty as well as limitations to the value of flexibility in all cases. We can make some observations about factors that drive value in flexibility in water supply planning.

Learning In all cases, we observe that the process of learning about uncertainty over time is useful in exercising flexible options. This is demonstrated by the result that the flexible option is only exercised in a subset of forward simulations. In Melbourne, the flexible option is exercised primarily in scenarios with high demand growth, after high demand growth has been observed in the first planning period (check on details of this). Similarly, in Riyadh and Mombasa, the flexible expansion option is exercised in response to new observation of rapid head drawdown and rapid drying and warming respectively.

Stochastic variability A challenge in the ability of flexible infrastructure planning to mitigate uncertainty is the presence of irreducible stochastic variability. In Melbourne, while the small flexible alternative has clear advantages over the large desalination plant, it does incur more reliability risk as it does not fully mitigate the impacts of multi-year droughts. In Mombasa, similarly, the flexible alternative is able to mitigate the worst reliability outages, but it still incurs substantially more shortage damages than the large robust alternative. This is not the case in Riyadh where stochastic variability does not play a role: the only uncertainty addressed is an information uncertainty. In Riyadh, therefore, the flexible alternative has a nearly identical reliability profile to that of the static build alternative. This is because the expansion policy derived by the SDP is effective in making sure short-term reliability outages do not occur in the two years between when the decision is made and the capacity comes online — and because there is no variability in supply or demand.

Value of reliability The value of reliability, operationalized in this dissertation as a penalty incurred for water shortages in which demand targets are not met, may or may not have a large impact on the value of flexibility. In the Melbourne application, the shortage penalty is treated as a deep uncertainty and addressed with scenario analysis. The results show that the value of flexibility is highly dependent on this value. In the base case of 25

 m^3 the flexible alternative performs well. However, when the shortage value is lowered to 5 m^3 or raised above 50 m^3 , the no-build alternative and large plant alternative respectively perform more favorably. In Riyadh, however, the value of flexibility is insensitive to the shortage value. This is related to lack of stochastic variability as discussed above; the flexible alternative almost never incurs shortage damages, and even a most shortage value of $5/m^3$, down from the base case of $25/m^3$ based on World Bank estimates of water productivity, is enough to promote the flexible alternative over the no-build alternative. In Mombasa, the value of flexibility is also insensitive to the shortage penalty. This is because of the substantially lower capital costs, such that total cost of the large dam is equal to 6.6 MCM of water shortages at the base shortage value of $15/m^3$. This means that the optimal SDP policy is incentivized to avoid even small reliability outages. The value of reliability therefore can have a large or modest impact on the value of flexibility, depending on the cost of the infrastructure and the probability of reliability outages.

Discounting We also find a large influence of the discount rate on the value of flexibility. Higher discount rates incentivize delayed capital investments and therefore promote flexible options that defer some or all of the capacity until later time periods. In the applications in Melbourne and Riyadh, we found high value of flexibility across all discount rate values tested. In Mombasa, however, in the scenario based on current demand in which only a dam is built, rather than a dam plus a desalination plant, we find that the flexible alternative has little value without a discount rate.

Economies of scale A key advantage of large, traditional infrastructure project is ability to leverage economies of scale; the unit cost of capacity decreases as the size of the project increases. Modular infrastructure projects, in which smaller volumes of capacity are added as needed, therefore poses a tradeoff between the ability to adapt and the inability to access economies of scale. In Mombasa, the key reason that the flexible dam design has limited value of flexibility in comparison to the large, robust dam is because of the high economies of scale for an earth dam: the large dam is only 27% more expensive for 50% more capacity. The economies of scale for reverse osmosis desalination, however, are much lower because the technology is inherently modular. Therefore, the applications focused on desalination planning rather than dam design faced a much more modest tradeoff. This suggests that

water planners choosing between different water supply technology options may want to consider the economies of scale each option faces and whether the technology can effectively take advantage of flexibility.

Type of flexibility The types of flexibility addressed in the applications varies. Chapter 4 evaluates a flexible infrastructure *planning* process in which the *timing* of capacity additions is flexible, while Chapters 3 and 5 assess flexible infrastructure *design* in which the *volume* of capacity additions is flexible. These were chosen to appropriately manage the uncertainties in each case. In the aquifer depletion problem in Chapter 4, it is known that new infrastructure will have to be developed — the question is when. By contrast, in Chapters 3 and 5, it is assumed that new capacity is needed immediately — the question is how much. The specific formulation of flexibility impacts its value. For example, the essential element for success of the flexibility in Chapter 4 is advance preparations, such as choosing a design and location in advance, such that the infrastructure can be brought online in only two years. This makes the risk of reliability much lower. The ability of planners to effectively prepare in advance, enabling them to execute flexible policies quickly and nimbly will have a substantial impact on the potential for flexible approaches to reduce cost risk without impacting reliability.

6.3 Limitations and extensions

Many opportunities exist to improve and extend the approach to flexible water supply infrastructure planning developed in this dissertation. Areas for future work include: the planning framework, planning applications, water resource modeling, social and institutional analysis and other domains. Key opportunities for extensions in each of these areas are summarized in Table 6.2.

Planning framework The planning framework makes a contribution in classifying uncertainties and linking them to appropriate uncertainty analysis tools. There are opportunities to further develop other dimensions of uncertainty and to refine the operationalization of the existing categories. Many dimensions and taxonomies of uncertainty exist beyond those discussed in this dissertation; future work could build theory around which dimensions of uncertainty are important for modeling and managing uncertainty, and whether disagreements on this topic in the literature are based on subjective choices or logical errors [116].

Table 6.2: Key opportunities for extensions and future wo	rk
---	----

Category	Extension opportunities		
	Refine operationalization of uncertainty dimensions		
Planning framework	Test additional uncertainty management strategies		
	Add additional uncertainties and categories as needed		
	Optimize value of flexibility		
Planning applications	Improve multi-objective analysis		
	Scale to large systems via integration with		
	simulation-based optimization approaches		
	Identify properties of water systems that drive value of flexibility		
Water resource modeling	Extend to other water resources planning problems e.g. hydropower, irrigation, thermal cooling, and flooding		
	Address additional measures of climate change beyond long-term mean T and P e.g. extreme events, autocorrelation, variance		
Social and institutional analysis	Use multi-agent modeling to address coupled interactions of planners, stakeholders, and end users		
	Integrate stakeholder collaboration to identify and evaluate uncertainties and opportunities for social learning		
Other domains	Extend to other engineering systems infrastructure domains e.g. electricity, transportation		
	Guide and evaluation water technology development		

There is substantial debate in the community as to what constitutes deep uncertainty, for example. Further work could explore literature in information theory or the philosophy of science on interpretations of probability to build logical arguments supporting a specific definition.

Opportunities exist to expand on and refine the "learning potential" dimension of uncertainty in future work. In this dissertation, we address only statistical uncertainties with high learning potential and used Bayesian inference to model the potential for future observations to update or reduce uncertainties. In reality, deep uncertainties can present opportunities for learning as well. For example, scenario or emissions uncertainty in climate change is a one of the most commonly cited deep uncertainties, and our understanding of it will certainty change and update over time. Scenario-based approaches for modeling the impacts of this learning process could better enable evaluation of adaptive infrastructure and management approaches to mitigating climate change uncertainty. Additionally, a limitation of the current implementation for statistical uncertainties is that they use only the existing models that are currently available; learning in the future may take the form of new model development.

Additionally, while the framework currently provides a means to test the value of flexibility in managing uncertainty through the use of engineering options analysis, it could be expanded to address other strategies for managing uncertainty as well. For example, information collection could be addressed as a decision variable in which planners can seek out additional information upfront rather than passively collecting it over time. Remotelysensed earth observation data, such as NASA satellite missions GRACE and its follow on mission GRACE-FO as well as SMAP, present opportunities for new sources of information to reduce hydrological uncertainties in planning decisions.

Planning applications There are a number of practical opportunities to improve the ability of the framework to realistically model planning decisions. For example, further computational advancements may be needed in order to include several high learning potential uncertainties in a single analysis. Additionally, in this dissertation we have not attempted to optimize the choice of static and flexible alternatives; we use the SDP to develop the optimal policy for exercising the option but not the best choice of flexibility. Instead, we have relied on planner-suggested alternatives and heuristic approaches to choose

the alternatives to compare. In practice, the specific choices of static and flexible alternative influences the relative value of flexibility. Future work could use screening models [11] or other optimization approaches to compare optimal static and flexible alternatives rather than heuristic-based alternatives.

Future work could also consider a wider range of infrastructure or other planning alternatives. For example, Chapter 4 in Saudi Arabia only evaluated a RO desalination plant. However, wastewater treatment for reuse and additional groundwater development could also be used to increase supply capacity. Across all the applications, demand-side measures to reduce water use through restrictions and incentives could be considered as an alternative to infrastructure development. Effective demand reduction measures are likely to play an important role as uncertainty increases in many water resource systems as a method to manage uncertainty. Integrating demand reduction into the planning framework would require the addition of economic models to asses the impact of — and uncertainty in — human responses to demand policies such as increased tariffs. This could leverage recent work integrating agent-based models into water resource system models to assess social responses to different policies.

Finally, we have only considered two planning objectives in these applications: reliability and cost. Planners have many other objectives as well, such as maintaining ecosystem services, equitable distribution and access, water quality, etc. One notable objective not addressed in this dissertation is the energy intensity of water supply; this is especially important in evaluating desalination investments which are highly energy intensive. Saudi Arabia is aiming to reduce the amount of its energy resource consumed domestically, and evaluating the energy intensity of new investments options is likely to promote options such as wastewater reuse with lower energy use. In order to implement this — as well as the proposed extensions to address a larger number of alternatives in larger scale systems future work could explore the use of multi-objective evolutionary algorithms [136]. These and other search algorithms that enable simulation-based optimization have been widely used to find Pareto optimal solutions across many planning objectives in large-scale systems more tractably that traditional optimization approaches.

Water resource modeling Future work can also extend the approaches developed here to integrate more complex hydrological models. Groundwater applications can include spatial heterogeneity — a different form of stochastic variability that exists in space rather than time — using a zonal approach. The ANN could be applied using additional input parameters for each of the zones. The ANN could also be used to predict head in multiple wells, which could help identify the shape of the cone of depression and potentially reduce parameter uncertainty more quickly. Additional extensions could apply to unconfined, renewable aquifers with seasonal variation and greater interaction with surface water. This poses a challenge to the assumption of path independence because hydraulic head would not be monotonically decreasing; new planning model formulations could take a hierarchical approach in which annual and seasonal variations are addressed as separate state variables.

The Bayesian climate modeling could also be extended to address uncertain climate variables beyond decadal mean temperature and precipitation. Water planning decisions are also impacted by changes in monthly and annual autocorrelation (e.g. to assess changes in the frequency of multi-year droughts) and variance in T and P. The Bayesian methods developed here could be extended to develop uncertainty estimates for these additional climate metrics and integrated into new stochastic weather generation approaches.

While this dissertation focuses on water supply planning, the planning framework could be applied to other planning questions in water resources. For example, in dam design for hydropower, we could evaluate the potential to learn about electricity demand over time and assess the value of flexible design. Similar extensions could be made in water use for thermal electric cooling and irrigation. The climate analysis could be extended to assess learning about the frequency of extreme events and used to inform flood planning decisions. Similarly, uncertainty in ecological indicators, such as the frequency and duration of low environmental flows, could be incorporated. Long-term future work could aim to integrate these water resources domains to assess learning and flexibility in an integrated water resources management approach.

Finally, application of this method across many different water systems could enable identification of properties of water systems which enable high value of flexibility. For example, the results from the three applications in this dissertation suggest that the relative influence of stochastic variability vs. information uncertainty has a strong influence on the value of flexibility. Designing a large-N study across many water systems could identify and test metrics like the ratio of stochastic variability to information uncertainty to see if they have statistically significant impacts on the value of flexibility. Different regions of the world face different ratios of internal variability, model uncertainty, and emissions uncertainty in climate change [69]. Testing the influence of these different types of uncertainty could enable the development of heuristics that identify systems where flexibility can be applied vs. systems where a robust approach is more favorable. This could be helpful for planners to identify and screen different climate adaptation scenarios without undertaking the full application of this modeling framework.

Social and institutional analysis There are a number of opportunities to extend the framework to address social and institutional uncertainties more comprehensively. First, in our current applications, we have modeled decisions from the perspective of a single planner, focusing on pragmatic alternatives the planner has the authority to implement. In reality, many different actors interact with water resource systems, and decisions made by the planner can influence the behavior of other actors and vice versa. Future work could explicitly model the behavior and decisions of multiple actors to include this twoway coupling of natural and human systems [147]. For small numbers of actors, such as multiple water utilities or planners in a region, game theoretic approaches for managing uncertainty could be used. Alternatively, we could model the impact of many individual actors through agent-based modeling, which has been applied to water resource systems to model the response of individual actors to changes in the water system at different scales [145]. This approach could be extended to address uncertainty by including variability in social responses using statistical or scenario-based approaches. For example, agent-based models could include uncertainty in the impact of demand-side restrictions or policies to reduce water use.

Second, institutional uncertainties could be addressed and included in the uncertainty framework. For example, project delays are a common and impactful uncertainty in infrastructure planning. Indeed, the Wonthaggi plant in Melbourne discussed in Chapter 3 faced lengthy construction delays. This could be included in the existing uncertainty framework by using historical data on construction times to develop probabilistic estimates of project completion time that could be included as statistical uncertainties. In the SDP model, this could be implemented by including probabilistic state transitions for planned infrastructure capacity, instead of assuming a fixed time for project construction. Other institutional uncertainties may require greater extension of the existing uncertainty framework. For example, the deep uncertainty analysis could be extended to identify and address institutional uncertainties that arise from ambiguity, or differences in stakeholder perspective, in a differentiated way. This could draw on recent literature integrating qualitative approaches into engineering systems analysis [154]. For example, qualitative scenario-discovery methods have been applied to identify deep uncertainties in multi-stakeholder decision processes [85]. Mixed-method approaches could be used to elicit the incentives, constraints, and uncertainties impacting the decisions of different stakeholders and then included in the optimization model; recent work has developed and applied this approach in electricity system planning [32].

Finally, future work could integrate stakeholder collaboration, an important element in crafting decision support for planners and policymakers that is directly usable [28]. A key motivation for the differentiated approach to diverse uncertainties developed in this dissertation is that in practice planners face many uncertainties in a single decision. Collaborative or participatory approaches could be used to identify these uncertainties and to classify them in a manner informed by stakeholders' assessment of the level of uncertainty and opportunities to learn in the future. Similarly, prior work in adaptive management has addressed the importance of social learning, in which exchanges across stakeholders and actors lead to greater understanding of the systems [163]. The approach developed here for assessing opportunities to learn about hydrological uncertainty in future could be extended to address social learning processes and its implications for planning decisions.

Other domains Finally, the methods and approaches developed in this dissertation could inform other fields of research as well. Many other engineering systems domains, such as electricity or transportation systems, address infrastructure planning decisions in the face of multiple, diverse uncertainties. The framework presented here could be broadly adapted to those systems. In particular, climate change uncertainty impacts many of these domains and the Bayesian approach developed in Chapter 5 could be applied to other infrastructure systems whose performance is impacted by climate to assess the value of flexible approaches to managing risk. The insights in this dissertation can also guide water technology development. For example, while much of research in desalination aims at improving efficiency and reducing cost of large scale technologies, we demonstrate that modular designs in which smaller volumes of capacity can be added even if the unit cost is somewhat higher can add value at a broader system-level scale. The framework developed here can be used to assess the impacts of new desalination or other water technology developments on water resource systems planning and performance.

Appendix A

Multiple uncertainties in Melbourne

Reservoir operations

For each generated simulation of future natural water supply, we calculate the total urban demand, demand for water from supply infrastructure systems (desalination plants and irrigation pipelines) and the unmet demand on an annual basis.

Total demand is modeled as a product of demand per capita and population. The base population in 2007 is 3.86 million. Each year, population grows at either a high or low growth rate of 150,000 or 50,000 respectively, based on historical data and forecasts [10]. The choice of the high or low growth rate is simulated in each 10-year planning period of the total 30-year model horizon. The probability of high growth is 0.5 in the first period. Condition probabilities are used to model mean reversion; if population growth was high in the first period, it has only 0.25 probability of being high again in the second period, and vice versa. This leads to 8 population growth outcomes, which range from 150,000 to 450,000 in growth over the 30-year planning period.

Once total demand is calculated, the demand from each infrastructure component and unmet demand is calculated using a water balance approach and simple operational rules, represented in Figure C-3 with parameter assumptions shown in Table A.1. The model uses five operational rules to track reservoir storage, water supplied and shortages:

- 1. Reservoir storage is calculated by adding the natural water inflow is added to the amount of water in reservoirs at the beginning of a given year.
- 2. Demand is calculated as the demand per capita times the population. Demand is then



Figure A-1: Schematic showing operational rules assumed in Melbourne analysis

subtracted from total reservoir storage, leaving either remaining demand or surplus water availability.

- 3. If the amount of water left is lower than the action threshold W_{action} , then water from the Sugarloaf pipeline is imported and added to reservoir storage in the infrastructure alternatives in which it is available (S2, S5, and S6). We calculate the amount of water available from the Sugarloaf pipeline as a linear function of the sum of amount of water available at the beginning of year $t(W_{t0})$ and the natural inflow in year $t(I_t)$. This reflects the assumption that the amount of water available from the Sugarloaf pipeline in a given year is correlated with general water availability in the Melbourne area in that year.
- 4. If reservoir storage is lower than both the action threshold W_{action} and the amount of water at the beginning of year $t(W_{t0})$, then water from the desalination plant is imported and added to reservoir storage, in the infrastructure alternatives in which it is available (S2 – S6). The latter condition reflects the assumption that the desalination plant is only used when the overall water availability is decreasing, and not when the reservoirs are recovering from a drought.
- 5. Finally, if the water left in the reservoirs after imports from available new infrastructure is lower than the critical threshold, $W_{critical}$. If this is the case, we leave reservoir storage at $W_{critical}$ and register the difference as shortage (unmet demand). This unmet demand would have to be met by sources other than natural inflow, the Sugarloaf pipeline, and the desalination plant, or through demand reductions. Note that the model does not include elastic demand reductions in response to shortages; this approach was taken to be able to simply and clearly measure and communicate the total volume of shortages.

Decision model

We use a decision analysis model to evaluate the six infrastructure alternatives and choose the best in each model run. Population growth uncertainty is incorporated directly into the decision analysis model as an uncertain event to demonstrate the impact of allowing a decision-maker to observe population growth over time and react. In each model run,

Parameter	Description	Value	Source/ comment
D^{cap}	Demand per capita	100 kL/p/y	
P_t	Population in year t		Based on population outcome in decision model
q_o	First parameter of regression of Sugarloaf pipeline water availability	-60	Estimate
q_1	Second parameter of regression of Sugarloaf pipeline water availability	0.089	Estimate
Irr_{min}^{av}	Water availability from pipeline during dry years	$20 \mathrm{MCM/y}$	Estimate
Irr_{max}^{av}	Maximum water availability from pipeline	$100 \mathrm{MCM/y}$	Estimate
S_{max}	Desalination plant capacity	$75 ext{ or } 150 ext{ MCM/y}$	Depends on size of plant (small or large) in year t
W_{max}	Maximum capacity of all reservoirs in Melbourne water supply system	1812 MCM	(Melbourne Water, 2015)
W_{action}	Action threshold (of storage level in Melbourne water supply system)	980 MCM	(Melbourne Water, 2015)
$W_{critical}$	Minimum required storage level in Melbourne water supply system	580 MCM	(Melbourne Water, 2015)

Table A.1: Parameter assumptions for water system moel

perfect information about water supply is assumed, and the choice of best infrastructure alternative is calculated using the standard backward recursion method of solving decision trees [129].

Figure A-2 shows an abbreviated schematic of the decision tree used in this case. In year 0, the decision-maker (DM) chooses from one of the six infrastructure alternatives (S1, S2,, S6). Population growth is either high or low over each 10-year planning period, as described in Section A2. Then, after observing whether population growth was high or low, the DM can then decide in year 10 whether to expand the small desalination plant capacity for the infrastructure alternatives that have a small plant with expansion option (S3 and S5). The population growth and planning process occurs again in year 20. We have incorporated a reversion to the mean in the population growth probabilities so that population outcomes tend to be close to the base case over the 30 year planning period; if population growth went up in the previous period, there is only a $\frac{1}{4}$ chance it will go up again in the second period and vice versa. Finally, the population again sees high or low growth until year 30. Each branch of the tree, consisting of a specific outcome of S (the choice of infrastructure alternative), P (the population growth outcome), and O (what expansion option choice was





made), incurs a specific payoff π . For infrastructure alternatives without a flexible expansion option, the DM does not make additional decisions after year 0, and different payoffs are incurred based on high or low population growth in each of the three 10-year periods. The best infrastructure alternative and accompanying expansion option decision (if relevant) are chosen to minimize the expected value of the payoff or total cost function.

Note that the implementation here takes into account path dependence, which is computationally tractable because of the simplified population model which assumes only two possible realizations of population growth in each period. In Chapters 4 and 5, the SDP implementation requires much more granularity in order to incorporate high fidelity physical models of hydrology and climate. Therefore, in those applications, an assumption of path independence is made.

Additional results

Additional results can be found in the Supplemental Information of the published paper, which can be found at: https://ascelibrary.org/doi/abs/10.1061/%28ASCE%29WR. 1943-5452.0000823

Appendix B

Groundwater uncertainty in Riyadh

Groundwater model

MODFLOW

MODFLOW is a numerical groundwater model developed in the 1980s by the USGS. In this paper, we use the version MODFLOW 2005 [65]. We assume the aquifer is confined and use a 2 dimensional version of the model. MODFLOW therefore solves the following version of the groundwater flow equation:

$$\frac{\partial}{\partial x} \left(K_x \delta \frac{\partial h}{\partial x} \right) + \frac{\partial}{\partial y} \left(K_y \delta \frac{\partial h}{\partial y} \right) + W(x, y, t) = S \frac{\partial h}{\partial t}$$
(B.1)

where K_x and K_y are hydraulic conductivity in the x and y directions respectively, W is the source term in this case reflecting pumping out of the aquifer and recharge into the aquifer, S is the storativity, δ is the aquifer thickness assumed constant, t is time, and h is the hydraulic head.

Artificial Neural Network

400 simulations of the MODFLOW model are used to train the artificial neural network (ANN). Latin hypercube sampling over the prior parameter distribution $p(\theta)$ is used to generate a unique combination of input parameters for each simulation; the set of these parameter combinations span the full possible range. The simulations are run with 100 time steps per year and recorded at a single grid cell representing the Buwayb well field. This yields a total of 4,000 data points for hydraulic head with varying inputs for K, S,



Figure B-1: Error histogram comparing drawdown estimates from MODFLOW to corresponding estimates from the ANN.

and t. This dataset is randomly split into a 70-15-15 train-validation-test partition. A feedforward ANN is trained using MATLAB's scaled conjugate gradient backpropagation algorithm. Many different network architectures varying the transfer function, number of hidden layers, and number of neurons are tested. We select the network with the lowest root mean square error (RMSE) on the test partition. This architecture has 2 hidden layers, with 6 neurons each, and a sigmoid transfer function. Because of the SDP formulation which imposes a drawdown-limit of 50 meters, only observations above the drawdown limit are included in calculating the RMSE; this allows us to choose the model that performs best in the range it will be used in the SDP. The RMSE calculated using this approach is 1.02 m, indicating excellent performance for our long-term regional planning application.

Cost assumptions

Equation 4.4 in Section 4.3 describes the formulation of the SDP for the groundwater application. Figure B-2 illustrates the cost assumptions used in the formulation. The marginal

cost of pumped groundwater P is the sum of two components: pumping costs and pretreatment costs because of the brackish quality of the water. Pumping costs were estimated using the cost of energy needed to raise water the height of drawdown in the well plus head losses due to friction estimated using the Darcy-Weisbach equation [77]. Pumping cost range between $0.40/m^3$ at the assumed starting depth of 337 m.a.s.l and $0.47/m^3$ when the 50 m depth limit is reached. Brackish treatment costs were assumed to vary between $0.3/m^3$ and $0.35/m^3$ for the starting depth and maximum depth respectively. The marginal cost of desalinated water is estimated as the sum of pumping costs through an existing pipeline from the desalination plant on the Arabian Gulf to Riyadh plus desalination opex. Pipeline pumping costs are assumed to be $1.35/m^3$ and were estimated as the cost of energy needed to raise water the elevation difference between the desalination plant at sea level and Riyadh at 612 m.a.s.l. plus head losses due to friction again estimated using the Darcy-Weisbach equation [77]; changes in elevation over the pipeline's path were not considered. RO desalination opex and capex were estimated to be $0.48/m^3$ and M227 based on a 108 MCM/y capacity plant using the Cost Estimator tool from Global Water Intelligence's "Desal Data" database [56]. This size was chosen to be equivalent to estimated withdrawals from the Minjur aquifer.



Figure B-2: Cost assumptions for groundwater SDP formulation. Pumping and brackish treatment costs are incurred for water supplied from groundwater; pipeline pumping and desalination opex are incurred for water supplied from desalination. Desalination capex is incurred when a new desalination plant is brought online.
Appendix C

Climate change uncertainty in Mombasa

Bayesian climate uncertainty analysis

An ensemble of 21 climate model projections are used. The models, as well as the ensemble number used, are listed below in Table C.1. More details on the statistical model used can be found in Smith et al. (2009) [148].

K-nn bootstrap for stochastic weather generation

The k-NN bootstrap approach from Rajagopalan and Lall (1999) [132] is implemented as follows:

- 1. Standardize the GCM's monthly temperature and precipitation values, based on the GCM-monthly climatology over each 20-year time period.
- 2. Calculate k-value. Because we implement the k-NN based on the climatology of the current month (total of 20 months in the 20 year window) and adjacent months (total of 40 months in the 20 years window), we have a total of 60 months being considered in the nearest neighbor. Following [132], we set k equal to the square root of 60, which we round down to 7.
- 3. For each value of 20-year mean T and P and time period in the SDP state space, a GCM is sampled.

Modeling Center	Institute ID	Model Name (ens. member)
Commonwealth Scientific and Industrial ResearchOrganization and Bureau of Meteorology, Australia	CSIRO/BOM	ACCESS 1.0 (1) ACCESS 1.3 (1)
Beijing Climate Center, China, Meteorological Administration	BCC	BCC-CSM1.1 (1)
EC-Earth Consortium	EC-EARTH	EC-EARTH (2, 8, 9, 12)
The First Institute of Oceanography, SOA, China	FIO	FIO-ESM $(2, 3)$
NOAA Geophysical Fluid Dynamics Laboratory	NOAA GFDL	GFDL-CM3 (1), GFDL-ESM2G (1), GFDL-ESM2M (1)
National Institute of Meteorological Research/Korea, Meteorological Administration	NIMR/KMA	HadGEM2-AO (1)
Met Office Hadley Centre	MOHC	HadGEM2-CC (1)
Japan Agency for Marine-Earth Science and Technology, Atmosphere and Ocean Research Institute (The University of Tokyo), and National Institute for Environmental Studies	MIROC	MIROC-ESM-CHEM (1) MIROC-ESM (1)
Atmosphere and Ocean Research Institute (The University of Tokyo), National Institute for Environmental Studies, and Japan Agency for Marine -Earth Science and Technology	MIROC	MIROC5 (1, 2, 3)
Norwegian Climate Centre	NCC	NorESM1-M (1), NorESM1-ME (1)

Table C.1: Climate model ensembles used

- 4. Initialize starting month by randomly selecting a year in sample time window and setting initial current month to January in that year.
- 5. Calculate the Euclidean distance for all of the current months' and adjacent months' standardized precipitation and temperature values (total of 60 Euclidean distances).
- 6. Rank the distances from smallest to largest and select the first k ranks $(R_1, R_2, ..., R_k)$.
- 7. Specify the sampling density, D_i for the first k ranked months as follows: $D_i = \frac{1/R_i}{\sum_{j=1}^k R_j}$
- 8. Sample the month from the distribution, D_i and specify the next month in the time series to be the month following the one sampled from D_i .
- 9. Repeat steps 5-8 using a unique sampling density D_i until the 20-year monthly time series is complete.
- 10. Transform the standardized monthly time-series back to monthly values by multiplying by the monthly standard deviation, adding the GCM's monthly cycle back in and then adding the annual mean values of T and P. If negative precipitation values exist, set them to zero.

CLIRUN II Rainfall-Runoff Model

CLIRUN II is a lumped watershed rainfall-runoff model designed to assess the impacts of climate change on runoff. It builds on the single layer model CLIRUN [83] by taking a two-layer approach to modeling soil moisture: the upper soil layer and lower groundwater layer allow for both fast and slow runoff response to precipitation [151]. Figure C-1 depicts the model structure and Figure C-2 shows the performance of our model calibration on the Mwache River. More details about the model can be found in Strzepek et al. (2011) [151]

Reservoir Operation Model

The reservoir operation model uses a water mass balance approach, shown in Equation C.1.

$$\Delta S_t = R_t + P_t + D_t - Y_t - E_t - O_t \tag{C.1}$$



Figure C-1: Schematic depicting structure of CLIRUN II. Model take mass balance of water approach, with six calibrated parameters that determine the flow rates. Reproduced with permission from Strzepek et al. (2011) [151]



CLIRUN II Calibration Fit

Figure C-2: Calibration fit for CLIRUN II rainfall-runoff model. Top panel compares modeled vs. observed values for monthly runoff from 1976 to 1990. Bottom panel compares modeled vs. observed values for monthly average streamflow across the 14-year time period.



Figure C-3: Reservoir operations model applied to a sample 5-year time series of runoff for a 120 MCM reservoir. Top panel shows reservoir inflows and storage volume. Bottom panel shows yield and unmet demand. In this sample time series, reliability outages occur at the end of year 1 due to several consecutive months of inflows lower than demand.

where S is reservoir storage, R is runoff, P is direct precipitation into the reservoir, D is imported desalination water, Y is reservoir yield delivered to serve demand, E is evaporation from the reservoir estimated using modified Hargreaves [7], O is downstream outflows, and t is time measured in monthly steps. Operating rules are applied such that yields are equal to the full domestic and agricultural demand, or as much as can be delivered. If desalination is available, it is used to refill reservoir storage when storage levels drop below 50% of capacity. Reservoir levels can not drop below the dead storage volume of 20 MCM. Outflows are released to prevent storage going above the reservoir capacity, and we assume no environmental outflow requirements. Figure C-3 demonstrates the reservoir operations for a 120 MCM dam without desalination for a sample 5-year time series of runoff.

Additional Results



Figure C-4: Confidence intervals for learning over time: alternate simulation in which precipitation and MAR decline, driving shortage risk higher.

Bibliography

- [1] ABC News. Victoria's desalination plant finally delivers as Government places order for more water. http://www.abc.net.au/news/2017-03-19/ victoria-desalination-plant-finally-delivers-water/8367554, March 19, 2017.
- [2] ACIL Tasman. Wholesale energy cost forecast for serving residential users. Technical report, Australian Energy Market Commission, 2011.
- [3] W.N. Adger, S. Agrawala, M.M.Q. Mirza, C. Conde, K. O'Brien, J. Pulhin, R. Pulwarty, B. Smit, and K. Takahashi. Assessment of adaptation practices, options, constraints and capacity. Climate Change 2007: Impacts, Adaptation and Vulnerability. Contribution of Working Group II to the Fourth Assessment Report of the Intergovernmental Panel on Climate Change. Cambridge University Press, Cambridge, 2007.
- [4] Amir Aghakouchak, David Feldman, Michael Stewardson, Jean-Daniel Saphores, Stanley Grant, and Brett Sanders. Australia's Drought : Lessons for California. *Science*, 343(March):1430–1431, 2014.
- [5] Mohammed Abdullah Al-Saleh. Declining groundwater level of the Minjur Aquifer, Tebrak area, Saudi Arabia. The Geographical Journal, 158(2):215–222, 1992.
- [6] Mohammad Bin Ibrahim Al-Saud. National Water Strategy. Technical report, Ministry of Water and Electricity, Riyadh, Saudi Arabia, 2013.
- [7] Richard G Allen, Luis S Pereira, Dirk Raes, Martin Smith, and W Ab. Crop evapotranspiration-Guidelines for computing crop water requirements-FAO Irrigation and drainage paper 56. *Irrigation and Drainage*, 300(56):1–15, 1998.
- [8] Mary P. Anderson, William W. Woessner, and Randall J. Hunt. Applied Groundawter Modeling. Academic Press, London, 2002.
- [9] M. J. Asher, B. F. W. Croke, A. J. Jakeman, and L. J. M. Peeters. A review of surrogate models and their application to groundwater modeling. *Water Resources Research*, 51(8):5957–5973, 8 2015.
- [10] Australian Bureau of Statistics. Regional Population Growth. http://www.abs. gov.au/ausstats/abs@.nsf/products/AC53A071B4B231A6CA257CAE000ECCE5? OpenDocument#PARALINK1, 2014.
- [11] Jonathan Baker, Paul Block, Kenneth Strzepek, and Richard de Neufville. Power of Screening Models for Developing Flexible Design Strategies in Hydropower Projects:

Case Study of Ethiopia. Journal of Water Resources Planning and Management, 140(12):04014038, 2014.

- [12] Eva H. Y Beh, Holger R. Maier, and Graeme C. Dandy. Adaptive, multiobjective optimal sequencing approach for urban water supply augmentation under deep uncertainty. *Water Resources Research*, 51:1529–1551, 2015.
- [13] Sarah Jayne Bell. The Place That Will Save Us When There's A Drought: London's Desalination Plant. Londonist, December 7, 2017.
- [14] Yakov Ben-Haim. Info-Gap Decision Theory: Decisions Under Severe Uncertainty. Elsevier, Oxford, 2001.
- [15] Keith Beven and Andrew Binley. The Future Of Distributed Models Model Calibration And Uncertainty Prediction. *Hydrological Processes*, 6(3):279–298, 1992.
- [16] Anik Bhaduri, Janos Bogardi, Afreen Siddiqi, Holm Voigt, Charles Vörösmarty, Claudia Pahl-Wostl, Stuart E. Bunn, Paul Shrivastava, Richard Lawford, Stephen Foster, Hartwig Kremer, Fabrice G. Renaud, Antje Bruns, and Vanesa R. Osuna. Achieving Sustainable Development Goals from a Water Perspective. *Frontiers in Environmental Science*, 4:64, 10 2016.
- [17] Paul Block and Balaji Rajagopalan. Interannual Variability and Ensemble Forecast of Upper Blue Nile Basin Kiremt Season Precipitation. *Journal of Hydrometeorology*, 8(3):327–343, 2007.
- [18] Brent Boehlert, Susan Solomon, and Kenneth M. Strzepek. Water under a changing and uncertain climate: Lessons from climate model ensembles. *Journal of Climate*, 28(24):9561–9582, 2015.
- [19] Casey Brown, Yonas Ghile, Mikaela Laverty, and Ke Li. Decision scaling: Linking bottom-up vulnerability analysis with climate projections in the water sector. Water Resources Research, 48(9):1–12, 2012.
- [20] Marcela Brugnach, Art Dewulf, Claudia Pahl-Wostl, and Tharsi Taillieu. Toward a relational concept of uncertainty: About knowing too little, knowing too differently, and accepting not to know. *Ecology and Society*, 13(2), 2008.
- [21] Wesley Burrows and John Doherty. Efficient Calibration/Uncertainty Analysis Using Paired Complex/Surrogate Models. Groundwater, 53(4):531–541, 2015.
- [22] Andrew Cala. Spain's Desalination Ambitions Unravel. The New York Times, October 9, 2013.
- [23] Michel-Alexandre Cardin, Mehdi Ranjbar-Bourani, and Richard de Neufville. Improving the Lifecycle Performance of Engineering Projects with Flexible Strategies: Example of On-Shore LNG Production Design. Systems Engineering, 18(3):253–268, 2015.
- [24] Central Intelligence Agency. The World Factbook: Kenya. https://www.cia.gov/ library/publications/the-world-factbook/geos/ke.html.

- [25] CES Consultants. Feasibility Study, Preliminary and Detailed Engineering Designs of Development of Mwache Multi-Purpose Dam Project along Mwache River: Hydrology Report. Technical report, Ministry of Regional Development, Nairobi, Kenya, 2013.
- [26] Robin Chadwick, Ian Boutle, and Gill Martin. Spatial patterns of precipitation change in CMIP5: Why the rich do not get richer in the tropics. *Journal of Climate*, 26(11):3803–3822, 2013.
- [27] Daniella Cheslow. Cape Town's 'Day Zero' has been postponed, but the city is still scrambling to deal with its water crisis. *PRI's The World*, March 28, 2018.
- [28] William C. Clark, Lorrae van Kerkhoff, Louis Lebel, and Gilberto C. Gallopin. Crafting usable knowledge for sustainable development. Proceedings of the National Academy of Sciences, 113(17):4570–4578, 2016.
- [29] Teresa B Culver and Christine A Shoemaker. Dynamic Optimal Control for Groundwater Remediation With Flexible Management Periods. Water Resources Research, 28(3):629–641, 1992.
- [30] Andrew Dannatt and Amelia Paszkowksi. Wadi Hanifah Water Balance Review. Technical Report February, Burohappold Engineering and Arrivadh Development Authority (ADA), 2017.
- [31] Alyssa M. Dausman, John Doherty, Christian D. Langevin, and Michael C. Sukop. Quantifying data worth toward reducing predictive uncertainty. *Ground Water*, 48(5):729–740, 2010.
- [32] Michael Davidson. Creating Markets for Wind Electricity in China: Case Studies in Energy Policy and Regulation. PhD thesis, Massachusetts Institute of Technology, 2018.
- [33] Richard de Neufville and Stefan Scholtes. *Flexibility in Engineering Design*. MIT Press, Cambridge, Massachusetts, 2011.
- [34] Richard de Neufville and Kim Smet. Engineering Options Analysis (EOA): Theory. In *Decision Making under Deep Uncertainty: From Theory to Practice (In Press)*, chapter 7. Springer.
- [35] Olivier L. de Weck, Daniel Roos, and Christopher L. Magee. Engineering Systems: Meeting Human Needs in a Complex Technological World. MIT Press, Cambridge, MA, 2011.
- [36] H Delottier, A Pryet, and A Dupuy. Why Should Practitioners be Concerned about Predictive Uncertainty of Groundwater Management Models? Water Resources Management, 31:61–73, 2017.
- [37] Yinghan Deng, Michel-Alexandre Cardin, Vladan Babovic, Deepak Santhanakrishnan, Petra Schmitter, and Ali Meshgi. Valuing flexibilities in the design of urban water management systems. *Water research*, 47(20):7162–74, 2013.
- [38] Department for Environment and Water, Government of South Australia. Millennium drought. https://www.environment.sa.gov.au/topics/river-murray/ about-the-river/millennium-drought, 2018.

- [39] Suraje Dessai and Mike Hulme. Assessing the robustness of adaptation decisions to climate change uncertainties: A case study on water resources management in the East of England. *Global Environmental Change*, 17(1):59–72, 2007.
- [40] Michael D Dettinger and John L Wilson. First Order Analysis of Uncertainty in Numerical Model of Groundwater Flow. Water Resources Research, 17(1):149–161, 1981.
- [41] Deutsche Gesellschaft fur Internationale Zusammenarbeit (GIZ) and Dornier Consulting International (DCI). Detailed Water Resources Studeies of Khuff Jilh Minjur Dhruma and Overlying Aquifers. Technical report, Ministry of Enivronment Water & Agriculture, 2016.
- [42] Julian Doczi. Managing Climate Risk for the Water Sector with Tools and Decision Support. In Sangam Shrestha, Mukand S. Babel, and Vishnu Prasad Pandey, editors, *Climate Change and Water Resources*, pages 239–290. CRC Press, 2014.
- [43] Rebecca S. Dodder, Joshua B. McConnell, Ali Mostashari, and Joseph M. Sussman. The Concept of the 'CLIOS Process': Integrating the Study of Physical and Policy Systems Using Mexico City as an Example. Massachusetts Institute of Technology Engineering Systems Symposium, pages 1–48, 2004.
- [44] John Doherty. Calibration and Uncertainty Analysis for Complex Environmental Models. Watermark Numerical Computing, Brisbane, Australia, 2015.
- [45] David Donoho and Miriam Gasko. Breakdown Properties of Location Estimates Based on Halfspace Depth and Projected Outlyingness. *The Annals of Statistics*, 20(4):1803– 1827, 1992.
- [46] D. Erdal and O. A. Cirpka. Joint inference of groundwater-recharge and hydraulicconductivity fields from head data using the Ensemble-Kalman filter. *Hydrology and Earth System Sciences Discussions*, 12(6):5565–5599, 2015.
- [47] European Comission. Adaptating infrastructure to climate change. Technical report, Communication from the Commission to the European Parliament, the Council, The European Economic and Social Committee and the Committee of the Regions, 2013.
- [48] Charles Fant, C Adam Schlosser, Xiang Gao, Kenneth Strzepek, and John Reilly. Projections of Water Stress Based on an Ensemble of Socioeconomic Growth and Climate Change Scenarios: A Case Study in Asia. *PloS one*, 11(3):e0150633, 2016.
- [49] Briony C. Ferguson, Rebekah R. Brown, Niki Frantzeskaki, Fjalar J. de Haan, and Ana Deletic. The enabling institutional context for integrated water management: Lessons from Melbourne. Water Research, 47(20):7300–7314, 2013.
- [50] Luc Feyen and Steven M. Gorelick. Framework to evaluate the worth of hydraulic conductivity data for optimal groundwater resources management in ecologically sensitive areas. Water Resources Research, 41(3):1–13, 2005.
- [51] Sarah M. Fletcher, Marco Miotti, Jaichander Swaminathan, Magdalena M. Klemun, Kenneth Strzepek, and Afreen Siddiqi. Water Supply Infrastructure Planning: A Decision-Making Framework to Classify Multiple Uncertainties and Evaluate Flexible

Design. Journal of Water Resources Planning and Management, 143(10):04017061, 2017.

- [52] Lionel Frost, Andrea Gaynor, Jenny Gregory, Ruth Morgan, Seamus O'Hanlon, Peter Spearritt, and John Young. *Water, history and the Australian city.* 2016.
- [53] Berry Gersonius, Richard Ashley, Assela Pathirana, and Chris Zevenbergen. Climate change uncertainty: Building flexibility into water and flood risk infrastructure. *Climatic Change*, 116(2):411–423, 2013.
- [54] Justin Gillis. For Drinking Water in Drought, California Looks Warily to Sea. New York Times, April 11, 2015.
- [55] Filippo Giorgi and Linda O. Mearns. Calculation of average, uncertainty range, and reliability of regional climate changes from AOGCM simulations via the "Reliability Ensemble Averaging" (REA) method. Journal of Climate, 15(10):1141–1158, 2002.
- [56] Global Water Intelligence. Desal Data Cost Estimator. https://www.desaldata. com/cost_estimator.
- [57] Josh Gordon. Desal water will be needed every year, says Victorian minister Lisa Neville. The Age, March 19, 2017.
- [58] Stanley B. Grant, Tim D. Fletcher, David Feldman, Jean Daniel Saphores, Perran L M Cook, Mike Stewardson, Kathleen Low, Kristal Burry, and Andrew J. Hamilton. Adapting urban water systems to a changing climate: Lessons from the millennium drought in southeast Australia. *Environmental Science and Technology*, 47(19):10727– 10734, 2013.
- [59] Robin Gregory. Using Stakeholder Values to Make Smarter Environmental Decisions. Environment: Science and Policy for Sustainable Development, 42(5):34–44, 2000.
- [60] David G. Groves and Robert J. Lempert. A new analytic method for finding policyrelevant scenarios. *Global Environmental Change*, 17(1):73–85, 2007.
- [61] Marjolijn Haasnoot, Jan H. Kwakkel, Warren E. Walker, and Judith ter Maat. Dynamic adaptive policy pathways: A method for crafting robust decisions for a deeply uncertain world. *Global Environmental Change*, 23(2):485–498, 2013.
- [62] Marjolijn Haasnoot, Hans Middelkoop, Astrid Offermans, Eelco van Beek, and Willem P A van Deursen. Exploring pathways for sustainable water management in river deltas in a changing environment. *Climatic Change*, 115(3-4):795–819, 2012.
- [63] Rimo P. Hämäläinen, Eero Kettunen, and Harri Ehtamo. Evaluating a Framework for Multi-Stakeholder Decision Support in Water Resources Management. Group Decision and Negotiation, 10:331–353, 2001.
- [64] Matt Hamilton. Santa Barbara to spend \$55 million on desalination plant as drought 'last resort'. Los Angeles Times, July 22, 2015.
- [65] Arlen W. Harbaugh. MODFLOW-2005, The U.S. Geological Survey Modular Ground-Water Model — the Ground-Water Flow Process. 2005.

- [66] Mark Harrison. Valuing the Future: The social discount rate in cost-benefit analysis. Technical Report April, Australian Government Productivity Commission, 2010.
- [67] Tsuyoshi Hashimoto, Jery R. Stedinger, and Daniel P. Loucks. Reliability, Resiliency, and Vulnerability Criteria for Water Resource System Performance Evaluation. Water Resources Research, 18(1):14–20, 1982.
- [68] Ahmed E Hassan, Hesham M Bekhit, and Jenny B Chapman. Using Markov Chain Monte Carlo to quantify parameter uncertainty and its effect on predictions of a groundwater flow model. *Environmental Modelling & Software*, 2008.
- [69] Ed Hawkins and Rowan Sutton. The Potential to Narrow Uncertainty in Regional Climate Predictions. Bulletin of the American Meteorological Society, pages 1095– 1107, August 2009.
- [70] Ed Hawkins and Rowan Sutton. The potential to narrow uncertainty in projections of regional precipitation change. *Climate Dynamics*, 37(1):407–418, 2011.
- [71] M. Heidari. Application of Linear System Theory and Linear Programming to Groundwater Management in Kansas. Water Resources Bulletin AWRA, 18(6):1003–1012, 1982.
- [72] I M Held and B J Soden. Robust responses of the hydrologic cycle to global warming. Journal of Climate, 19:5686–5699, 2006.
- [73] J Herman, P Reed, H Zeff, and G Characklis. How Should Robustness Be Defined for Water Systems Planning under Change? Journal of Water Resources Planning and Management, 141(10):4015012, 2015.
- [74] JA Jennifer A. Hoeting, David Madigan, Adrian E. AE Raftery, and Chris T. Volinsky. Bayesian model averaging: A tutorial. *Statistical science*, 14(4):382–401, 1999.
- [75] Mashor Housh, Avi Ostfeld, and Uri Shamir. Limited multi-stage stochastic programming for managing water supply systems. *Environmental Modelling and Software*, 41:53–64, 2013.
- [76] G.H. Huang and D. P. Loucks. An Inexact Two-Stage Stochastic Programming Model for Water Resources Management under Uncertainty. *Civil Engineering and Environmental Systems*, 17(2):95–118, 5 2000.
- [77] Ned H. C. Hwang and Robert J. Houghtalen. Fundamentals of Hydraulic Engineering Systems. Upper Saddle River, New Jersey, 1996.
- [78] IPCC. Climate change 2013: the physical science basis. Contribution of working group I to the fifth assessment report of the intergovernmental panel on climate change. 2013.
- [79] Marion W. Jenkins, Jay R. Lund, and Richard E. Howitt. Using economic loss functions to value urban water scarcity in California. *Journal of the American Water Works Association*, 95(2):58–70, 2003.
- [80] M Jeuland and D Whittington. Water resources planning under climate change: Assessing the robustness of real options for the Blue Nile. Water Resources Research, pages 2086–2107, 2014.

- [81] Ling Ji, Ping Sun, Qiang Ma, Na Jiang, Guo He Huang, and Yu Lei Xie. Inexact two-stage stochastic programming for water resources allocation under considering demand uncertainties and response-A case study of Tianjin, China. Water (Switzerland), 9(6):1–14, 2017.
- [82] Juliette Jowit. Thames Water opens first large-scale desalination plant in UK. The Guardian, June 2, 2010.
- [83] Z. Kaczmarek. Water balance model for climate impact analysis. Acta Geophysica Polonica, 41(4):423–437, 1993.
- [84] D Kang and K Lansey. Multiperiod planning of water supply infrastructure based on scenario analysis. Journal of Water Resources Planning and Management, 140(1):40– 54, 2014.
- [85] Christopher W. Karvetski and James H. Lambert. Evaluating Deep Uncertainties in Strategic Priority-Setting with an Application to Facility Energy Investments. Systems Engineering, 15(2):483–9, 2012.
- [86] J. R. Kasprzyk, P. M. Reed, B. R. Kirsch, and G. W. Characklis. Managing population and drought risks using many-objective water portfolio planning under uncertainty. *Water Resources Research*, 45(12):1–18, 2009.
- [87] Joseph R. Kasprzyk, Patrick M. Reed, Gregory W. Characklis, and Brian R. Kirsch. Many-objective de Novo water supply portfolio planning under deep uncertainty. *Environmental Modelling and Software*, 34:87–104, 2012.
- [88] Frank H. Knight. *Risk, Uncertainty, and Profit.* Library of Economics and Liberty, 1921.
- [89] Jan H. Kwakkel, Marjolijn Haasnoot, and Warren E. Walker. Developing dynamic adaptive policy pathways: a computer-assisted approach for developing adaptive strategies for a deeply uncertain world. *Climatic Change*, 132(3):373–386, 2015.
- [90] Jan H. Kwakkel, Warren E. Walker, and Vincent A.W.J. Marchau. Classifying and communicating uncertainties in model-based policy analysis. *International Journal of Technology, Policy and Management*, 10(4):299, 2010.
- [91] Upmanu Lall and Ashih Sharma. A nearest neighbor bootstrap for resampling hydrologic time series. Water Resources Research, 32(3):679–693, 1996.
- [92] Robert J. Lempert and David G. Groves. Identifying and evaluating robust adaptive policy responses to climate change for water management agencies in the American west. *Technological Forecasting and Social Change*, 77(6):960–974, 7 2010.
- [93] Robert J. Lempert, David G. Groves, Steven W. Popper, and Steve C. Bankes. A General, Analytic Method for Generating Robust Strategies and Narrative Scenarios. *Management Science*, 52(4):514–528, 4 2006.
- [94] Judit Lienert, Jochen Monstadt, and Bernhard Truffer. Future scenarios for a sustainable water sector: a case study from Switzerland. *Environmental science & technology*, 40(2):436–42, 2006.

- [95] H.F. Lins and E.Z. Stakhiv. Managing the Nation's water in a changing climate. Journal of the American Water Resources Association, 34(6):1255–1264, 1998.
- [96] X. M. Liu, G. H. Huang, S. Wang, and Y. R. Fan. Water resources management under uncertainty: factorial multi-stage stochastic program with chance constraints. *Stochastic Environmental Research and Risk Assessment*, 30(3):945–957, 2016.
- [97] Kathleen G. Low, Stanley B. Grant, Andrew J. Hamilton, Kein Gan, Jean-Daniel Saphores, Meenakshi Arora, and David L. Feldman. Fighting drought with innovation: Melbourne's response to the Millennium Drought in Southeast Australia. Wiley Interdisciplinary Reviews: Water, pages n/a-n/a, 2015.
- [98] Zhiming Lu, David Higdon, Dongxiao Zhang, and Dongxiao Zhang. A Markov chain Monte Carlo method for the groundwater inverse problem. In *Developments in water science*, volume 55, pages 1273–1283. 2004.
- [99] R. Duncan Luce and Howard Raiffa. Games and Decisions:. Wiley, New York, 1957.
- [100] Ben Marchant, Jonathan Mackay, and John Bloomfield. Quantifying uncertainty in predictions of groundwater levels using formal likelihood methods. 2016.
- [101] João Marques, Maria Cunha, and Dragan Savić. Using Real Options in the Optimal Design of Water Distribution Networks. Journal of Water Resources Planning and Management, (1992):04014052, 2014.
- [102] Lawrence E. McCray, Kenneth A. Oye, and Arthur C. Petersen. Planned adaptation in risk regulation: An initial survey of US environmental, health, and safety regulation. *Technological Forecasting and Social Change*, 77(6):951–959, 2010.
- [103] David McInerney, Robert Lempert, and Klaus Keller. What are robust strategies in the face of uncertain climate threshold responses?: Robust climate strategies. *Climatic Change*, 112(3-4):547–568, 2012.
- [104] Melbourne Water. Customers and Prices. http://www.melbournewater.com.au/ aboutus/customersandprices/Pages/Bulk-water.aspx, 2015.
- [105] Melbourne Water. Water Data. http://www.melbournewater.com.au/waterdata/ waterstorages/Pages/Inflow-over-the-years.aspx, 2015.
- [106] Melbourne Water. Water Outlook for Melbourne. Technical report, Melbourne Water, 2015.
- [107] Chiyuan Miao, Qingyun Duan, Qiaohong Sun, Yong Huang, Dongxian Kong, Tiantian Yang, Aizhong Ye, Zhenhua Di, and Wei Gong. Assessment of CMIP5 climate models and projected temperature changes over Northern Eurasia. *Environmental Research Letters*, 9(5), 2014.
- [108] P C D Milly, Julio Betancourt, Malin Falkenmark, Robert M Hirsch, Zbigniew W Kundzewicz, Dennis P Lettenmaier, and Ronald J Stouffer. Climate change. Stationarity is dead: whither water management? *Science (New York, N.Y.)*, 319(5863):573– 574, 2008.

- [109] Ministry of Agriculture and Water. Water Atlas of Saudi Arabia. Technical report, Ministry of Agriculture and Water in cooperation with the Saudi Arabian-United States Joint Commission on Economic Cooperation, Riyadh, Saudi Arabia, 1984.
- [110] Paul Moody and Casey Brown. Modeling stakeholder-defined climate risk on the Upper Great Lakes. Water Resources Research, 48(10):1–15, 2012.
- [111] Catherine Moore and John Doherty. Role of the calibration process in reducing model predictive error. Water Resources Research, 41(5):1–14, 2005.
- [112] M. Granger Morgan and Max Henrion. Uncertainty: A guide to dealing with uncertainty in quantitative risk and policy analysis. Cambridge University Press, New York, 1990.
- [113] Sudeep Nair, Biju George, Hector M. Malano, Meenakshi Arora, and Bandara Nawarathna. WaterâĂŞenergyâĂŞgreenhouse gas nexus of urban water systems: Review of concepts, state-of-art and methods. *Resources, Conservation and Recycling*, 89:1–10, 8 2014.
- [114] National Land Commission, Government of Kenya. Mwache Dam construction in Kwale County. http://www.landcommission.go.ke/article/ mwache-dam-construction-in-kwale-county, 20174.
- [115] National Research Council. Adaptive Management for Water Resources Project Planning. The National Academies Press, Washington, DC, 2004.
- [116] Grey S. Nearing, Yudong Tian, Hoshin V. Gupta, Martyn P. Clark, Kenneth W. Harrison, and Steven V. Weijs. A philosophical basis for hydrological uncertainty. *Hydrological Sciences Journal*, 61(9):1666–1678, 7 2016.
- [117] Shlomo P. Neuman, Liang Xue, Ming Ye, and Dan Lu. Bayesian analysis of dataworth considering model and parameter uncertainties. Advances in Water Resources, 36:75–85, 2012.
- [118] M. New, M. Hulme, and P. Jones. Representing twentieth-century space-time climate variability. Part II: Development of 1901-96 monthly grids of terrestrial surface climate. *Journal of Climate*, 13(13):2217–2238, 2000.
- [119] Robert O. Ojwang, Jörg Dietrich, Prajna Kasargodu Anebagilu, Matthias Beyer, and Franz Rottensteiner. Rooftop rainwater harvesting for Mombasa: Scenario development with image classification and water resources simulation. Water (Switzerland), 9(5), 2017.
- [120] D. O'Malley and V. V. Vesselinov. Groundwater remediation using the information gap decision theory. *Water Resources Research*, 50(1):246–256, 2014.
- [121] Claudia Pahl-Wostl. Transitions towards adaptive management of water facing climate and global change. Integrated Assessment of Water Resources and Global Change: A North-South Analysis, pages 49–62, 2007.
- [122] S Pallottino, G Sechi, and P Zuddas. A DSS for water resources management under uncertainty by scenario analysis. *Environmental Modelling & Software*, 20(8):1031– 1042, 2005.

- [123] Reed N. Palmer and Gregory W. Characklis. Reducing the costs of meeting regional water demand through risk-based transfer agreements. *Journal of Environmental Man*agement, 90(5):1703–1714, 2009.
- [124] Maria P. Papadopoulou, George F. Pinder, and George P. Karatzas. Flexible timevarying optimization methodology for the solution of groundwater management problems. *European Journal of Operational Research*, 180(2):770–785, 2007.
- [125] R. C. Peralta and I. M. Kalwji. Groundwater optimization handbook. CRC Press, Boca Raton, FL, 2012.
- [126] Michael G. Porter. A Tale of Two Cities : Desalination and Drought in Perth and Melbourne A report prepared for NCEDA under : 'Desalination for Australian Economic Development'. pages 1–43, 2013.
- [127] Warren Powell. Approximate dynamic programming: Solving the curses of dimensionality. Wiley, Hoboken, New Jersey, 2010.
- [128] Martin L. Puterman. Markov Decision Processes: Discrete Stochastic Dynamic Programming. John Wiley & Sons, Hoboken, New Jersey, 1994.
- [129] Howard Raiffa. Risk Sharing and Group Decisions. In Decision Analysis: Introductory Lectures on Choices under Uncertainty, chapter 8, pages 188–238. Addison-Wesley, Reading, Massachusetts, 1968.
- [130] J. Räisänen and T. N. Palmer. A probability and decision-model analysis of a multimodel ensemble of climate change simulations. *Journal of Climate*, 14(15):3212–3226, 2001.
- [131] Jouni Räisänen. How reliable are climate models? Tellus, Series A: Dynamic Meteorology and Oceanography, 59(1):2–29, 2007.
- [132] B. Rajagopalan and U. Lall. A k-nearest-neighbor Simulator for Daily Precipitation and Other Variables. Water Resources Research, 35(10):3089âĂŞ3101, 1999.
- [133] Carmen G. Rawls and Mark A. Turnquist. Pre-positioning and dynamic delivery planning for short-term response following a natural disaster. *Socio-Economic Planning Sciences*, 46(1):46–54, 2012.
- [134] P A Ray, P H Kirshen, and D W Watkins Jr. Staged Climate Change Adaptation Planning for Water Supply in Amman, Jordan. Journal of Water Resources Planning and Management, 138(5):403–411, 2012.
- [135] Patrick A. Ray and Casey M. Brown. Confronting Climate Uncertainty in Water Resources Planning and Project Design: The Decision Tree Framework. Technical report, 2015.
- [136] P. M. Reed, D. Hadka, J. D. Herman, J. R. Kasprzyk, and J. B. Kollat. Evolutionary multiobjective optimization in water resources: The past, present, and future. *Advances in Water Resources*, 51:438–456, 2013.
- [137] Ortwin Renn, Thomas Webler, Horst Rakel, Peter Dienel, and Branden Johnson. Public participation in decision making : A three-step procedure. *Policy Sciences*, 26:189–214, 1993.

- [138] W. Richardson, C. Stochastic simulation of daily precipitation, temperature and solar radiation. Water resources research, 17(1):182–190, 1981.
- [139] D.M. Rizzo and D.E. Dougherty. Design optimization for multiple management period groundwater remediation injection and extraction. Water Resources Research, 32(8):2549–2561, 1996.
- [140] Rodrigo Rojas, Samalie Kahunde, Luk Peeters, Okke Batelaan, Luc Feyen, and Alain Dassargues. Application of a multimodel approach to account for conceptual model and scenario uncertainties in groundwater modelling. *Journal of Hydrology*, 394(3-4):416-435, 2010.
- [141] Peter J. Rousseeuw, Ida Ruts, and John W. Tukey. The Bagplot: A Bivariate Boxplot. The American Statistician, 53(4):382–387, 1999.
- [142] Mukta Sapkota, Meenakshi Arora, Hector Malano, Magnus Moglia, Ashok Sharma, Biju George, and Francis Pamminger. An Integrated Framework for Assessment of Hybrid Water Supply Systems. *Water*, 8(1):4, 2015.
- [143] Saudi Arabian Gerenal Investment Authority. Riyadh Region Economic Report. Technical report, Saudi Arabian Gerenal Investment Authority, Riyadh, Saudi Arabia, 2014.
- [144] CA Schlosser, Kenneth Strzepek, and Xiang Gao. The Future of Global Water Stress: An Integrated Assessment. *Earth's Future*, 2(8):341–361, 2014.
- [145] Maja Schlüter and Claudia Pahl-Wostl. Mechanisms of Resilience in Common-pool Resource Management Systems: an Agent-based Model of Water Use in a River Basin. *Ecology and Society*, 12(2):23, 2007.
- [146] Anneli Schöniger, Walter A Illman, Thomas Wöhling, and Wolfgang Nowak. Finding the right balance between groundwater model complexity and experimental effort via Bayesian model selection. *Journal of Hydrology*, 531:96–110, 2015.
- [147] M Sivapalan, M Konar, V Srinivasan, A Chhatre, A Wutich, C A Scott, and J L Wescoat. Socio-hydrology : Use-inspired water sustainability science for the Anthropocene. *Earth's Future*, 2:225–230, 2014.
- [148] Richard L. Smith, Claudia Tebaldi, Doug Nychka, and Linda O. Mearns. Bayesian Modeling of Uncertainty in Ensembles of Climate Models. *Journal of the American Statistical Association*, 104(485):97–116, 2009.
- [149] Eugene Z. Stakhiv. Pragmatic approaches for water management under climate change uncertainty. Journal of the American Water Resources Association, 47(6):1183–1196, 2011.
- [150] Kenneth Strzepek, Michael Jacobsen, Brent Boehlert, and James Neumann. Toward evaluating the effect of climate change on investments in the water resources sector: insights from the forecast and analysis of hydrological indicators in developing countries. *Environmental Research Letters*, 8(4):044014, 12 2013.

- [151] Kenneth Strzepek, Alyssa McCluskey, Brent Boehlert, Michael Jacobsen, and Charles Fant IV. Climate Variability and Change : A basin scale indicator approach to understanding the risk of climate variability and change: to water resources development and management. Technical report, Word Bank, 2011.
- [152] Kenneth M. Strzepek and Alyssa L. Mccluskey. Modeling the Impact of Climate Change on Global Hydrology and Water Availability. Technical report, The World Bank, September 2010.
- [153] Adam Richard Swanson. How to model the value of "real options," as determined by flexible design principles, for hydropower facilities in developing nations given the uncertainties of climate change, energy demand, and cost overruns. PhD thesis, University of Colorado, 2017.
- [154] Zoe Szajnfarber and Erica Gralla. Qualitative methods for engineering systems: Why we need them and how to use them. Systems Engineering, 20(6):497–511, 2017.
- [155] Nathan Taylor and Andrew Western. Avoiding Day Zero in Australia. Pursuit (University of Melbourne), February 2018.
- [156] C. Tebaldi and R. Knutti. The use of the multi-model ensemble in probabilistic climate projections. *Philosophical Transactions of the Royal Society A: Mathematical, Physical and Engineering Sciences*, 365(1857):2053–2075, 2007.
- [157] Claudia Tebaldi, Richard L. Smith, Doug Nychka, and Linda O. Mearns. Quantifying uncertainty in projections of regional climate change: A Bayesian approach to the analysis of multimodel ensembles. *Journal of Climate*, 18(10):1524–1540, 2005.
- [158] The World Bank. World Bank Open Data. https://data.worldbank.org/ indicator/ER.GDP.FWTL.M3.KD?locations=SA, 2010.
- [159] Matthew Tonkin and John Doherty. Calibration-constrained Monte Carlo analysis of highly parameterized models using subspace techniques. Water Resources Research, 45(12), 12 2009.
- [160] Yeou-Koung Tung. Groundwater Management By Chance-Constrained Model. Journal of Water Resources Planning and Management, 112(1):1–19, 1986.
- [161] Sean W. D. Turner, David Marlow, Marie Ekström, Bruce G. Rhodes, Udaya Kularathna, and Paul J. Jeffrey. Linking climate projections to performance: A yieldbased decision scaling assessment of a large urban water resources system. Water Resources Research, 50(4):3553–3567, 4 2014.
- [162] Christian Urich and Wolfgang Rauch. Exploring Critical Pathways for Urban Water Management: Identifying Robust Strategies under Deep Uncertainties. Water Research, 66:374–389, 2014.
- [163] Yorck von Korff, Katherine A. Daniell, Sabine Moellenkamp, Pieter Bots, and Rianne M. Bijlsma. Implementing participatory water management: Recent advances in theory, practice, and evaluation. *Ecology and Society*, 17(1), 2012.

- [164] Charles J Vörösmarty, Pamela Green, Joseph Salisbury, and Richard B Lammers. Global Water Resources: Vulnerability from Climate Change and Population Growth. 289(5477):284–288, 2000.
- [165] Warren E. Walker, Marjolijn Haasnoot, and Jan H. Kwakkel. Adapt or perish: A review of planning approaches for adaptation under deep uncertainty. *Sustainability*, 5:955–979, 2013.
- [166] W.E. Walker, P. Harremoëes, J. Rotmans, J.P. Sluijs, M.B.A. Van Der Asselt, P. Van Janssen, and Krayer Von Krauss M.P. A Conceptual Basis for Uncertainty Management. *Integrated Assessment*, 4(1):5–17, 2003.
- [167] Tao Wang and Richard De Neufville. Building Real Options into Physical Systems with Stochastic Mixed-Integer Programming. Options, pages 1–35, 2004.
- [168] Tao Wang and Richard De Neufville. Real Options "in" Projects. In 8th Real Options Annual International Conference, pages 1–19, Paris, France, 2005.
- [169] James F. Williams and Ibrahim Al-Sagaby. Simulated changes in water level in the minjur aquifer, riyadh area, saudi arabia. Technical report, Saudi Arabian Ministry of Agriculture and Water; United States Geological Survey, 1982.
- [170] World Bank. Coastal Region Water Security and Climate Resilience Project for Kenya. http://projects.worldbank.org/P145559/?lang=en&tab=overview.
- [171] World Bank Group. Enhancing the Climate Resilience of Africa's Infrastructure: The Power and Water Sectors. The World Bank, Washington, DC, 2015.
- [172] Liang Xue. Application of the Multimodel Ensemble Kalman Filter Method in Groundwater System. Water, 7(2):528–545, 2 2015.
- [173] Liang Xue, Dongxiao Zhang, Alberto Guadagnini, and Shlomo P. Neuman. Multimodel Bayesian analysis of groundwater data worth. Water Resources Research, 50:8481–8496, 2014.
- [174] David N. Yates. WatBal: An Integrated Water Balance Model for Climate Impact Assessment of River Basin Runoff. International Journal of Water Resources Development, 12(2):121–140, 1996.
- [175] T.-C. Jim Yeh, Lynn W Gelhar, and Allan L Gutjahr. Stochastic Analysis of Unsaturated Flow in Heterogeneous Soils 1. Statistically Isotropic Media. Water Resources Research, 21(4):447–456, 1985.
- [176] Kenneth C Young. A Multivariate Chain Model for Simulating Climatic Parameters from Daily Data. Journal of Applied Meteorology, 33(6):661–671, 1994.
- [177] Harrison B. Zeff, Jonathan D. Herman, Patrick M. Reed, and Gregory W. Characklis. Cooperative drought adaptation: Integrating infrastructure development, conservation, and water transfers into adaptive policy pathways. *Water Resources Research*, 52:7327–7346, 2016.
- [178] Stephen X. Zhang and Vladan Babovic. A real options approach to the design and architecture of water supply systems using innovative water technologies under uncertainty. *Journal of Hydroinformatics*, 14(1):13, 2012.